

The mental health and wellbeing of children with disabilities

NATHANIEL SCHERER

Thesis submitted in accordance with the requirements for the degree of Doctor of Philosophy of the University of London

August 2024

Department of Population Health

Faculty of Epidemiology and Population Health

LONDON SCHOOL OF HYGIENE & TROPICAL MEDICINE

No funding received

Research group affiliation: International Centre for Evidence in Disability

Declaration

I, Nathaniel Scherer, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Abstract

In this thesis, I explore factors and strategies that influence and promote mental health and wellbeing among children with disabilities.

Discussion is based on findings from four published articles. Paper 1 presents a scoping review of mental health support for deaf and hard of hearing children. Qualitative research in paper 2 explores communication, inclusion and mental health among deaf and hard of hearing children in the Gaza Strip. Paper 3 outlines the process of developing mental health and psychosocial support guidelines for deaf and hard of hearing children in the Gaza Strip. Paper 4 presents qualitative research on the experiences of adults and children with intellectual disabilities and their families in England and Scotland during the COVID-19 pandemic.

Using a social-ecological framework, I synthesise and discuss findings in the context of the wider literature. I consider factors that influence the mental health and wellbeing of children with disabilities across four levels (environment, community, caregivers, child), as well as strategies for support, including treatment and methods to promote mental health. Findings show that disability inclusion is central to the mental health and wellbeing of children with disabilities, influencing the societies in which children live and the systems available to support them. Considering environment, government action (and inaction) can marginalise children with disabilities and their families, negatively impacting mental health and wellbeing. This was exemplified during the COVID-19 pandemic. At community-level, stigma and discrimination contribute to exclusion of children with disabilities, affecting mental health. At the level of the caregiver, caregiver mental health and caregiver knowledge on disability influence child wellbeing. At the individual-level, identity and selfesteem of children with disabilities influence their mental health and wellbeing.

This thesis contributes to our understanding of mental health and wellbeing among children with disabilities and examines the implications for research, policy and practice.

Acknowledgements

So many people have supported me throughout my career and this PhD. Thank you to my supervisors, Dr Sarah Polack, Dr Tracey Smythe and Dr Ritsuko Kakuma. I am hugely grateful for your support and endless encouragement. It has been a pleasure and a privilege to work with you all. Sarah, you have been a close mentor since I joined LSHTM. Your kindness, patience and wisdom know no end. I truly appreciate all that you have done for me and the opportunity to learn from you. Tracey, your productivity, attention to detail and drive are an inspiration. You have sparked novel ideas and encouraged innovation. Ritz, I've been lucky to benefit from your expertise and guidance since the early days of my MSc and research career. I value your thoughtful advice, critical reflection and collaboration.

Thank you to Professor Hannah Kuper and Dr Julian Eaton. Hannah, you have given me huge opportunity and support over the past seven years; not least inspiring me to start a PhD. I would not be where I am today without your influence. Julian, you have long championed my work in global disability and mental health, including the research in this thesis. Thank you for your mentorship and friendship. Thank you also to the wider team at the International Centre for Evidence in Disability and LSHTM. I could not ask for better colleagues.

Thank you to my collaborators on the research included in this thesis. In the Gaza Strip, thank you to Ramadan Hussein, Naim Kabaja, Fidaa Shurrab, Markus Baldus and everyone at Atfaluna Society for Deaf Children and CBM who supported the research. Thank you also to the local steering committee and technical reviewers for your partnership. It was a pleasure to conduct this research and develop the guidelines with you. In the UK, thank you to my colleagues at the International Centre for Evidence in Disability and the University of Glasgow, Professor Tom Shakespeare, Dr Shaffa Hameed, Dr Veronika Reichenberger, Professor Nicholas Watson, Dr Charlotte Pearson, Dr Jane Cullingworth and Dr Richard Brunner. I learned so much about qualitative research from each of you.

Thank you to all of the research participants for your generosity of time and spirit. Without you this research would not have been possible.

I dedicate this thesis to my family and friends. Thank you for everything. The past decade has been amazing thanks to you. Thank you in particular to my grandparents and uncle for your overwhelming generosity. Thank you Mim for your love, laughter and support to get me over the final hurdle. Finally, thank you to my wonderful Mum. You made everything possible.

Format of thesis

This thesis is presented in accordance with the London School of Hygiene & Tropical Medicine research degree regulations for the PhD by Publication.

- Section 1 introduces key concepts relevant to the thesis, including disability, mental health and the mental health of children with disabilities.
- Section 2 provides an overview of the research included in the thesis and four peer-reviewed and published journal articles.
- Section 3 presents the analytic commentary, in which I critically discuss my findings within the wider literature.
- Section 4 gives reflection on my research practice and development.
- Appendices contain research paper cover sheets and copyright information for my research.

List of included studies

Paper 1

Scherer N, Bright T, Musendo DJ, O'Fallon T, Kubwimana C, Eaton J, Kakuma R, Smythe T, Polack S. Mental health support for children and adolescents with hearing loss: scoping review. *BJPsych Open* (2021) 8 (1), e9 10.1192/bjo.2021.1045

Paper 2

Scherer N, Smythe T, Hussein R, Wapling L, Hameed S, Eaton J, Kabaja N, Kakuma R, Polack S. Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip. *PLOS Global Public Health* (2023) 3 (6), e0001635 10.1371/journal.pgph.0001635

Paper 3

Scherer N, Hussein R, Eaton J, Kabaja N, Kakuma R, Smythe T, Polack S. Development of mental health and psychosocial support (MHPSS) guidelines for deaf and hard of hearing children in the Gaza Strip. *PLOS Global Public Health* (2023) 3 (10), e0002427 10.1371/journal.pgph.0002427

Paper 4

Scherer N, Wiseman P, Watson N, Brunner R, Cullingworth J, Hameed S, Pearson C, Shakespeare T. 'Do they ever think about people like us?': The experiences of people with learning disabilities in England and Scotland during the COVID-19 pandemic. *Critical Social Policy* (2022) 43 (3), 423-447, 10.1177/02610183221109147

Acronyms and abbreviations

ADHD: Attention deficit hyperactivity disorder GRIPP: Getting Research into Policy and Practice IASC: Inter-Agency Standing Committee ICF: International Classification of Functioning, Disability and Health LMIC: Low- and middle-income country LSHTM: London School of Hygiene & Tropical Medicine mhGAP: Mental Health Gap Action Programme MHPSS: Mental health and psychosocial support MICS: Multiple Indicator Cluster Survey OPD: Organisation of persons with disabilities RCT: Randomised controlled trial UNCRPD: United Nations Convention on the Rights of Persons with Disabilities UNICEF: United Nations Children's Fund WHO: World Health Organization

List of figures

Figure 1. The International Classification of Functioning, Disability and Health (ICF)
Figure 2. Social-ecological framework of mental health and psychosocial support
Figure 3. Adapted social-ecological framework of mental health and psychosocial support for children with disabilities

Table of Contents

Abs	tract		3			
Ack	nowled	edgements	4			
Fori	mat of	f thesis	5			
List	of incl	luded studies	5			
Acro	onyms	s and abbreviations	6			
List	of figu	ures	6			
1. Ir	ntrodu	uction to disability and the mental health of children with disabilities	9			
1	.1.	Defining disability	9			
1	.2.	People with disabilities globally				
1	.3.	Disability and health	11			
1	.4.	Mental health	11			
1	.5.	Mental health and wellbeing of children with disabilities	12			
1	.6.	Terminology	13			
2. C	vervie	ew of thesis	14			
2	.1.	Aim of thesis	14			
2	.2.	esearch presented in the thesis				
	2.2.1	L. Mental health support for deaf and hard of hearing children in the Gaza Strip	15			
	2.2.2	2. Children with intellectual disabilities and their caregivers in the UK during COVI	ID16			
2	.3.	Key concepts in the analytic commentary	93			
3. A	nalytic	c commentary	95			
3	.1.	Vulnerability and mental health	95			
3	.2.	Social-ecological framework of mental health: enabling environment	96			
	3.2.1	L. Government policy and action	96			
	3.2.2	2. MHPSS systems for children with disabilities	99			
	3.2.3	3. Crisis and complex settings	101			
3	.3.	Social-ecological framework of mental health: community	102			
	3.3.1	L. Stigma, discrimination and social exclusion	102			
3	.4.	Social-ecological framework of mental health: caregivers	103			
	3.4.1	L. Mental health and wellbeing among caregivers of children with disabilities	104			
	3.4.2	2. Caregiver confidence, knowledge and skills	105			
3	.5.	Social-ecological framework of mental health: children with disabilities	106			
	3.5.1	L. Access to education	106			
	3.5.2	2. Identity, self-esteem and wellbeing	108			
	3.5.3	 Resilience and self-management 	111			
3	.6.	Social-ecological framework of mental health: summary	112			

3.7.	Reco	ommendations for policy and research	115		
3.7.	1.	Policy	115		
3.7.	2.	Research	116		
4. Discussion and personal reflections116					
4.1.	Disc	cussion on my research	116		
4.2.	Refl	lexivity	118		
5. Conclusion					
References					
Appendices					

1. Introduction to disability and the mental health of children with disabilities

1.1. Defining disability

The United Nations Convention on the Rights of Persons with Disabilities (UNCRPD) defines persons with disabilities as those with "long-term physical, mental, intellectual or sensory impairments which in interaction with various barriers may hinder their full and effective participation in society on an equal basis with others" [1].

Conceptualisation of disability is complex and multidimensional, and the UNCRPD definition is an evolution of medical and social models [2]. In the medical model, disability is synonymous with impairment in functioning. In this model, impairment restricts participation in daily activities, such as education and employment, causing disability [2]. The medical model would assert that a Deaf person cannot participate in society on an equal basis to hearing peers as a result of their hearing loss, which restricts oral communication with the majority of the population and limits response to auditory environmental cues. Under this model, medical interventions are recommended to alleviate disability. For example, hearing aids or a Cochlear implant to improve an individual's hearing. The social model of disability rejected this concept. Under this model, disability is a result of barriers in society, rather than an individual's impairment [2]. For example, a Deaf person may have difficulty communicating with a health professional, not because they cannot hear, but because the health facility does not provide a sign language interpreter. If environmental, attitudinal and institutional barriers are removed, then people with disabilities can participate in society on an equal basis as people without disabilities. This social model was a major leap forward in the conceptualisation of disability and supported global disability activism. However, the social model has been critiqued for being too simple and for undermining the impact of impairment on disability.

More recently, the World Health Organization (WHO) *International Classification of Functioning, Disability and Health* (ICF) biopsychosocial model of disability integrates concepts from both the medical and social model [3]. In line with the UNCRPD, the ICF framework (Figure 1) conceptualises disability through the interaction of a health condition, personal factors and environmental factors.



Figure 1. The International Classification of Functioning, Disability and Health (ICF)

In this model, a health condition (e.g. depression) can lead to an impairment in functioning (e.g. cognitive function), which can contribute to activity limitations (e.g. difficulty completing tasks) and thus participation restrictions (e.g. employment). The extent of activity and participation restriction is influenced by environmental factors (e.g. attitude of employer) and personal factors (e.g. family support). The components of the ICF model interact with one another to influence the extent to which a person will experience disability. In this thesis, I use the UNCRPD definition and the ICF framework when describing disability and children with disabilities.

1.2. People with disabilities globally

Approximately 16% of the global population (1.3 billion) has a disability [4]. Nearly 10% of children (240 million) have a disability [5]. Estimates indicate that 80% of people with disabilities live in lowand middle-income countries (LMICs), with prevalence higher in LMICs compared to high-income countries [6]. For instance, the prevalence in West and Central Africa (15%) and Middle East and North Africa (13%) is more than double-that of Europe and Central Asia (6%). Disability is more common in LMICs due to a combination of factors, including increased malnutrition and infectious diseases, poorer sanitation and greater exposure to environmental hazards, and lower availability of healthcare services [6].

Disability can have a major impact on people's lives. People with disabilities are more likely to live in poverty [7], less likely to be employed [8], and are more likely to experience violence and abuse [9]. Children with disabilities experience exclusion and deprivation. Compared to peers without disabilities, children with disabilities are 34% more likely to be stunted, 49% more likely to never have attended school, 41% more likely to feel discriminated against, 51% more likely to be unhappy and 20% less likely to have expectations of a better life [10].

1.3. Disability and health

Article 25 of the UNCRPD states that "persons with disabilities have the right to the enjoyment of the highest attainable standard of health without discrimination on the basis of disability" [1]. Although important to note that they are not a homogenous group, people with disabilities typically have greater healthcare needs than people without disabilities [4]. They may be at greater risk of poor health as a result of an underlying health condition. For example, someone who is blind as a result of diabetic retinopathy will have diabetes, which may increase the risk of stroke. People with disabilities may also need additional support for their disability, such as rehabilitation services or assistive technology. Despite this greater need, barriers in society, such as inaccessible health facilities or stigma from health professionals, mean that people with disabilities are less likely to access healthcare services [4, 11]. Poorer access to health service results in poorer health. With greater health needs, limited access to healthcare exacerbates health inequity for people with disabilities [4]. Health inequities mean that people with disabilities have higher mortality and morbidity. Mortality among people with disabilities is 2.4 times higher than people without disabilities, people with disabilities have a life expectancy that is 10 to 20 years lower, and children with disabilities have five times the odds of being seriously ill in the last year [12, 13]. In the context of the global agenda, this inequity limits achievement of Sustainable Development Goal 3, Good Health and Wellbeing and the policy of "leave no-one behind" [14]. The WHO has called for renewed efforts to promote disability inclusion in the health sector to alleviate health inequalities [4].

1.4. Mental health

Mental health conditions can negatively impact a person's ability to participate in all aspects of life, including school, work and local communities. The 2019 *Global Burden of Diseases, Injuries, and Risk Factors Study* found depression and anxiety to be the two most common and disabling mental health conditions globally; they were among the 25 leading causes of health-related burden [15]. Mental health conditions accounted for 125.3 million years lived with disability, and between 1990 and 2019, mental health conditions were the second leading cause of years lived with disability globally [15]. In 2019, 21.5 million children and adolescents (<20 years) experienced a mental health condition [16]. Between 1990 and 2019, the prevalence of depression and anxiety increased by 48.1% [15]. It is expected that prevalence will continue to increase as the global population grows.

Health systems must be able to address the mental health needs of a growing population, but across the world, including high-income countries such as the UK, provision, quality of care and access to services is often poor [17, 18]. Service provision is particularly low in LMICs [19, 20]. Key challenges

include inadequate policy and legislation, limited resources and funding (including limited trained mental health specialists), and limited availability of contextually and culturally adapted evidencebased mental health interventions [20]. The field of 'Global Mental Health' developed in the early 2000s to address the mental health treatment gap in low-resource settings. At the time, 80% of people in LMICs needing mental health support did not access services [21]. Ten years on from the 2007 *Lancet Series on Global Mental Health*, the 2018 *Lancet Commission on Global Mental Health* and *Sustainable Development* called for continued expansion of mental health within global health priorities in order to achieve the *Sustainable Development Goals* [19]. The Commission called for continued efforts to reduce the treatment gap, as well as gaps in prevention and quality of care.

Global Mental Health has traditionally focused on a medical approach to mental health. Given the gap in service provision, treatment and intervention have been prioritised. However, in recent years, the field of Global Mental Health has shifted towards human rights approaches, many of which are founded in the principles of the UNCRPD [19]. Approaches in Global Mental Health are shifting to address social determinants of mental health conditions, value human rights and better engage people with lived experience. The *QualityRights* initiative highlights this recent shift. This global initiative aims to provide training and guidance on developing health systems that are person-centred and based on human-rights, in line with the UNCRPD [22].

In this thesis, I will discuss mental health systems across both high and low-resource settings. Two of the included papers in this thesis are from a low-resource setting. One of my included papers is conducted in the UK and another is a global systematic review (of which much of the included research comes from high-income settings). However, my wider research experience has largely been conducted in LMICs and I hold an MSc in Global Mental Health. Thus, my reflections in this thesis may at times lean towards discussion in the context of Global Mental Health and low-resource settings, although I aim to present reflections applicable to diverse contexts.

1.5. Mental health and wellbeing of children with disabilities

The UNCRPD definition of disability includes people with long-term mental health conditions experiencing psychosocial disability. Mental health conditions are a substantial impact in and of themselves, but they are also associated with comorbid presentation of health conditions [23]. It is on this group that this thesis is focused. I present and discuss findings relevant to the mental health and wellbeing of children with physical, sensory and intellectual disabilities. I do not discuss children with mental health conditions or psychosocial disability that do not have a co-morbid physical, sensory or intellectual disability. Evidence suggests that children with disabilities have a higher prevalence of mental health conditions than children without disabilities [24-29]. *The Lancet Commission on Global Mental Health and Sustainable Development* calls for directed action to address the needs of people with disabilities, who may be at increased risk due to marginalisation, discrimination, isolation and a lack of access to fundamental rights and services [19]. The Commission notes children and adolescents with disabilities, and in particular children with intellectual disabilities, who are at risk of forced detention and denial of mental health care. Although growing, current evidence on the mental health of children with disabilities is of low quality, there is limited literature from LMICs, and there is limited research into the social and environmental factors that influence poor mental health in this population [24-29].

1.6. Terminology

For the purposes of this thesis, I use the following definitions and conceptualisations regarding mental health and wellbeing:

- Mental health: The World Health Organization defines mental health as a "state of mental well-being that enables people to cope with the stresses of life, realize their abilities, learn well and work well, and contribute to their community" [30].
- Mental health conditions: Mental health conditions can be described as "common" (e.g. mild depression and anxiety) and "severe" (e.g. moderate to severe depression, bipolar disorder, schizophrenia) [31]. Severe mental health conditions are less prevalent and may result in more severe impairment. I use the term "conditions", although other terms are common in the literature, including "disorders", "illnesses" and "problems". I have aimed to be consistent in my use of mental health conditions, as other language may be inappropriate or stigmatising.
- Wellbeing: The World Health Organization defines wellbeing (or well-being) as a "positive state experienced by individuals and societies. Similar to health, it is a resource for daily life and is determined by social, economic and environmental conditions. Wellbeing encompasses quality of life and the ability of people and societies to contribute to the world with a sense of meaning and purpose" [32].

In this thesis, I use mental health in the context of alleviating mental health conditions or promoting good mental health. The range of mental health conditions is broad and diverse, but in this thesis, I use the term to refer to common mental health conditions, including depression and anxiety, rather than severe conditions, such as psychotic disorders. Severe mental health conditions are important

to consider, but they have not been the focus of my research and are not discussed in detail. I use wellbeing to describe a state of psychosocial wellbeing that encompasses good mental health and positive emotions, as well as feelings of self-acceptance, purpose, autonomy and positive relationships.

2. Overview of thesis

2.1. Aim of thesis

In this thesis, I present a portfolio of four published articles that explore mental health and wellbeing among children with disabilities [33-36]. The four articles were derived from two research projects conducted in the Gaza Strip and the UK during COVID-19. The analytic commentary presented alongside these articles synthesises the findings and presents my discussion of the research in the context of existing literature. In this analytic commentary, I examine factors that contribute to the mental health and wellbeing of children with disabilities and I discuss how my research has contributed to understanding on this topic. I highlight gaps in the evidence and directions for future research, policy and practice.

It is important to note that my research focused on deaf and hard of hearing children and youth with intellectual disabilities. Portions of my discussion will focus primarily on these groups, although I present discussion and draw from relevant literature on children with disabilities more broadly.

2.2. Research presented in the thesis

This thesis includes the following first-authored, peer-reviewed articles:

- Paper 1 [33]: Scherer N, Bright T, Musendo DJ, O'Fallon T, Kubwimana C, Eaton J, Kakuma R, Smythe T, Polack S. Mental health support for children and adolescents with hearing loss: scoping review. *BJPsych Open* (2021) 8 (1), e9 10.1192/bjo.2021.1045
- Paper 2 [35]: Scherer N, Smythe T, Hussein R, Wapling L, Hameed S, Eaton J, Kabaja N, Kakuma R, Polack S. Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip. *PLOS Global Public Health* (2023) 3 (6), e0001635 10.1371/journal.pgph.0001635
- Paper 3 [34]: Scherer N, Hussein R, Eaton J, Kabaja N, Kakuma R, Smythe T, Polack S.
 Development of mental health and psychosocial support (MHPSS) guidelines for deaf and hard of hearing children in the Gaza Strip. *PLOS Global Public Health* (2023) 3 (10), e0002427 10.1371/journal.pgph.0002427

 Paper 4 [36]: Scherer N, Wiseman P, Watson N, Brunner R, Cullingworth J, Hameed S, Pearson C, Shakespeare T. 'Do they ever think about people like us?': The experiences of people with learning disabilities in England and Scotland during the COVID-19 pandemic. *Critical Social Policy* (2022) 43 (3), 423-447, 10.1177/02610183221109147

Sections 2.2.1. and 2.2.2. provide an overview of the two research projects that these papers relate to. The published articles are presented immediately after, before the analytic commentary.

2.2.1. Mental health support for deaf and hard of hearing children in the Gaza Strip There are 70 million deaf and hard of hearing children worldwide [37]. Deaf and hard of hearing children are at increased risk of mental health conditions [38, 39], but there is a dearth of highquality evidence examining this relationship and intervention/promotion strategies, particularly in LMICs [28].

Papers 1-3 come from a collaborative research project in the Gaza Strip between 2020-2022, on which I was lead researcher. This project aimed to understand the factors that influence mental health and wellbeing among deaf and hard of hearing children in the Gaza Strip, explore mental health and psychosocial support (MHPSS) strategies and develop school-based guidelines on MHPSS for deaf and hard of hearing children. The project was conducted by the London School of Hygiene & Tropical Medicine (LSHTM), Atfaluna Society for Deaf Children and CBM International. I was lead researcher at LSHTM. The Gaza Strip is a Palestinian Territory under Israeli occupation. At the time of study, available evidence showed that youth in the Gaza Strip experienced high rates of depression, anxiety and post-traumatic stress disorder, yet faced difficulties accessing services [40, 41]. In a 2019 *Multiple Indicator Cluster Survey* (MICS), disability prevalence among children aged 5-17 in the Gaza Strip was estimated to be high at 12.8%; 7.4% had anxiety and 0.8% (~50,000 children) had difficulty hearing [42].

The process to develop the guidelines was collaborative and participatory. With local partners, we formed a local steering committee, comprised of Deaf adults, family members, members of an organisation of deaf and hard of hearing people, teachers, mental health specialists and government officials. I worked with this committee and Atfaluna to develop the research priorities and methods, to analyse and interpret findings, and to develop guideline content. Consultation with local Deaf leaders, international deaf and hard of hearing people with expertise on mental health, and the World Federation of the Deaf further supported development of these evidence-based guidelines. Research to inform the guidelines included systematic review (paper 1), qualitative research (paper 2) and pilot study (paper 3).

Paper 1 presents a systematic review of mental health support for deaf and hard of hearing children worldwide [33]. Findings identify MHPSS strategies and considerations for inclusive service provision. This was the first global review to synthesise evidence on mental health support for deaf and hard of hearing children. Paper 2 used qualitative methods to understand factors that influence the mental health and wellbeing of deaf and hard of hearing children in the Gaza Strip. In-depth interviews were conducted with deaf and hard of hearing children, caregivers and teachers in mainstream and special schools [35]. Factors that negatively influenced mental health and wellbeing included lack of accessible communication, social exclusion, and negative attitudes. Paper 3 describes the process of developing the MHPSS guidelines, including information on co-development, additional findings from the qualitative research and results from two pilot studies [34]. Briefly, the pilot studies demonstrated feasibility and acceptability of the guidelines. Teachers said they would need support from the school administration to implement over the long-term. The research project was conducted during the COVID-19 pandemic, impacting data collection methods, as described in papers 2 and 3. The guidelines are freely available online in <u>English</u> and <u>Arabic</u>.

At the time of writing this thesis, there is major crisis and conflict in the Gaza Strip. When conducting our research in 2020, the Gaza Strip was a humanitarian setting, given the occupation and blockade by Israel, which limited economic opportunities and access to daily needs [43]. However, the region had not experienced major armed conflict for some years, certainly not at the current level. Armed conflict was not reported by partners and participants as a major influence in relation to the mental health of deaf and hard of hearing children. Findings in papers 1-3 and reflections in the analytic commentary are relevant to the time of our research (2020-2022) and the data obtained. Discussion in this thesis may not be relevant to the current context in the Gaza Strip, where children and families are experiencing increased conflict, trauma and psychological stress.

2.2.2. Children with intellectual disabilities and their caregivers in the UK during COVID-19

In the UK, people with disabilities were three times more likely to die during the COVID-19 pandemic than people without disabilities [44]. Risk of death involving COVID-19 was 1.4-1.6 greater for people with disabilities [44], and people with intellectual disabilities experienced a 7.2-8.2-fold higher rate of COVID-related death [45].

Paper 4 presents qualitative research into the experiences of adults and children with learning disabilities (globally termed intellectual disability) and their caregivers in England and Scotland during the COVID-19 pandemic [36]. The research was conducted in collaboration with the University of Glasgow in response to rapid funding calls at the start of the pandemic. Findings showed that adults and children with intellectual disabilities and caregivers felt abandoned by the

government. As a result, children and caregivers experienced strain on their mental health and wellbeing. The project elicited other published works, to which I am co-author and to which I refer to throughout the thesis [46-48]. The research findings have been used by disability researchers and advocates in the *UK COVID-19 Inquiry*, outlining the structural inequalities that contributed to increased risk of mortality, morbidity and hardship for people with disabilities during the pandemic [49]. The pandemic exacerbated existing exclusion and inequities for people with disabilities. As such, there have been calls for a post-pandemic response that better includes people with disabilities in policy and programme development and implementation [50].

Note that paper 4 focuses on both adults and children with intellectual disabilities. The sample comprised of adults with intellectual disabilities, caregivers and family members of adults and children with intellectual disabilities, and representatives from organisations supporting adults and children with intellectual disabilities. In the analytic commentary, I draw on findings relevant specifically to children with intellectual disabilities and their family members.



Review

BJPsych

Mental health support for children and adolescents with hearing loss:

scoping review

Nathaniel Scherer, Tess Bright, David John Musendo, Timothy O'Fallon, Chris Kubwimana, Julian Eaton, Ritsuko Kakuma, Tracey Smythe and Sarah Polack

Background

Children with hearing loss are at increased risk of mental health conditions, including behavioural problems, but there is limited evidence about available mental health support.

Aims

We aimed to map the evidence on mental health support for children and adolescents with hearing loss.

Method

Medline, Embase, PsycINFO and grey literature databases were searched until April 2021. Articles of any study design were eligible if they described an intervention supporting the mental health of children with hearing loss. No restrictions were placed on geography or publication date. Four reviewers independently screened results by title, abstract and full text. Study characteristics and outcome data were extracted, with results narratively synthesised.

Results

From 5629 search results, 27 articles were included. A large majority of the studies (81%, n = 22) were from high-income settings, with two-thirds (67%, n = 18) conducted in the USA. Less than half (41%, n = 11) of the articles adopted experimental research designs, and the majority of studies included small samples. The interventions presented were diverse, with the majority either therapy based (30%, n = 8) or skills training (30%,

n = 8). Interventions included ice-skating, parent-child interaction therapy and resilience training. When measured, interventions demonstrated at least some evidence of effectiveness, although this was not always assessed with gold-standard methodology.

Conclusions

The evidence is lacking in breadth, study quality and geographical spread. That said, what is available indicates a range of effective approaches to support the mental health of children with hearing loss. Additional research is needed to improve the breadth of evidence on mental health support for this population.

Keywords

Childhood experience; outcome studies; psychosocial interventions; rehabilitation; primary care.

Copyright and usage

© The Author(s), 2021. Published by Cambridge University Press on behalf of the Royal College of Psychiatrists. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (https://creativecommons.org/ licenses/by/4.0/), which permits unrestricted re-use, distribution, and reproduction in any medium, provided the original work is properly cited.

Background

As of 2019, an estimated 20% of the global population (or 1.57 billion) have hearing loss, with 430 million having moderate-to-complete hearing loss.¹ This is a 79% increase in reported rates from 1990, and by 2050, an estimated 2.45 billion people are estimated to have hearing loss, with 698 million having moderate-to-complete hearing loss. Although hearing loss is more common among older adults, there are approximately 70 million children aged 0–15 with hearing loss across the world.¹

Children with hearing loss may experience language deprivation, impacting development, communication and socioemotional skills.^{2,3} As a result, children with hearing loss are at increased risk of mental health conditions, such as anxiety and depression, with several studies demonstrating significantly higher prevalence of these conditions among children with hearing loss, compared with children without hearing loss. $^{4-9}$ Evidence also shows that children with hearing loss are more likely to experience behavioural problems, including conduct and hyperactivity disorders.^{10,11} Half of mental health conditions start by the age of 14, yet these often go undetected and untreated.¹² Among the general population, these conditions in childhood are associated with an increased risk of mental health concerns in adulthood, lower family income, lower probability of employment and lower probability of being married.13,14 Mental health conditions may also further disrupt ongoing child development, a challenge already present for children with hearing loss. Addressing mental health concerns, while

promoting emotional, behavioural and psychological well-being, is imperative among children with hearing loss, in order to support a healthy childhood and reduce the risk of adverse experiences in adulthood.

Youth mental health has received growing attention in recent years, as evidenced by the inclusion of 104 network meta-analyses and meta-analyses of randomised controlled trials (RCTs) on mental health interventions for children and adolescents in a recent umbrella review published in 2021.15 However, there is limited synthesised evidence on interventions and types of support available for children with hearing loss that promote psychological, emotional and behavioural well-being, prevent mental health conditions and treat conditions that do arise. In 2019, a systematic review on the assessment and treatment of behavioural disorders in children with hearing loss found limited evidence of interventions to address behavioural problems.¹¹ Evidence on interventions available for other mental health conditions is lacking. Information on the types of interventions applied, their characteristics and evidence of their effectiveness is needed to inform support programmes for children with hearing loss, whether these be in school, the community or delivered through the healthcare system.

Aims

This study aimed to systematically identify and map the evidence on mental health support for children and adolescents with hearing loss. We aimed to answer the following questions.

- (a) What are the characteristics of the available literature describing mental health support for children with hearing loss?(b) What mental health interventions and support have been pro-
- vided or evaluated for children with hearing loss?
- (c) What evidence is available on the effectiveness of these interventions?
- (d) What are the current gaps in the available evidence?

Method

With expected limited and heterogeneous data, we chose to conduct a scoping review, rather than an alternative evidence synthesis, in order to map the available evidence.¹⁶ The protocol for the review was registered on the Open Science Framework on 3 March 2020 (osf.io/8qdbz/). Ethics approval was obtained from the London School of Hygiene & Tropical Medicine Research Ethics Committee (Ref: 19144).

The scoping review has been conducted and reported according to the PRISMA Extension for Scoping Reviews (PRISMA-ScR) guideline and followed Arksey & O'Malley's framework for scoping reviews.^{17,18} We conducted the review across five stages: (a) identifying the research question; (b) identifying relevant studies; (c) study selection; (d) charting the data; and (e) collating and summarising the results.

Eligibility criteria

2

- (a) Population: children aged 6–18, with diagnosed, proxy-report or self-reported hearing loss, of any severity. Included participants did not need to have a diagnosis of a mental disorder.
- (b) Intervention: any initiative designed to improve the mental health and well-being of children with hearing loss, including interventions developed specifically for children with hearing loss or those adapted from interventions aimed at the general population. Interventions promoting developmental skills and protective factors, such as emotional regulation, resilience and self-esteem, were included if connection was made to child mental health and well-being. Interventions were also included if they addressed behavioural problems or disorders, such as hyperactivity or aggression. Activities could be focused on promotion, prevention or treatment, such as social interventions, skills development, targeted recreational activities and therapies. They could be delivered by any personnel in any setting. Interventions for parents, caregivers or other adults (for example teachers) were included if children with hearing loss were an intended beneficiary. Similarly, interventions were included if provided to a diverse group, as long as children with hearing loss were one of the beneficiaries.
- (c) Comparator: studies with and without a control or comparison group were included. If a control or comparator is present, they must also be children with hearing loss of the same age, but who have not received the intervention.
- (d) Outcomes: studies with or without assessment of any outcome were included. Where applicable, outcomes of interest included scores on mental health screening tools, acceptability/feasibility, cost-effectiveness and other reasonable data.
- (e) Study design: published literature of any study design (quantitative, qualitative and mixed methods). Descriptive literature (i.e. without research methods applied) was included, if sufficient detail was provided on the intervention(s) available for children with hearing loss. Reviews and opinion pieces were excluded. There were no restrictions placed on geographic location, although articles needed to be in English. There were no limits placed on publication date. Grey literature,

including dissertations and conference presentations, was included. Only articles with an available full text were included.

Search strategy

Articles were identified through a systematic search of Medline, Embase and PsycINFO. The search was initially conducted on 27 April 2020 and updated on 27 April 2021. An example of the search strategy is available in Supplementary File 1 available at https://doi.org/10.1192/bjo.2021.1045. Reference lists of each included study were examined in search of additional articles for inclusion.

The search for grey literature was conducted through OpenGrey and Google Scholar. Additionally, experts in this field were contacted for recommendations of known reports, and the websites of notable disability and hearing loss organisations were manually searched (Supplementary File 2).

Study selection

Authors N.S., T.B., D.J.M. and T.O.'F. independently screened all titles and abstracts against the eligibility criteria. Each record was screened by two reviewers. Eligible full-text articles were then independently reviewed by two reviewers. Although excluded from the final synthesis, systematic reviews and full-length books identified in the database search were manually screened by full-text, with included articles and book chapters selected for full-text screen, if relevant. Records identified from the reference lists of included articles were discussed between the two reviewers, with a third and fourth reviewer consulted if needed. This review process was conducted using Covidence software.

Data extraction and charting

N.S. extracted the data for each study, using a custom form, developed in Excel. This extraction form was first piloted on three included articles, with amendments made as necessary. T.B., D.J. M. and T.O.'F. each independently reviewed one-third of data extracted. Discrepancies were discussed and resolved with N.S., with support from a third reviewer if needed. Data extracted included:

- (a) Publication details: author, year of publication, title, country, aims/objectives, study design.
- (b) Characteristics of mental health support: type (promotion, prevention, treatment), intended outcome, setting, delivery agent, intervention components.
- (c) Outcomes: type (effectiveness, feasibility, acceptability, servicedelivery related), measurement tools, findings (both narrative and statistical).

Quality assessment

For an assessment of risk of bias, we used the critical appraisal tools from the Joanna Briggs Institute (JBI). There are various tools provided by JBI, designed for use with different designs, such as case report or RCT. Each tool can be used to determine the extent to which a study has addressed the possibility of bias in its design, conduct and analysis, via a series of relevant questions and standar-dised responses.¹⁹

Studies were assessed by N.S., with scores reviewed by S.P. Based on the assessment against a JBI checklist, each study was rated as having high, medium or low risk of bias.

(a) Low risk: all or almost all of the criteria were fulfilled, and those that were not fulfilled were thought unlikely to alter the conclusions.

Downloaded from https://www.cambridge.org/core. 06 Dec 2021 at 10:38:33, subject to the Cambridge Core terms of use

Mental health support for children and adolescents with hearing loss



Fig. 1 PRISMA flow chart.

- (b) Medium risk: some of the criteria were fulfilled, and those not fulfilled were thought unlikely to alter the conclusions of the study.
- (c) High risk: few or no checklist criteria were fulfilled, and the conclusions of the study were thought likely to alter if these had been met.

Synthesis of results

Under the umbrella of mental health, the focus of this review, we have grouped articles by interventions that address psychological well-being and behavioural problems. Psychological well-being, in this instance, refers broadly to emotional well-being and mental health, and includes diagnosis or symptoms of mental health conditions, as well as related domains, such as resilience and self-esteem. Behavioural problems include disorders and concerns relating to disruptive and challenging behaviours, such as aggression, impulsivity and defiance. Specific behavioural disorders include attention-deficit hyperactivity disorder or oppositional defiant disorder.

Under these two subheadings, we grouped articles by intervention type. Findings have been summarised narratively, with the focus on the characteristics of the interventions. Information on effectiveness has been narratively presented, where this information was available.

When data presented in an article is disaggregated by age, with participants older or younger than the inclusion criteria included, only that attributable to children within the age range of 6–18 years was included in the synthesis.

Results

The database search generated 5629 results, from which 1253 duplicates were removed. In total, 4376 records were screened by title and abstract, from which 4245 were excluded based on the criteria above (Fig. 1). There were 130 full-text articles that were eligible for screening, with 6 additional records included from systematic reviews screened at full text (n = 3), records identified via the reference lists of included articles (n = 2) and the grey literature (n = 1). From 136 full texts assessed, 27 articles were included in the synthesis.^{20–46}

Study characteristics

Of the 27 included articles, 18 (67%) come from North America, and specifically the USA (Table 1). Of the remaining, four (15%) were conducted in Europe, four (15%) the Middle-East, and one (4%) in Asia. None were from Africa, Latin America or Australasia. Nine (33%) were published after the year 2010, with 18 (67%) prior to 2010. Eleven (41%) were published prior to the year 2000.

Eighteen (67%) provide information on support focused on psychological well-being, and related domains. Of these, four (15%) targeted protective psychological factors (such as resilience and emotion management), one (4%) social anxiety disorder, one (4%) substance misuse disorder and one (4%) obsessive-compulsive disorder. Fourteen (52%) of the 27 included articles focused support

Downloaded from https://www.cambridge.org/core. 06 Dec 2021 at 10:38:33, subject to the Cambridge Core terms of use.

Table 1 Summary of article character	eristics	
Variable and category	n	%
Region		
Asia	1	4
Europe	4	15
Middle-East	4	15
North America	18	67
Country		
Finland	1	4
India	1	4
Iran	3	11
Poland	1	4
Turkey	1	4
UK	2	7
USA	18	67
Country income status		
High	22	81
Upper-middle	4	15
Lower-middle	1	4
Study design		
Before and after	3	11
Case report	4	15
Case series	5	19
Pliot study	2	/
RCI	6	22
Qualitative	1	4
Descriptive	6	22
Decade of publication		45
1970	4	15
1980	1	4
1990	6	22
2000	/	26
2010	/	20
Source	Z	/
Journal	22	OF
Book	23	00
Thosis	2	7
Montal hoalth domain	2	'
Psychological well being	10	47
Rehavioural problems	14	52
Sample size	14	52
<10	0	33
11_20	1	15
21_50	4	15
51-100	+ 2	7
>100	2	7
N/A	5	10
Risk of bias	5	17
LOW	7	26
Medium	10	37
High	10	15
N/A (description only)	4	22
	0	22

RCT, randomised controlled trial; N/A, not applicable.

on behavioural problems, such as conduct disorder. Some articles focused on both psychological well-being and behavioural problems and the percentages presented therefore total greater than 100%.

Twenty (74%) of the interventions included treatment options, 17 (63%) promotion activities and 7 (26%) prevention techniques. Eight (30%) interventions were therapeutic in nature, nine (33%) were a structured training programme, two (7%) based on physical exercise, and one (4%) medical. Seven (26%) others were intervention packages, comprised of multiple single interventions.

Eighteen (67%) described activities delivered in schools, nine (33%) of which were special schools, three (11%) special education classrooms in a mainstream school and one (4%) a mainstream school. Four (15%) were conducted in residential settings, with schooling on site. Six (22%) were conducted in healthcare facilities,

one (4%) a community setting, and two (7%) artificial/research settings. One (4%) study did not provide information on the setting.

Nine (33%) of the included articles presented a case report or case series, six (22%) conducted randomised controlled trials, five (19%) used quasi-experimental methods (of which two were pilot studies) and one (4%) qualitative. Six (22%) of the articles described an intervention or support structure, without any research methods applied. One-third (33%) of articles included a sample size of less than ten participants, with only two studies (7%) including more than 100 participants.

Summarising the ages and degree of hearing loss among participants included presented a challenge. Studies provided various classifications on the severity of hearing loss, and we have converted these under the same terms in Table 2. The ages of participants and severity of hearing loss overlapped across studies, and summary statistics were difficult to present with much coherence.

Quality assessment

The six (22%) articles that describe (rather than evaluate) an intervention were not included in the quality assessment. Of the 21 remaining, 7 (33%) were rated as having a low risk of bias, 10 (48%) medium and 4 (19%) high. Potential sources of bias among those rated to have a medium or high risk, included: inappropriate statistical analysis, including the absence of statistical power or sample size calculations in RCTs; no clinical information provided for participants in a case series; and unsuitable measurement of outcomes, including collection time points and screening method.

Types of support

Psychological well-being

As seen in Supplementary Table 1, 17 articles targeted psychological well-being and related factors (for example resilience).

Five (29%) of these focused on forms of counselling, psychotherapy or group support.^{28,31,37,43,44} Examples included peersupport groups and therapeutic play. Four (24%) provided structured training and skills-based programmes, including group assertiveness training for social anxiety disorder.^{20,23,35,36} Two (12%) were focused on physical exercise and related activities, including iceskating and dance lessons.^{26,30} One (6%) study investigated the impact of hearing aids on psychological well-being.³⁸

The remaining five (29%) articles provided information on intervention 'packages' that included more than one intervention approach.^{22,24,40,45,46} These were typically delivered in healthcare or residential settings. Individual interventions described in the packages were typical of those provided in mental health programmes for the general population, and included cognitive-behavioural therapy, medication, art therapy and peer-support groups.

Behavioural problems

Fourteen articles provided support for behavioural problems. In four articles, the intervention or support provided addressed both psychological well-being and behavioural problems.

Five of the 14 (36%) provided counselling or variations of psychotherapy, including video-counselling.^{25,28,39,42,44} Three (21%) provided parents with training, and typically included skills-building exercises, using role-play and similar techniques.^{32,33,41} Three (21%) focused training efforts on children.^{29,24,35} For example, the PATHS (Promoting Alternative Thinking Strategies) curriculum, which includes daily activities at school, delivered by teachers.³⁴ One (7%) supported physical activity, in this instance, iceskating.³⁰ A further two (14%) articles described a support package including various activities.^{21,27} This included Walden House, a residential programme for children with hearing loss

Downloaded from https://www.cambridge.org/core. 06 Dec 2021 at 10:38:33, subject to the Cambridge Core terms of use

Construction and the second of the second second second	11 - 100 - Marine - 11							
Table 2 Individual article characteristics								
							Participants	
First author (year)	Country	Mental health domain	Study design	Control group	Age, years	Degree of hearing loss	(male/female)	Risk of bias
Ahmadi et al (2017) ²⁰	Iran	Social anxiety disorder	RCT	Y	12-16	Moderate to profound	48	Medium
Altshuler & Spady (1978) ²¹	USA	Behavioural problems	Case series	N	4-10	Profound	3 (2/1)	Medium
Anonymous (1978) ²²	USA	Psychological well-being	Descriptive ^a	N	12+	Moderate to profound	N/A	N/A
Ashori & Najafi (2021) ²³	Iran	Emotional regulation	RCT	Y	15-19	Mild to severe	34 (0/34)	Medium
Bernstein & Denno (2005) ²⁴	USA	Obsessive-compulsive disorder	Case report	N	3-21	Mild to profound	1 (0/1)	Low
Boham & Selkowitz (1981) ²⁵	USA	Behavioural problems	Descriptive	N	6–19	Moderate to profound	N/A	N/A
Borowiec et al (2019) ²⁶	Poland	Self-esteem	RCT	Y	9–13	Severe to profound	28 (16/12)	Medium
Burnes et al (1992) ²⁷	USA	Behavioural problems	Descriptive	N	Not stated	Moderate to profound	N/A	N/A
Chapel (2005) ²⁸	USA	Psychological well-being and behavioural problems	Case series	Ν	Not stated	Moderate to profound	4 (3/1)	Low
Donovan (2003) ²⁹	ASU	Behavioural problems	Case series	N	4-7	Moderate to profound	3 (2/1)	Medium
Dursun et al (2015) ³⁰	Turkey	Psychological well-being and	Before and after	N	8-16	Not stated	40 (24/16)	Medium
	runkey	behavioural problems	before and after		0 10	Not stated	40 (24) 10)	Weddin
Elkayam & English (2003) ³¹	USA	Psychological well-being	Pilot study	N	12–18	Mild to severe	15	Medium
Forehand et al (1974) ³²	USA	Behavioural problems	Case report	N	7	Moderate to severe	1 (1/0)	Medium
Garcia & Turk (2007) ³³	UK	Behavioural problems	Case report	N	10	Profound	1 (1/0)	Low
Greenberg & Kusché (1998) ³⁴	USA	Behavioural problems	RCT	Y	6–12	Severe to profound	57 (27/30)	Low
Hatamizadeh et al (2020) ³⁵	Iran	Resilience and behavioural problems	RCT	Y	12–15	Mild to severe	122 (74/48)	Low
Johnson & Sandberg (1992) ³⁶	USA	Substance abuse disorder	Descriptive	N	11-12	Mild to profound	N/A	N/A
Lasanen et al $(2019)^{37}$	Finland	Psychological well-being	Qualitative	N	7–17	Mild to severe	16 (4/12)	Low
Nehra et al (2001) ³⁸	India	Psychological well-being	Before and after	Y	15-19	Moderate to profound	14	Medium
Osborne (1977) ³⁹	USA	Behavioural problems	Before and after	N	2-21	Mild to profound	280	High
Sarti (1993)40	USA	Psychological well-being	Case series	N	10-13	Mild to profound	5 (4/1)	High
Shinn (2013)41	USA	Behavioural problems	Case report	N	9	Moderate	1 (1/0)	Low
Sullivan et al (1992) ⁴²	USA	Behavioural problems	RCT	Y	12-16	Mild to profound	71 (51/21)	Medium
Tinsley & Jedlicka (2012)43	USA	Psychological well-being	Pilot study	N	7–19	Not stated	20	High
Troester (1996)44	USA	Psychological well-being, resilience and behavioural problems	Case series	Ν	Not stated	Moderate to profound	3 (3/0)	High
Vreeland & Tourangeau (2003)45	USA	Psychological well-being	Descriptive	N	8-21	Mild to profound	N/A	N/A
Wright et al (2012) ⁴⁶	UK	Psychological well-being	Descriptive	Ν	8–18	Not stated	N/A	N/A
RCT, randomised controlled trial; Y, yes; N, a. Description of an intervention with no re	, no; N/A, not appli esearch methods a	cable. pplied.						

and behavioural problems, providing problem-solving skills, family therapy and role-play exercises.²⁷

Theoretical underpinnings

Twenty three of the 27 articles provided information on the theory or evidence used to develop the intervention. The remaining four articles provided descriptions of logic-based arguments for the intervention model, without substantial information on evidence or theory. Of the 23 articles providing this information, 12 (52%) provided evidence of previous use or effectiveness of the intervention among children without disabilities or hearing loss, four provided (17%) evidence of its use with children with and without hearing loss, and four (17%) articles provided evidence of the intervention among children with hearing loss only. Four (17%) based the intervention on theories developed for children without hearing loss, such as the ABCD (Affective-Behaviour-Cognitive-Dynamic) model of child development.

Adaptations made to interventions

Of the 27 articles, 16 described specific considerations and adaptations made to the intervention to accommodate the needs of children with hearing loss, particularly when the support provided was initially developed for the general population.

Of these, 13 (81%) described considerations of communication methods and needs, such as reading ability. Nine (56%) used sign language or provided sign language interpretation. Three (19%) made adaptations to the physical environment in which the intervention was delivered. For instance, seating groups in a circle to maximise visual contact, or providing pictures and posters to facilitate understanding. Three (19%) adapted the core structure, components, tools and methods of established interventions. For example, established programmes for the general population, such as D.A.R.E. and the PATHS curriculum were adapted and prepared with support from hearing loss specialists.

Outcomes

Effectiveness

In total, 19 articles provided information on the effectiveness of the intervention or programme.

Eleven of these presented outcomes with regards to psychological well-being, with all but one finding some evidence of a positive impact of the intervention. Specifically, three (27%) provided evidence on the effectiveness of training programmes for children with hearing loss, including resilience training. One of these, targeting social anxiety disorder, reported no significant differences in symptomology between intervention and control groups in children with profound hearing loss, but did see a difference for those with moderate hearing loss. Two (18%) demonstrated the positive effect of physical activity interventions; dance lessons and iceskating. Therapy-based interventions were found beneficial by two (18%) studies, one of which, a qualitative study, provided evidence of the positive perceived benefit from participants of regular peer-support groups, although changes in well-being were not measured nor observed with other standardised methods. One (9%) other found no difference in participants pre- and post-intervention after receiving child-centred play therapy. One demonstrated significant improvements in well-being, including reduced symptoms of anxiety and depression, after children had used hearing aids for 6 months. Two studies (18%) provided evidence on the effectiveness of intervention packages comprised of various approaches, including a package for obsessive-compulsive disorder. Of those described, five (45%) demonstrated improvements in scores on quantitative standardised measurement tools (such as the Strengths and Difficulties Questionnaire) after the intervention was implemented, of which four (36%) RCTs showed significant improvement in the intervention group and not the control group.

With regards to behavioural problems, 11 studies reported on outcomes, of which all but one found some evidence of a positive impact of the intervention. Five (45%) found evidence on the benefit of training programmes, including parent behaviour training and teacher delivered resilience behaviour training for children. Three (27%) demonstrated the positive impact of therapy-based interventions on behaviour, although one (9%) showed no benefit. One study (9%) showed improvements among children on the Strengths and Difficulties Questionnaire, including the behavioural domain, after regular ice-skating sessions. One (9%) showed the benefit of a triadic intervention model for behavioural problems, focused on a core of reinforcement in schools. Seven (64%) reported positive improvements in behavioural outcomes, assessed through standardised measurement tools. Three RCTs (27%) demonstrated significant improvement in the intervention group and not the control, after delivery.

Other outcomes

Four articles provided evidence about the acceptability of the intervention to participants, delivery agents and/or parents. In one pilot study, this was the primary outcome that was measured quantitatively by questionnaire, with participants reporting good acceptance of the counselling method provided to support psychological wellbeing. In another, teachers reported feeling motivated to deliver the functional communication training for behavioural problems, and were happy with the training provided. In the third, both parents and delivery agents (clinicians) provided positive feedback in a directed workshop, with regards to the structure and components of the Webster-Stratton programme for behavioural problems. In the final case study of parent-child interaction therapy for behavioural problem, the mother expressed the value of learning to play with her child and manage their behaviour in a way that was fun and engaging for them.

Discussion

Summary of evidence

To the best of our knowledge, this is the first attempt to systematically identify and map the evidence on mental health support for children and adolescents with hearing loss. The scoping review identified 27 articles, two-thirds of which were conducted in the USA and two-thirds published before 2010. A number of interventions and support initiatives were described in the literature, including a broad range of therapy and counselling, and targeted training programmes, for both children and caregivers.

The interventions identified were, by and large, found to be effective at supporting the mental health and well-being of children with hearing loss. Although not all articles provided information on outcomes, nor all via quantitative assessment, those that did offered some encouraging signs as to the benefits of mental health support for children with hearing loss. Dance lessons, resilience training programmes and hearing aids demonstrate the diverse gamut of effective interventions that can be utilised to support children with hearing loss. Interventions provided treatment, prevention and promotion options in various settings, supporting recent advocacy for population- and community-level support for child mental health.42 As the evidence suggests, an effective and sustainable approach to child mental health includes support from an early age, improving information and awareness, providing social and emotional learning activities, increasing detection and identification, and improving access to treatment and rehabilitation. The

6

Downloaded from https://www.cambridge.org/core. 06 Dec 2021 at 10:38:33, subject to the Cambridge Core terms of use

interventions in this review provide some examples of interventions that may be effective within such a system, although as we will go on to discuss, the strength of the current evidence is limited.

Most of the interventions identified in this review were provided in school, which follows evidence from among the general population as to the benefits of integrated mental health provision within education systems.⁴⁸ Interventions at school level provide a continuum of care that benefits a child's mental health, as well as their educational attainment. Only one identified intervention was conducted within a mainstream classroom, despite the recent global movement towards inclusive education.⁴⁹ That said, a recent report outlines the limited transformations in education systems across the world, with most countries facing difficulties in inclusive provision, with understanding of what this means, and resources and evidence on good practice variable across regions.⁵⁰ As reform continues to spread, there will need to be evidence gathered on appropriate mental health interventions in inclusive classrooms, to ensure children with hearing loss are not excluded.

When considering interventions for children with hearing loss, it is important to note components and delivery methods unique to this group. Many of the interventions were initially developed for use in the general population, and more than half of the included articles described interventions that had been developed from evidence for children without hearing loss and adapted for the needs of children with hearing loss. Adaptations were most commonly focused on communication, and often sign language. Communication is often cited as a major barrier and facilitator to accessing healthcare for people with hearing loss, and it is appropriate to see this as the focus of many adaptations.⁵¹ In some settings (typically highincome settings) such as the UK, and as seen in the included article from Wright et al (2012), specialist mental health services exist for children with hearing loss, where sign language and other accommodations are embedded in the service provided.⁴⁶ However, in other settings, provision of communication technologies and sign language interpretation may not always be possible, given restricted resources. Often, and especially in low- and middle-income countries, mental health interventions are adapted from one context to another, in order to meet the culture, needs and resources of a population. In low- and middle-income countries, this often includes adapting interventions developed in high-income settings, although there are good examples of adaptation in the other direction, in which we see novel and effective mental health interventions developed in low-resource settings and adapted to those in high-income settings (an example being the Friendship Bench, developed in Zimbabwe and adapted to New York City in the USA).⁵² Adaptation is common and needed, and the well-evidenced stages of this process should be noted by those looking to adapt mental health interventions for children with hearing loss, in order to ensure appropriate intervention components and delivery mechanisms. This will include standard adaptations to the context, but also specific adaptation to the needs of children with hearing loss, such as those seen in this review. We did not find any studies that provided evidence of an adaptation process through formative research, theory of change workshop, feasibility study, or similar methods, and this would be a useful area of research going forwards, to promote interventions for children with hearing loss that are contextually and culturally appropriate, and hopefully then effective and sustainable. These approaches also promote the participation of stakeholders, including those with hearing loss and/or mental health conditions.

Evidence gap

Although 27 articles were identified, there is concern about the breadth, scope and strength of the evidence within the literature available. With two-thirds (67%) of articles published before 2010,

and 41% pre-2000, it is evident that the published literature on this topic is not growing at a fast rate. Interest and research in disability has steadily increased since the United Nations Convention on the Rights of Persons with Disabilities (2008), and calls for further evidence on scalable mental health interventions have been delivered by leading figures for many years.⁵³ Despite an increased focus on global mental health and disability rights, this review suggests that mental health support for children with hearing loss is not gaining sufficient interest in the research community. The Lancet Commission on Global Mental Health and Sustainable Development, published in 2018, may stimulate additional research in the future, calling as it does for a focus on culturally appropriate and participatory approaches to translating evidence for promotion and care in mental health across diverse populations.⁵⁴ Our approach to classification using a continuum Our approach to classification using a continuum from well-being to mental health diagnosis draws on these ideas. Exploring the evidence among adult populations is also needed given the higher prevalence of hearing loss in this age group.

Furthermore, nearly all (81%) of the included articles come from high-income settings, with two-thirds (67%) conducted in the USA. Just one (4%) was conducted in a lower-middle or lowincome setting. There are no articles from Latin America or Africa, and just one study from Asia. The geographical scope of study into appropriate interventions for children with hearing loss must increase. Given that 80% of the world's population and people with disabilities are living in low- and middle-income countries, there is an urgent need for evidence on contextualised interventions.55 Relying on data from high-income settings has caused concern in the field of global mental health, as doing so reduces long-term effectiveness and sustainability of mental health interventions, especially when biomedical models of care are transferred to settings where they may not be culturally accepted, contextually feasible or appropriate.⁵⁴ There is growing evidence on new and adapted mental health interventions for the general population in low- and middle-income settings, and as shown in this review, there is the possibility of adapting these to suit the needs of children with hearing loss (for example sign language provision). Evidence is needed on the process of adaptation and contextualisation, in order to inform delivery in the given country, and to provide a research framework for others.

Most of the studies were assessed to have medium or high risk of bias. Just six (22%) of the studies available and included are RCTs. The majority of included articles offer interesting case reports, but these provide the rationale for larger-scale trials, rather than highquality evidence in and of themselves. Most of the studies included have also been conducted with very small sample sizes. Nearly half (48%) of studies contained a sample size smaller than 20, increasing to two-thirds (67%) when including those articles with no sample size available. To improve the evidence base with which to stimulate service provision and policy, there needs to be experimental research, with large samples, with which to build confident conclusions and inform scalable interventions. Potential sources of bias must be addressed, including appropriate statistical methods, including sample size calculations for sufficient power.

Limitations

There are several limitations to this review. First, we excluded articles that were not in the English language, and there may well be evidence missed that has been published in other languages. This is particularly important when considering the limited evidence from South America and East Asia. Our age range, 6-18 years, is broad and although not necessarily a limitation, it is important that readers of this review pay close attention to the targeted age range of interventions of interest. Younger children and older

adolescents will respond differently to interventions, and will have different mental health needs, and not all interventions will be applicable across age groups. Second, we could not source 19 full-text articles, in which there may have been relevant and included studies. We contacted authors directly and utilised various institutional access agreements, but still could not find these for assessment. Third, we did not assess publication bias. When interpreting the results on effectiveness, readers should be conscious of the risk of positive publication bias and a potential lack of interest to publish results on ineffective interventions. Finally, in their framework for scoping reviews, Arksey & O'Malley recommend an optional sixth step; consultation with stakeholders.¹⁸ Such consultation is time-consuming and costly, and was not feasible when undertaking this study, and may be an important area to address in any future and updated reviews.

Implications

This review has identified a number of articles and interventions, which generally showed positive results, but there are concerns over the breadth of the evidence available. There is evidence of some effective interventions, such as dance lessons and resilience training, but the applicability of the available evidence is limited by geographic location and publication date, and there is a need for more studies applying high-quality research methods. To improve the current evidence base, we must strengthen the quality of the research methods used and provide further research from low- and middle-income countries on adapting interventions to local contexts and on interventions tested at scale. Having a community of researchers set future research priorities, as informed by practitioners and people affected, will focus and strengthen research activities going forwards.

Recommendations on mental health support for children with hearing loss

Based on the findings of this review, we have listed below a number of considerations for delivering mental health support for children with hearing loss, including adaptations to existing mental health interventions.

- (a) Consider the intervention models identified in this review, including the following.
 - (i) Peer-support: children with hearing loss can benefit from interaction with peers, reducing isolation, while helping them learn more about themselves and their experience as a child with hearing loss. Peer-support can act as both a treatment and health promotion option.
 - (ii) Resilience training: identified by a number of studies in this review, resilience training may be an effective way of preventing mental health conditions, especially if implemented at an early age. This type of training may be particularly appropriate to integrate into schools.
 - (iii) Emotional and behavioural management training: this type of structured training programme can be provided for both parents and children. Helping parents learn the tools with which to support their child's emotional regulation and behaviour may help create a home environment that promotes well-being and positive coping strategies. Training children themselves, possibly in school, will further help children with hearing loss build the toolkit with which to understand and approach the challenges they face.
 - (iv) Physical activity: physical activity is a simple way to build well-being and promote mental health. In this review, the physical interventions identified were

specialist in nature – ice-skating and dance therapy are not necessarily available in all settings – but adapting these techniques and principles to a different context may prove effective.

- (b) Consider the following when adapting or developing an intervention.
 - (i) Knowledge and awareness: it is important that the delivery agent is given appropriate training and sensitisation on disability and hearing loss. There may also be cause to provide information to children without hearing loss in the same setting, in order to reduce any apparent stigma or discrimination; for example, to hearing children in a mainstream classroom.
 - (ii) Communication methods: each child will have different communication needs, depending on their severity of hearing loss and personal preferences. Include appropriate training for delivery agents on communication methods, including clarity of speech and the language used. Where necessary, provide sign language interpretation, and where relevant provide training on hearing aid or cochlear implant use, as this knowledge will facilitate improved communication and reduce stigma.
 - (iii) Physical environment: consider the environment in which the intervention is delivered. This may be a mainstream classroom, for instance, that is not set up to accommodate the needs of children with hearing loss. Considering the configuration of seating in a group is one example; children sitting in a circle may help improve visual contact and support those who lipread. Keeping the environment quiet and free from distractions will also help children with hearing loss, as might keeping the door closed and adding soft material to the underside of chairs.
 - (iv) Intervention materials and techniques: where suitable, use visual aids and cues to support the intervention components. One example may include pictures matched to emotions, to aid the understanding and communication of feelings. Other techniques to reinforce behaviours and actions may include role-play, storytelling and peerfeedback.
- (c) Consider the following during intervention development and implementation.
 - (i) Interventions identified in this review benefitted from clear communication and coordination between different groups involved. For instance, interventions in schools were strengthened when teachers and parents worked together.
 - (ii) Conduct feasibility studies and pilot studies, where possible, in order to make any necessary amendments early in the implementation process. This will also help inform others in the future.
 - (iii) Providing a range of interventions within a service or facility (such as a school), where this is feasible, may provide the best model with which to support a personalised approach for each child.
- (d) Consultation and coordination may help promote feasible, acceptable and sustainable intervention programmes.
 - (i) Consult the children, their parents, mental health and hearing specialists, and the delivery agents (such as teachers). Understand their needs and context in the adaptation or development phase.
 - (ii) In all, talk to the children who are the intended target of the intervention about their own needs and preferences for intervention components, including communication methods.

Downloaded from https://www.cambridge.org/core. 06 Dec 2021 at 10:38:33, subject to the Cambridge Core terms of use

Nathaniel Scherer , international Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, UK; Tess Bright, International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, UK; David John Musendo, International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, UK; Timothy O'Fallon, International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, UK; Chris Kubwimane, DeafReach, UK; Julian Eaton ©, Centre for Global Mental Heath, London School of Hygiene & Tropical Medicine, UK; Ritsuko Kakuma, Centre for Global Mental Heath, London School of Hygiene & Tropical Medicine, UK; Ritsuko Kakuma, Centre for Global Mental Heath, London School of Hygiene & Tropical Medicine, UK; Tracy Smythe, International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, UK; Sarah Polack, International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, UK

Correspondence: Nathaniel Scherer. Email: nathaniel.scherer@lshtm.ac.uk First received 20 Jul 2021, final revision 20 Sep 2021, accepted 11 Oct 2021

Supplementary material

Supplementary material is available online at https://doi.org/10.1192/bjo.2021.1045

Data availability

Data availability is not applicable to this article as no new data were created or analysed in this study.

Acknowledgements

The search string for this review was compiled with support from an Information Specialist at the London School of Hygiene & Tropical Medicine.

Author contributions

N.S., T.B., C.K., J.E., R.K., T.S. and S.P. conceptualised the research question, design and methodology. N.S., T.B., D.J.M. and T.O.F. acquired, analysed and interpreted the data, with support from all authors. S.P. provided review of the quality appraisal. R.K., T.S. and S.P. provided supervision throughout. N.S. drafted the manuscript, with input and critical review from all authors. All authors read and approved the final manuscript.

Funding

This work was supported by CBM International and BMZ (Bundesministerium für wirtschaftliche Zusammenarbeit und Entwicklung) under the project 'Improving the psychosocial resilience of children with and without disabilities in the Gaza Strip'. Funding for the Covidence systematic review software was provided through the PENDA project (Programme for Evidence to Inform Disability Action) funded by the UK's Foreign Commonwealth and Development Office.

Declaration of interest

None.

References

- 1 Haile LM, Kamenov K, Briant PS, Orji AU, Steinmetz JD, Abdoli A, et al. Hearing loss prevalence and years lived with disability, 1990–2019: findings from the Global Burden of Disease Study 2019. *Lancet* 2021; 397: 996–1009.
- 2 Briscoe J, Bishop DV, Norbury CF. Phonological processing, language, and literacy: a comparison of children with mild-to-moderate sensorineural hearing loss and those with specific language impairment. J Child Psychol Psychiatry 2001: 42: 329–40.
- 3 Calderon R, Greenberg M. Social and Ernotional Development of Deaf Children: Family, School, and Program Effects. Oxford University Press, 2011.
- 4 Ariapooran S. Relationship between mother- child interaction and symptoms of anxiety disorders in deaf and hard-of-hearing children. *Empowering Except Child* 2018; 9: 51–63.
- 5 Gentili N, Holwell A. Mental health in children with severe hearing impairment. Adv Psychiatr Treat 2018; 17: 54–62.
- 6 Lawrence BJ, Jayakody DMP, Bennett RJ, Eikelboom RH, Gasson N, Friedland PL. Hearing loss and depression in older adults: a systematic review and meta-analysis. *Gerontologist* 2020; 60: e137–54.
- 7 Shoham N, Lewis G, Favarato G, Cooper C. Prevalence of anxiety disorders and symptoms in people with hearing impairment: a systematic review. Soc Psychiatry Psychiatr Epidemiol 2019; 54: 649–60.
- 8 Theunissen SC, Rieffe C, Kouwenberg M, De Raeve L, Soede W, Briaire JJ, et al. Anxiety in children with hearing aids or cochlear implants compared to normally hearing controls. *Laryngoscope* 2012; **122**: 654–9.
- 9 Theunissen SC, Rieffe C, Kouwenberg M, Soede W, Briaire JJ, Frijns JH. Depression in hearing-impaired children. Int J Pediatr Otorhinolaryngol 2011; 75: 1313–7.

- 10 Stevenson J, Kreppner J, Pimperton H, Worsfold S, Kennedy C. Emotional and behavioural difficulties in children and adolescents with hearing impairment: a systematic review and meta-analysis. *Eur Child Adolesc Psychiatry* 2015; 24: 477–96.
- 11 Bigler D, Burke K, Laureano N, Alfonso K, Jacobs J, Bush ML. Assessment and treatment of behavioral disorders in children with hearing loss: a systematic review. Otolaryngol Head Neck Surg 2019; 160: 36–48.
- 12 Kessler RC, Berglund P, Demler O, Jin R, Merikangas KR, Walters EE. Lifetime prevalence and age-of-onset distributions of DSM-IV disorders in the National Comorbidity Survey Replication. Arch Gen Psychiatry 2005; 62: 593–602.
- 13 Clark C, Rodgers B, Caldwell T, Power C, Stansfeld S. Childhood and adulthood psychological ill health as predictors of midlife affective and anxiety disorders: the 1958 British Birth Cohort. Arch Gen Psychiatry 2007; 64: 668–78.
- 14 Goodman A, Joyce R, Smith JP. The long shadow cast by childhood physical and mental problems on adult life. Proc Natl Acad Sci 2011; 108: 6032.
- 15 Correll CU, Cortese S, Croatto G, Monaco F, Krinitski D, Arrondo G, et al. Efficacy and acceptability of pharmacological, psychosocial, and brain stimulation interventions in children and adolescents with mental disorders: an umbrella review. World Psychiatry 2021; 20: 244–75.
- 16 Munn Z, Peters MDJ, Stern C, Tufanaru C, McArthur A, Aromataris E. Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach. BMC Med Res Methodol 2018; 18: 143.
- 17 Tricco AC, Lillie E, Zarin W, O'Brien KK, Colquhoun H, Levac D, et al. PRISMA extension for scoping reviews (PRISMA-ScR): checklist and explanation. Ann Intern Med 2018; 169: 467–73.
- 18 Arksey H, O'Malley L. Scoping studies: towards a methodological framework Int J Soc Res Methodol 2005; 8: 19–32.
- 19 Moola S, Munn Z, Tufanaru C, Aromataris E, Sears K, Sfetcu R, et al. Chapter 7: Systematic reviews of etiology and risk. In JBI Manual for Evidence Synthesis (eds E Aromataris and Z Munn). JBI, 2020.
- 20 Ahmadi H, Daramadi PS, Asadi-Samani M, Givtaj H, Sani MRM. Effectiveness of group training of assertiveness on social anxiety among deaf and hard of hearing adolescents. Int Tinnitus J 2017; 21: 14–20.
- 21 Altshuler KZ, Spady F. The emotionally disturbed deaf child: a first program of research and therapy. *J Commun Disord* 1978; **11**: 171–86.
- 22 Anonymous. Gold award: mental health services for the deaf–St. Elizabeths Hospital, Washington, D.C. Hosp Community Psychiatry 1978; 29: 674–7.
- 23 Ashori M, Najafi F. Emotion management: life orientation and emotion regulation in female deaf students. J Child Fam Stud 2021; 30: 723–32.
- Bernstein V, Denno LS. Repetitive behaviors in CHARGE syndrome: differential diagnosis and treatment options. *Am J Med Genet*2005; 133: 232–9.
 Boham BE, Selkowitz S, Video-counseling for multi-handicapped hearing-
- impaired students. Am Ann Deaf 1981; **126**: 587–90.
- 26 Borowiec J, Hökelmann A, Osiński W. The level of self-esteem of deaf children: can participating in dance lessons with vibrational headphones improve it? Arts Psychother 2019; 64: 34–8.
- 27 Burnes S, Seabolt D, Vreeland J. Deaf culturally affirmative programming for children with emotional and behavioral problems. JADARA 2019; 26: 12–17.
- 28 Chapel SL. Child-Centered Play Therapy with Deaf Children: Exploring Linguistic and Cultural Implications (Counselor Education Master's Theses). The College at Brockport: State University of New York, 2005.
- 29 Donovan AL. Functional Communication Training in Young Deaf Children: Effects of Reinforcement Magnitude on Requesting and Time-on Task (PhD Doctoral Thesis). Faculty of the Graduate School, University of Minnesota, 2003.
- 30 Dursun OB, Erhan SE, Ibiş E, Esin IS, Keleş S, Şirinkan A, et al. The effect of ice skating on psychological well-being and sleep quality of children with visual or hearing impairment. *Disabil Rehabil* 2015; 37: 783–9.
- 31 Elkayam J, English K. Counseling adolescents with hearing loss with the use of self-assessment/significant other questionnaires. J Am Acad Audiol 2003; 14: 485–99.
- 32 Forehand R, Cheney T, Yoder P. Parent behavior training: effects on the noncompliance of a deaf child. J Behav Ther Exp Psychiatry 1974; 5: 281–3.
- 33 Garcia R, Turk J. The applicability of Webster-Stratton Parenting Programmes to deaf children with emotional and behavioural problems, and autism, and their families: annotation and case report of a child with autistic spectrum disorder. *Clin Child Psychiatry* 2007; 12: 125–36.
- 34 Greenberg MT, Kusché CA. Preventive intervention for school-age deaf children: the PATHS curriculum. J Deaf Stud Deaf Educ 1998; 3: 49–63.
- 35 Hatamizadeh N, Adibsereshki N, Kazemnejad A, Sajedi F. Randomized trial of a resilience intervention on resilience, behavioral strengths and difficulties of mainstreamed adolescent students with hearing loss. Int J Pediatr Otorhinolaryngol 2020; 128: 109722.

Downloaded from https://www.cambridge.org/core. 06 Dec 2021 at 10:38:33, subject to the Cambridge Core terms of use.

Scherer et al

- 36 Johnson R, Sandberg K. Drug Abuse Resistance Education (D.A.R.E): special consideration for students who are deaf or hard of hearing. JADARA 1992; 26: 9–11.
- 37 Lasanen M, Määttä K, Uusiautti S. 'I am not alone' an ethnographic research on the peer support among northern-Finnish children with hearing loss. *Early Child Dev Care* 2019; 189: 1203–18.
- 38 Nehra A, Mann SB, Sharma SC, Verma SK. Effect of hearing AIDS on psychosocial dysfunctioning. Indian J Otolaryngol Head Neck Surg 2001; 53: 291–4.
- Osborne JG. Installing and maintaining a behavior modification system in a school for the deaf. *Rehabil Psychol* 1977; 24: 1–58.
- 40 Sarti DM. Reaching the deaf child: a model for diversified intervention. Smith Coll Stud Soc Work 1993; 63: 187–98.
- 41 Shinn MM. Parent-child interaction therapy with a deaf and hard of hearing family. *Clin Case Stud* 2013; 12: 411–27.
- 42 Sullivan PM, Scanlan JM, Brookhouser PE, Schulte LE, Knutson JF. The effects of psychotherapy on behavior problems of sexually abused deaf children. *Child Abuse Negl* 1992; 16: 297–307.
- 43 Tinsley K, Jedlicka H. Equine Assisted Counseling With Deaf Families. In Harnessing the Power of Equine Assisted Counseling: Adding Animal Assisted Therapy to Your Practice (ed. K Trotter). Routledge, 2012.
- 44 Troester JD. A therapeutic play group for children with hearing impairments. J Child Adolesc Group Ther 1996; 6: 101–9.
- 45 Vreeland J, Tourangeau J. Culturally affirmative residential treatment services for deaf children with emotional and behavioral disorders. In Mental Health Care of Deaf People: A Culturally Affirmative Approach (eds Glickman NS, Gulati S): 239–60. Lawrence Erlbaum Associates Publishers, 2003.
- 46 Wright B, Walker R, Holwell A, Gentili N, Barker M, Rhys-Jones S, et al. A new dedicated mental health service for deaf children and adolescents. Adv Ment Health 2012; 11: 95–105.

- 47 Petersen I, Evans-Lacko S, Semrau M, Barry MM, Chisholm D, Gronholm P, et al. Promotion, prevention and protection: interventions at the population- and community-levels for mental, neurological and substance use disorders in low- and middle-income countries. Int J Ment Health Syst 2016; 10: 30.
- 48 Fazel M, Hoagwood K, Stephan S, Ford T. Mental health interventions in schools 1: mental health interventions in schools in high-income countries. *Lancet Psychiatry* 2014; 1: 377–87.
- 49 Schuelka M, Johnstone CJ. Global trends in meeting the educational rights of children with disabilities: from international institutions to local responses. *Reconsidering Dev* 2012; 2: 1–14.
- 50 United Nations Educational Scientific and Cultural Organization. Towards Inclusion in Education: Status, Trends and Challenges. UNESCO, 2020.
- 51 Kuenburg A, Fellinger P, Fellinger J. Health care access among deaf people. J Deaf Stud Deaf Educ 2015; 21: 1–10.
- 52 Rosenberg T. Depressed? Here's a Bench. Talk to Me. New York Times. 22 Jul 2019.
- 53 Patel V, Sartorius N. From science to action: the Lancet Series on Global Mental Health. Curr Opin Psychiatry 2008; 21: 109–13.
- 54 Patel V, Saxena S, Lund C, Thornicroft G, Baingana F, Bolton P, et al. The Lancet Commission on global mental health and sustainable development. *Lancet* 2018; 392: 1553–98.
- 55 World Bank, World Health Organization. The World Report on Disability. WHO, 2011.



10

Do

nloaded from https://www.cambridge.org/core. 06 Dec 2021 at 10:38:33, subject to the Cambridge Core terms of use

PLOS GLOBAL PUBLIC HEALTH

RESEARCH ARTICLE

Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip

Nathaniel Scherero^{1*}, Tracey Smythe^{1,2}, Ramadan Hussein³, Lorraine Wapling⁴, Shaffa Hameed¹, Julian Eaton^{5,6}, Naim Kabaja³, Ritsuko Kakuma⁵, Sarah Polack¹

 International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Division of physiotherapy, Department of Health and Rehabilitation Sciences, Stellenbosch University, Stellenbosch, South Africa, 3 Atfaluna Society for Deaf Children, Gaza City, Palestinian Territories, 4 International Disability Research Centre, University College London, London, United Kingdom, 5 Centre for Global Mental Health, London School of Hygiene & Tropical Medicine, London, United Kingdom, 6 CBM Global Disability Inclusion, London, United Kingdom

* nathaniel.scherer@lshtm.ac.uk



OPEN ACCESS

Citation: Scherer N, Smythe T, Hussein R, Wapling L, Hameed S, Eaton J, et al. (2023) Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip. PLOS Glob Public Health 3(6): e0001635. https://doi.org/ 10.1371/journal.poph.0001635

Editor: Julia Robinson, PLOS: Public Library of Science, UNITED STATES

Received: December 5, 2022

Accepted: April 19, 2023

Published: June 6, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgph.0001635

Copyright: © 2023 Scherer et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Excerpts from the transcripts of this qualitative research have been made available within the article. Providing

Abstract

Deaf and hard of hearing children are at risk of exclusion from community life and education, which may increase their risk of mental health conditions. This study explores the experience of deaf and hard of hearing children in the Gaza Strip, with particular focus on the factors that contribute to psychological wellbeing and distress. In-depth interviews were conducted with 17 deaf and hard of hearing children, 10 caregivers of deaf and hard of hearing children and eight teachers of deaf and hard of hearing children in mainstream and special schools, across the Gaza Strip. Further, three focus group discussions were held with deaf and hard of hearing adults and disability leaders, mental health specialists and other teachers of deaf and hard of hearing children. Data collection was completed in August 2020. Key themes identified in the analysis included lack of accessible communication, community exclusion, negative attitudes towards hearing impairment and deafness and the impact on deaf and hard of hearing children's sense of self, and limited family knowledge on hearing impairment and deafness. Further findings focused on strategies to improve the inclusion of deaf and hard of hearing children and how to promote wellbeing. In conclusion, participants in this study believed that deaf and hard of hearing children in the Gaza Strip are at increased risk of mental health conditions. Changes are needed across community and government structures, including education systems, to promote the inclusion of deaf and hard of hearing children and to support their psychological wellbeing. Recommendations from the findings include increasing efforts to improve awareness and reduce stigma, providing better access to sign language for deaf and hard of hearing children, and offering training for teachers of deaf and hard of hearing children, especially in mainstream environments.

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0001635 June 6, 2023

1/20

anonymised excerpts (i.e. quotations) in publiclyavailable publications was confirmed by participants during the informed consent process. However, full-transcripts are not available via a public data repository. Sharing full-transcripts, even with known identifiers removed, was not approved by the ethics committees and was not confirmation sought from participants during the informed consent procedure. For queries, please contact the Corresponding Author, or the LSHTM Research Ethics Committee via email (ethics@lishtm.ac.uk).

Funding: This research was funded by the German Federal Ministry for Economic Cooperation and Development (BMZ; Bundesministerium für wirtschaftliche Zusammenarbeit und Entwicklung) through CBM International under the project "Improving the psychosocial resilience of children with and without disabilities in the Gaza Strip" to NS, SP, JE, RH and NK. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. Authors NS and RH received salary through this funding for the research undertaken.

Competing interests: The authors have declared that no competing interests exist.

Introduction

One in five people experience hearing impairment, of whom more than 80% live in low- and middle-income countries (LMICs) [1]. Whilst hearing impairment is more common in older persons, estimates indicate that there are 70 million children with hearing impairment globally, of whom 34 million have moderate to profound hearing impairment [1].

Deaf or hard of hearing children across the world can experience challenges in daily life, with regards to educational attainment, social inclusion and community participation [2–4]. Deaf and hard of hearing children are more likely than hearing peers to report poorer quality of life in school and social domains [3]. These negative life experiences are associated with a negative impact on emotional wellbeing and mental health, and there is increased risk of mental health conditions among deaf and hard of hearing adult and child populations, including anxiety and depression [5–13].

Context and aim of the study

Little research exists on the experiences of deaf and hard of hearing children in LMICs. The vast majority of research comes from high-income settings and findings are not representative of LMICs, where there exist different conceptions and attitudes towards disability and hearing impairment, and often more limited availability of specialised support and services. Further, little evidence is available on the experiences of deaf and hard of hearing in conflict-affected settings, where, in general, people are at greater risk of mental health conditions [14]. Deaf and hard of hearing children in conflict-affected settings may face double-jeopardy to their psychological wellbeing, both from the stressors experienced by many deaf and hard of hearing children and those that arise from being affected by conflict, such as trauma and lack of access to services [15, 16].

This study explores the experiences of deaf and hard of hearing children in the Gaza Strip, an area that has been affected by conflict and blockade, resulting in a deteriorating socioeconomic situation that has negatively impacted on the economy, education, food security, and access to basic services [17]. There is some evidence to suggest that the Gaza Strip and the West Bank has a high prevalence of hearing impairment among infants and children [18, 19]. Historically, this has resulted from high instances of consanguinity, although this practice is becoming less common [18]. However, there is scarce research available on the lived experience of deaf and hard of hearing children in the Gaza Strip, including information on experiences in school and the wider community.

Ongoing conflict, economic hardship and restricted life prospects are associated with high rates of depression, anxiety and post-traumatic stress disorder for youth in the Gaza Strip [20–25]. These stressors make it difficult for children and adolescents to develop a strong sense of self and to develop aspirations for the future [25]. Despite mental health services being available in the Gaza Strip, many young people face organisational and cultural barriers to access, and services often do not meet their need [20, 25]. Nevertheless, research has suggested that education and family support have improved wellbeing of youth in the Gaza Strip, providing inspiration and building self-esteem [25]. Although there is evidence on mental health among youth generally, there is limited information on factors that contribute to psychological wellbeing among deaf and hard of hearing children in the Gaza Strip.

Given the dearth of evidence on deaf and hard of hearing children in the Gaza Strip, this study aimed to investigate the experiences of deaf and hard of hearing children in daily life and the factors that contribute to their inclusion, quality of life and mental health. This information can help contribute to the development of context appropriate policies and interventions.

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0001635 June 6, 2023

2/20

Materials and methods

This study used qualitative research methods to explore the experiences of deaf and hard of hearing children in the Gaza Strip. In-depth interviews were conducted with 35 participants, including 17 primary school age deaf and hard of hearing children, 10 caregivers, and eight teachers of deaf and hard of hearing children in mainstream and special schools. Three focus group discussions were held with five deaf and hard of hearing adults and disability leaders, seven mental health and psychosocial support specialists and school counsellors, and 13 teachers of deaf and hard of hearing children in mainstream and special education settings.

The research and analysis were guided by a steering committee of key stakeholders in the Gaza Strip. This group comprised: five deaf and hard of hearing people and representatives from organisations of deaf and hard of hearing people; two caregivers of deaf and hard of hearing children; four teachers of deaf and hard of hearing children; three mental health and psychosocial support specialists; and five representatives from government level institutions, including the Ministry of Education.

Study context

The research was conducted across each region of the Gaza Strip. In the Gaza Strip, deaf children with severe to profound hearing impairment are typically taught in special schools. Hard of hearing children with mild to moderate hearing impairment are typically taught in mainstream schools. Mainstream schools are government run and free to attend. Special schools are not supported by government financing and are often managed by non-governmental organisations (NGOs). Families pay a fee for their child to attend a special school. Each mainstream and special school is required to have a school counsellor who is tasked with promoting mental wellbeing and supporting children experiencing distress. When a child presents with severe symptoms of psychological distress, they are referred by the school to counsellors at district level who provide more in-depth assessment and support. At this district level, there are mental health specialists trained to work with deaf and hard of hearing children.

Participants

A summary of participant characteristics is presented in S1 Appendix.

Participants for the in-depth interviews were purposively sampled through mainstream and special schools across four regions in the Gaza Strip (North, Gaza City, Middle, South) in order to maximise variation, as based on Patton's maximum variation sampling strategy [26]. Participants were selected for variation across age, sex, severity of hearing impairment and type of school attended (mainstream or special school). To aid the sampling strategy, we segmented the target group by the criteria listed above, aiming for equal representation in each category. From this, we developed a minimum sample size of 32, with additional participants included if topics required further investigation [26]. In total, 35 participants were included in the in-depth interviews, comprised of 17 deaf and hard of hearing children aged 6–12, 10 caregivers of deaf and hard of hearing children, and eight teachers of deaf and hard of hearing children in mainstream and special schools. Severity of hearing impairment was confirmed by an audiologist. Hearing profiles, assistive technology usage and communication mode of the deaf and hard of hearing children are available in S1 Appendix.

Participants for the three focus group discussions were recruited through local organisations of persons with disabilities (OPDs), disability NGOs, mental health associations and schools. The first group comprised of three deaf and hard of hearing adults and sign language users, one individual with a physical impairment and one individual with a visual impairment, who also works as a counsellor. These individuals were representatives from OPDs and NGOs

supporting deaf and hard of hearing people. The second group comprised five school counsellors and two mental health professionals working with children in the community. The third included seven special school teachers and six mainstream school teachers. These teachers were not the same as those included in the in-depth interviews.

Data collection

Data collection was conducted in July 2020. Interviews were held in a trusted, central location, as requested by participants, many of whom expressed concern about being audio-recorded in their own home. As the research was conducted during the COVID-19 pandemic, government guidance on safety was followed.

Semi structured, in-depth interviews with deaf and hard of hearing children, caregivers and teachers sought to explore the views, experiences, emotions and perspectives of deaf and hard of hearing children and caregivers in the Gaza Strip, both in school and the wider community. In addition, focus group discussions were held with individuals who have expertise in understanding the experiences and needs of deaf and hard of hearing children and their families. Through dynamic group discussion, these focus groups aimed to elicit multiple opinions on the experiences of deaf and hard of hearing children, as well as recommendations to improve quality of life and psychological wellbeing. Topic guides are available in S2 Appendix.

The interview guides were developed by the research team, piloted with six participants (two children, two caregivers, two teachers) and adapted as needed thereafter. Questions explored the experiences of deaf and hard of hearing children, their families and their teachers, focusing on experiences in education, healthcare and community life. Specific questions were asked on communication, mental health, wellbeing and attitudes towards hearing impairment and deafness. Topic guides for the focus group discussions aimed to further understand these areas, explore the wider context for deaf people in the community, and appraise the factors contributing to wellbeing. Data was collected during the COVID-19 pandemic, however the study sought to understand experiences in more typical times, and we did not explicitly explore the experiences of deaf and hard of hearing children and families in the context of the pandemic.

Interviews were conducted in Arabic with those using oral communication. Palestinian sign language interpretation was available for those using sign language for communication. Emotion cards were used to help children understand a question or to describe their feelings. Children were interviewed individually, unless the child or caregiver desired otherwise. There was occasionally need for a family member to sit in as interpreter, when a child communicated through a signed language developed with family at home. Each interview lasted between 30–60 minutes. Five pilot interviews were conducted, in which to practice interviewing techniques and refine the topic guides. The data from these interviews has been included in the analysis. All interviews were audio-recorded. Sign language users had their response reported orally by an interpreter. Audio-recordings were transcribed verbatim. Each was translated and transcribed directly into English, by an independent transcriber, and subsequently checked against the audio file by the interviewer. Each transcript was anonymised and stored on a secure server.

Data analysis

Data was analysed using thematic analysis [27]. In step one, NS familiarised themself with the data, noting down initial ideas for coding. In step two, NS developed a coding framework, which was discussed and refined with RH, TS and SP. NS subsequently coded transcripts in NVivo 12 using this coding framework, which was piloted and iteratively adapted throughout

analysis. In step three, emerging themes were identified and refined by NS, comparing relationships between codes and across groups. In step four and five, NS reviewed these themes with RH, RK, TS and SP, mapping the themes against the data set and the emerging narrative. Where RH, RK, TS and SP had comments and queries, NS reviewed the coding and themes, refine and recategorizing codes, where needed. In the final step, participant quotes and narratives were extracted for the report. The themes, narrative and recommendations were discussed with the steering committee, before producing the final report.

Reflexivity

Interviews were conducted by a trained interviewer (RH), who received a three-day, one-toone training with the lead author (NS), a qualitative researcher in global disability and mental health. The interviewer's background was as an experienced audiologist at a major service unit for deaf and hard of hearing people in the Gaza Strip; Atfaluna Society for Deaf Children. The interviewer knew some local sign language, but not enough to conduct interviews without support from an interpreter. The interviews themselves were held at Atfaluna's headquarters, at the request of participants. With Atfaluna known to many deaf and hard of hearing people and their families in the Gaza Strip, their responses may have been influenced by the dynamic. To mitigate this, we assured participants that their responses and data would be kept confidential and that their access to services would not be affected as a result of the interview. For many, Atfaluna offers a safe space for deaf and hard of hearing children and families, and in many instances, the interviewer reported the benefit of this association on building comfort and rapport.

The analysis was led by NS, who is based in the United Kingdom. NS supported the data collection remotely through co-training, regular post-interview discussions with the interviewer and remote participation in workshops/advisory committees as travel to Gaza strip was not possible due to the COVID-19 pandemic. To limit potential biases in analysis led by an off-site foreign researcher, the interpretations were reviewed by the research team in the Gaza Strip with experience in supporting deaf and hard of hearing children in the region. The findings were further discussed with the steering committee in the Gaza Strip.

Reliability and validity

To promote reliability, validity and rigour of this qualitative research, we adopted a number of methodological strategies, as recommended in the literature [28, 29]: (1) Reflexivity, as we've described above, our efforts to reflect on our position as researchers sought to provide transparency on our biases and the influence of their on the analysis; (2) Validation of findings was achieved by presenting our initial analysis to the steering committee and groups of deaf and hard of hearing adults, to ensure that our interpretation of the findings were appropriate to the context in the Gaza Strip; (3) Peer debriefing was utilised throughout coding and analysis between NS and co-authors, to ensure exploration into all relevant themes; (4) Triangulation was core to the methodology, with authors adopting multiple data sources and data collection methods, to strive for a comprehensive exploration of this topic; (5) Transparency of procedures was achieved by keeping a paper trail of methodology, records and explanations, and presenting these in the manuscript.

Ethical considerations

Ethical approval for this study was obtained from the Research Ethics Committee at the London school of Hygiene & Tropical Medicine (19144) and the Palestinian Health Research Council (PHRC/HC/697/20).

Informed written consent was received from all participants. Caregivers or guardians provided informed consent for children and further assent was sought from children using a simplified information sheet.

Results

Across the interviews, there was the consistent understanding from adult participants that deaf and hard of hearing children in the Gaza Strip were at increased risk of psychological distress and mental health concerns. Various associated factors were reported, including language deprivation, community attitudes and family support.

Lack of accessible communication and language deprivation

Participants reported that deaf and hard of hearing children with milder levels of hearing impairment were less likely to experience challenges with communication and community participation, and they were often able to communicate orally. However, deaf and hard of hearing children with moderate to profound hearing impairment faced daily challenges with communication and engagement with others in the community as sign language use is limited across the region. Communication challenges restricted social inclusion and participation, negatively impacting on the psychological wellbeing of deaf and hard of hearing children.

Although Palestinian sign language use is increasing across the Gaza Strip, it is still limited in healthcare, mainstream schools and community settings. Teachers, caregivers and deaf and hard of hearing adults told us how many deaf and hard of hearing children experience language deprivation, with families unable to teach them sign language or refusing to allow their child to learn.

"I: Do they use sign language?

P: No it is not allowed... once her uncle tried to use sign language and I prevented him from doing so. If I wanted her to learn and use sign language, I would give her sign language courses."

(Father of a child aged 10-12 with severe hearing impairment)

We were told that many families in the Gaza Strip do not know Palestinian sign language and do not use it at home, even if their child uses the language at school and despite many of the children interviewed telling us that they prefer to communicate using sign language. Many of the families interviewed perceived sign language negatively and desired their child to use oral speech, even when the child had difficulty doing so. Some attributed sign language for their child's poor oral skills and some did not understand it to be a language itself and their child's natural form of communication. Limited sign language use and support resulted in examples of family and child having limited communication or regular misunderstanding and miscommunication.

"If their cousins gather and talk, she will withdraw because she cannot hear and she cannot understand what they are saying, as they are talking fast. The same thing when my sisters and I gather. This thing bothers me and bothers her, but what we can do?"

(Mother of a child aged 10-12 with moderate hearing impairment)

The lack of engagement with sign language was not the case for all families, and there were positive examples of caregivers supporting the use of sign language by their child and seeking

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0001635 June 6, 2023

6/20

training for themselves. After learning themselves, there were examples told to us of family members training others, including hearing siblings, aunts, uncles and cousins. Support for sign language and accessible communication was often seen in families with multiple deaf and hard of hearing children, where siblings could communicate with and support one another. One child interviewed even taught their hearing friends at a mainstream school some simple signs, so they could better communicate and bond. However, despite wishing to support sign language use, some family members reported limited awareness on where to receive sign language training and communication support. Some children used home sign language as a result of limited exposure to Palestinian sign language and can only communicate with a few others as a result; often deaf and hard of hearing siblings from whom they learned to communicate.

Communication and social inclusion

Some children interviewed had strong oral skills. They found it easier to participate in community life and were integrated into life at a mainstream school. That said, many of these children still expressed a preference for sign language, because oral communication was often tiring and difficult to maintain, especially in group situations and noisy environments.

Many deaf and hard of hearing children in mainstream schools described difficulty in keeping up with teachers and peers, who often have limited awareness of deaf and hard of hearing children's communication needs and how best to support them. We were told by many, including teachers, that difficulty with communication in mainstream schools significantly hampered the learning and education for many deaf and hard of hearing children. Limited communication also makes inclusion difficult with hearing peers in school and the community. This was reported to cause isolation and psychological distress. Some caregivers told us that they do not allow their deaf or hard of hearing child out into the community because of their difficulties communicating with others.

"His nature is sociable, he tries to communicate with other kids and join them, but they cannot understand what he is saying, which makes him upset."

(Father of a child aged 6–9 with moderate hearing impairment)

In contrast, the special school environment was more supportive of sign language and children and teachers in these settings reported very few instances of miscommunication or isolation of deaf children. A deaf adult told us how, as a child, he had to move back to a special school after joining a mainstream school, as communication in the mainstream environment was too challenging and teachers didn't account for his deafness and communication needs. For example, he lipreads and teachers regularly faced away from him when talking. He expressed regret that he had to leave, as he wanted to learn in a mainstream environment.

Language deprivation, challenges in communication and resulting social isolation were noted by many of the participants as the major influence on a child's mental health and wellbeing.

I: Are deaf children at risk of mental health issues?

P: Yes, because they cannot keep up with the family talking. What's causing a psychological problem is that a family will shorten up a long conversation into two words. This upsets a child and makes them prefer loneliness.

(Deaf adult and representative from an organisation of persons with disabilities, communicating via a sign language interpreter) Participants, including deaf adults, teachers and mental health specialists, discussed communication in the context of protecting and promoting mental health and wellbeing. Communication and the resulting social integration were described as the main source of a child's confidence, self-esteem and wellbeing. Participants recommended that families and teachers be given sign language training to combat the language deprivation experienced by many deaf and hard of hearing children in the Gaza Strip and to promote their psychological wellbeing.

"When the student has good communication in sign language with their family members, this comforts and relieves them very much. When the mother communicates in sign language with her child, the child becomes able to express and explain everything. The same thing happens when they communicate with their teacher, they feel that they are just as other classmates. When they go on a school journey and they communicate with teachers and other people, they break the ice and overcome the fear with outside. As a result, they become more confident... This gradually strengthens their personalities and their selfconfidence."

(Teacher in a special school)

Deaf adults told that us that communication was improving in the Gaza Strip, with Palestinian sign language more widely used than when they were children. However, improvements in the attitude towards and understanding of sign language was slow, and many families were still resistant to its use. Services and programmes to support use of sign-language, as well as speech and language therapy and occupational therapy, were often unavailable to caregivers, except through one or two specialist organisations. Foremost was Atfaluna Society for Deaf Children, which was often cited as the primary support for deaf children in the Gaza Strip, although not all participants were aware of the available support from this organisation and other specialist organisations supporting deaf and hard of hearing children. They did not have information on support available, where to seek support and how to access it. Deaf adults also told us that it was difficult to find qualified sign language interpreters in the region and sign language training was sparse. The services were not sufficient to meet the need.

"Frankly, the Department of Education hasn't given us sign language courses. I use simple, easy things for students. For example, go, come, and understand. . . I had four students with hearing disabilities and I couldn't communicate well with them."

(Teacher in a mainstream school)

Assistive technology and communication

Assistive technology was commonly discussed in the context of communication. Many caregivers described hearing aids as useful for children who use them and many wanted their child to use one, even if the improvements in hearing were marginal. However, some caregivers and children spoke of ill-functioning or ill-fitting hearing aids, negating any benefits for communication and limiting their ability to engage in community and classroom activities. Caregivers told us that paying for and maintaining the hearing aids is expensive, with some caregivers forgoing food to afford batteries. Repair services are also limited and there can be long waitlists if a hearing aid is damaged, especially if originally sourced from a country out of the Gaza Strip. Cochlear implants were perceived positively among the participants, with some having gone abroad to receive them, to countries such as Egypt and Australia. However, this was not available for all families of deaf and hard of hearing children. Many of those that had done so still

relied on financial support from an NGO or international sponsor. Cochlear implantation is becoming more widely available in the Gaza Strip and many families note this as an appealing option to them. Deaf adults interviewed, however, were often disparaging of such assistive technologies and encouraged families to embrace their child's hearing impairment and to cement a strong Deaf identity in them.

Community attitudes impacting confidence and identity

Stigma and discrimination in the community also presented barriers to social inclusion and participation for deaf and hard of hearing children, and had a negative impact on psychological wellbeing. This was not the case for all deaf and hard of hearing children and adults, and some reported positive relationships with neighbours and hearing peers. However, we were told that many in the community still held discriminatory views and behaviours. Awareness of disability and deafness was said to be poor. Knowledge was often based on a medical or charity model of disability and deafness, with little understanding of rights-based perspectives in line with the United Nations Convention on the Rights of Persons with Disabilities (UNCRPD) [30]. This limited awareness was seen most commonly among mainstream school teachers, who often had very little knowledge of deafness and supporting deaf and hard of hearing children in their classroom.

Respondents felt that it was common for the community to look on families and deaf and hard of hearing children with sadness and pity, and we were told that deaf people were seen to be unequal or were simply not acknowledged in society.

"For example, if we are in a car, people will stare at him [her deaf son], and some of them ask, 'Why is he like this?'. Someone said, 'Poor boy'. I said, 'Why? What's wrong with him [her deaf son]? Thank God, they are a gift'."

(Mother of child aged 6-9 with severe hearing impairment)

Participants reported that negative attitudes often proceed to more overt acts of discrimination. We were told that hearing people often use derogatory terms when talking about deaf people, including an Arabic slur which translates loosely to 'dumb' or 'retarded' (derogatory terms themselves across much of the world).

"They whisper and point at me that I'm dumb [Arabic slur]"

(Child aged 10-12 with severe hearing impairment)

Discrimination was said to severely impact on deaf and hard of hearing children (and adults) in the Gaza Strip. Children and caregivers told us of the psychological distress caused when community members or other children made them feel different or marginalised. Children told us they often cried as a result. There were several examples of such instances. A father told us how he moved his family to another region of the Gaza Strip, as his son was facing such severe discrimination; others used to steal his hearing aids and run away. Similarly, a mother told us how her daughter was bullied constantly at school.

"She was bullied at school by other children. They would remove her hearing aid and bully her. She would come home crying. I tried to speak to the teachers about this, but they didn't help. No one understands her communication or possibly needs. I decided to move her to another school but the same thing happened. I eventually moved her to a special school,
where things got better. She has since joined a mainstream government school after her communication improved."

(Mother of child aged 10-12 with moderate hearing impairment)

There were a few examples reported in which bullying and discrimination dissipated once caregivers and teachers explained deafness to other children, other teachers and community members. Some caregivers reported speaking about deafness to teachers individually, as well as presenting to other caregivers and children at the school. Some teachers with experience and knowledge about deafness instructed other teachers at their school, with support from the school counsellor and headteacher. These efforts were reported to improve awareness of the experience of deaf and hard of hearing children and helped dissipate some of the stigma and discrimination seen. Caregivers are important advocates in these initiatives.

"It was hard at the beginning. They mocked him every single day, which made him sad. He was crying all the time. I explained his case to the children and now they understand him and they started playing with him."

(Mother of child aged 6-9 in a mainstream school with moderate hearing impairment)

Special schools offered a more inclusive and supportive environment, with no discrimination reported to us amongst teachers or other children. However, there were no reports of systematic governmental or organisational training and resources for special school teachers or others to instruct mainstream teachers on supporting the inclusion of deaf and hard of hearing children in mainstream schools.

Further, few deaf and hard of hearing spaces and community structures were reported. Participants advocated for more deaf and hard of hearing friendly spaces, where deaf and hard of hearing children could learn about their hearing impairment, Deaf culture, sign language and be engaged with other deaf and hard of hearing people in a safe space. Participants also suggested that deaf and hard of hearing children need to be more included in community activities, to reduce isolation and improve psychological wellbeing.

"For us as deaf people, not more than 3% are included in society. At the present time, the community should make more effort to give deaf people their own space, and to involve them in community activities, so that people know more about the skills and experiences of deaf people."

(Deaf adult and representative from an organisation of persons with disabilities, communicating via a sign language interpreter)

Family knowledge and internalised stigma

It was clear that many caregivers and families interviewed were proud of their deaf and hard of hearing children and expressed to us deep love, but some had not fully accepted their child's hearing impairment and deafness. Caregivers were said to occasionally hide their deaf or hard of hearing child, so as not to affect their siblings' future and experience in the community. Others did not want to see their child wearing a hearing aid and refused to let them learn sign language. A deaf adult told us of a story in which a mother refused for her deaf son to marry a deaf woman, as she feared their child would also be deaf; she wanted him to marry a hearing woman instead.

Limited acceptance seemed to result from limited understanding and awareness of deafness and the rights of deaf and hard of hearing people, paired with the influence of negative community attitudes. Many of the caregivers interviewed had never received information or support on hearing impairment when their child was diagnosed, despite wishing for such. Caregivers called for structured training programmes with other caregivers, access to online resources and support from specialist organisations. Those that did receive such support reported positive benefits and typically presented with more positive attitudes to hearing impairment.

For some other caregivers, internalised stigma among families appeared to result from the challenges they perceive a result of having a deaf or hard of hearing child.

"It affects all my life system. It takes all of me, my efforts, my time, and it takes me from my other children."

(Mother of child aged 6-9 with severe hearing impairment)

As a result of negative community and family attitudes, some children told us that they did not want to be deaf or hard of hearing, or to be different, and many were embarrassed and ashamed to use sign language in public or to be seen wearing a hearing aid. One father said his daughter feared she would never get married because of her hearing aid.

"Once she said, 'I know why they are treating me like that, it is because I wear a hearing aid.' And one time I saw her crying because her sister got engaged and she said, 'I will never get married because of my hearing aid'."

(Father of child aged 10-12 with severe hearing impairment)

The discrimination in the community, and the internalised stigma among families and children, negatively impacts on deaf and hard of hearing children's sense of identity and ultimately their mental health and wellbeing. Children were said to get upset and angry at being different to others. One mother told us of how she enrolled her daughter in psychosocial counselling, as she was very down and tired of being "different".

Building a sense of self and wellbeing

Participants believed community awareness and improving understanding of deafness and hearing impairment was of the utmost importance in addressing stigma, discrimination and improving engagement with Deaf culture and norms, such as sign language. Addressing stigma and discrimination was said to be vital in improving the psychological wellbeing of deaf and hard of hearing children. Some told us that this awareness is continually improving in the Gaza Strip, particularly among the younger generation, who are more knowledgeable and supportive of deaf and hard of hearing people, as a result of greater exposure to the internet and international campaigns.

Children that engaged with deaf and hard of hearing peers and Deaf communities reported the positive impact on quality of life and psychological wellbeing. They enjoyed having deaf and hard of hearing friends and interacting with others with a shared experience. Many deaf and hard of hearing children in mainstream education said that they had few deaf and hard of hearing friends, although they would like to. Those that did have deaf and hard of hearing friends valued those relationships. "The first is to focus on things that help to integrate them into society, and the second is to spread culture about hearing disability among

children so that they accept deaf colleagues or any other disability without making fun of them."

(Adult with a mobility impairment and representative from an organisation of persons with disabilities)

Building a sense of self and building self-esteem was said to be key in promoting mental health and wellbeing among deaf and hard of hearing children. We were told that building a sense of identity and confidence in children starts with the family. Participants told us how important it was for families to learn about hearing impairment, deafness, deaf and hard of hearing rights, communication, sign language and how best to support their child. Many participants believed that deaf and hard of hearing children develop a strong sense of self and mental wellbeing when families are accepting and supportive of their children. Participants noted that caregivers must promote belief and expectation in their child.

Further, we were told of the importance of caregiver involvement in their child's education and life experience at school, especially in mainstream education. Doing so was considered to improve awareness and inclusion at schools, and caregivers expressed eagerness to work closely with teachers to ensure consistent communication and support.

Examples of good practice were seen in some special schools, including awareness raising and psychosocial support. This included sessions with caregivers on how to communicate with and support their deaf or hard of hearing child. Further, we were recommended that deaf and hard of hearing role models visit schools, both to raise awareness amongst the students on deafness and inclusion and to help build the self-esteem and aspirations of deaf and hard of hearing children and their families.

"We have a lot of attitudinal problems with adults and teenagers. We keep implementing awareness raising activities and psychosocial support sessions. These are very important to be included, especially as caregivers lack sign language, so they don't communicate with their children. Teachers manage the whole communication process and guide the students. We also implement awareness raising activities with caregivers to guide them on how to communicate with and treat their children. We advise them to talk to their children, to understand them, and to identify and analyse their problems."

(Teacher at a special school)

Promoting deaf and hard of hearing inclusion

As presented, participants reported a number of factors that impact on deaf and hard of hearing children's mental health and psychological wellbeing. Intervention and support is needed across a number of areas to promote wellbeing, including reducing stigma and improving communication. Deaf adults and teachers emphasised that the development of support for deaf and hard of hearing children and families at school and in the community is not going to happen without government support. According to the deaf adults we spoke with, inclusion at government level was viewed as talk with no action. We were told that this stemmed from limited understanding of hearing impairment and deafness at government level, which, in turn, contributed to the lack of inclusion and awareness in the community.

"Even the government decision makers know nothing about us [deaf people]. What we are, what we feel and what we need. They don't know us. There's no awareness of what disability

really is. They think we deaf people are less intelligent than others, they think our learning abilities are slow and hopeless. . . We, like all other disabilities, are no different and no less."

(Deaf adult and representative from an organisation of persons with disabilities, communicating via a sign language interpreter)

Deaf and hard of hearing children in the Gaza Strip were reported to be facing increased risk of mental health issues. Efforts are needed to build self-esteem and identity in deaf children, so that they may reach their potential and realise personal wellbeing.

"When a child with disability is left behind because of their disability, this would complicate their mental health and cause them lots of psychosocial problems. We need to improve their skills, look into their talents and work on developing them. This would improve their self-trust and help them overcome their mental health and psychosocial problems."

(School counsellor at a mainstream school)

For this to occur, families, teachers and the community need the understanding on how to support inclusion of deaf and hard of hearing children, which includes knowledge on communication and deaf culture.

Discussion

Participants in this study reported that deaf and hard of hearing children in the Gaza Strip are at increased risk of psychological distress, resulting from language deprivation, discrimination and exclusion from community participation. These findings are consistent with previous literature on the social inclusion and mental health of deaf and hard of hearing populations [5–13].

Children in special school environments commonly used sign language and engaged with deaf and hard of hearing peers. Often, their caregivers were well-informed on the support available, on hearing impairment, on deafness and on deaf and hard of hearing rights, as a result of support and outreach from the special schools. These children appeared to be at lower risk of poor mental health. Children in mainstream environments were most likely to feel included if they used oral communication. These children talked of having hearing friends and generally positive experiences in the community. They tended to have milder hearing impairment and/or used a hearing aid or cochlear implant. They did not often report experiences of mental distress, likely related to fewer challenges in daily life and feeling part of the wider community. As is consistent with previous literature, our interviews indicated that the children most at risk of poor mental health were those experiencing language deprivation and social exclusion, especially in a mainstream schooling environment [31-33]. Often these deaf and hard of hearing children had moderate or severe hearing impairment, yet families did not support them to learn sign language, despite their limited capacity for oral communication. Many faced challenges with learning in school, including making friends. These children appeared to live with a sense of self that is in conflict; neither part of the hearing community and neither part of the Deaf community. This is consistent with previous literature on social identity and mental health among deaf and hard of hearing populations, whereby deaf and hard of hearing people associating with neither hearing or d/Deaf groups can lack a strong sense of self, negatively impacting their mental health [31, 32, 34, 35]. In contrast, mental health and wellbeing is often better in deaf and hard of hearing people using oral language and who associate with the hearing community, or those who have a strong sense of Deaf identity and use sign language. Deaf identities and wellbeing is a nuanced area and one with limited

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0001635 June 6, 2023

literature. We recommend additional research in the future into these topics, especially in LMICs and conflict affected populations.

Our findings suggest that, in general, Deaf culture and awareness is limited in the Gaza Strip. Many deaf and hard of hearing children do not have a positive self-image and many deaf and hard of hearing children in the Gaza Strip experience prejudice and discrimination at some point in the community. This was not universal, with some describing positive experiences. Largely, prejudice and stigma appeared to stem from a lack of awareness on hearing impairment and deafness, and understanding of Palestinian sign language as a language. This lack of awareness and knowledge was viewed as a challenge at both community and government level, and was shown in school settings. Stigma is a common barrier experienced by vouth with disabilities, but the current state of evidence on stigma-reduction activities for children with disabilities in LMICs is limited [36]. Most studies are of low methodological quality and as a result, current stigma reduction interventions are not evidence-based and often strategies do not target different levels of stigma [36]. That said, there are some promising stigmareduction interventions for children with disabilities identified in recent systematic reviews, including community education and social contact [36]. Recommendations from additional evidence reviews include the active involvement of people with disabilities (including youth) and caregivers in the development and delivery of stigma-reduction interventions and interventions that empower people with disabilities [37, 38]. Evidence on interventions is rarely focused solely on deaf and hard of hearing children, and there must be appropriate development efforts in partnership with deaf and hard of hearing children and families if looking to implement a deaf and hard of hearing stigma-reduction intervention in the Gaza Strip.

Key in our findings is the language deprivation experienced by some deaf and hard of hearing children in the Gaza Strip. Language deprivation in early life is a fundamental issue and has lifelong consequences [39]. Beyond isolation and loneliness, language deprivation will negatively impact on a child's key developmental skills [39]. Language is also the foundation of an individual's culture and social identity. Depriving deaf and hard of hearing children the right to natural sign language limits the progression of deaf and hard of hearing children's social identity, their self-esteem and their mental wellbeing. In the Gaza Strip, there are a number of factors contributing to this language deprivation. First, is the lack of opportunities for early exposure to sign language for deaf and hard of hearing children, training services for families and a lack of sign language interpreters. The infrastructure is not currently able to meet the demand. Second, this study indicated is the lack of family commitment to sign language. This was not the case for all families in this study, but it was evident in many. The reluctance to allow deaf and hard of hearing children to learn sign language appeared to be related to community stigma and limited information and awareness both on deafness and the fact the sign language is a language. In addition, limited training infrastructure resulted in limited opportunities to learn sign language, limited motivation to do so and limited knowledge on the value of sign language. Addressing inaccessible language and communication (and language deprivation in early life) should be a priority for supporting deaf and hard of hearing people in the Gaza Strip. Promoting Palestinian sign language use across the region will promote the inclusion of deaf and hard of hearing children and will help develop stronger deaf and hard of hearing communities. In their position paper and charter on the language rights of deaf children, the World Federation of the Deaf outlined actions that governments across the world must take, in order to promote the right of deaf children to fully develop their cultural and linguistic identity, as enshrined in the United Nations Convention on the Rights of Persons with Disabilities [40, 41]. They call for governments to "implement programmes to support the teaching of sign language to family members and carers of deaf children, in cooperation with Deaf communities and deaf sign language teachers." The position paper provides specific evidence and

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0001635 June 6, 2023

action-based recommendations for governments and community institutions to follow, including the need for early exposure to sign language and promoting family support. As part of the programme in which this research has been placed, CBM (an international disability NGO) and Atfaluna Society for Deaf Children have developed opportunities to learn Palestinian sign language for over 1,000 deaf and hard of hearing children and their families in the Gaza Strip.

As well as improved access to Palestinian sign language, many deaf and hard of hearing children would benefit from improved access to oral-communication services and devices. Our findings reveal a need to improve access to assistive technology, including hearing aid and cochlear implant assessment, provision and maintenance. Many respondents in this study reported experiencing difficulties with poorly functioning assistive products, which hindered oral communication for those that could use and may prefer it. In order to provide early support and intervention, the government in the Gaza Strip may wish to provide systematic newborn hearing screening programmes. Regardless of a family's decision on the use of hearing aids, cochlear implants and audiology therapies, we encourage health professionals to immediately refer deaf and hard of hearing children to Deaf associations, networks of parents of deaf and hard of hearing children when they first identify hearing loss.

In addition to the above recommendations, training for families may be important to improve deaf and hard of hearing awareness and mitigate the risk of language deprivation. Evidence shows that early intervention with caregivers promotes better awareness on disability, rights and supporting the child [42]. Interventions may include structured caregiver training programmes, such as the Juntos programme, which comprises interventions based on 10 module programme, delivered by caregivers with lived experience to a group of new caregivers and caregivers [43–45]. These programmes help caregivers learn the skills to support their child, whilst also offering peer support and sense of community, although these programmes have not been evaluated with families of deaf and hard of hearing children. Other similar interventions include the parent-mediated World Health Organization Caregiver Skills Training programme [46]. The use of social media may also be important to connect caregivers, share information on deafness and develop community.

With the importance of education and the role that school plays in the inclusion and wellbeing of deaf and hard of hearing children, it is important that schools are be able to support deaf and hard of hearing children. Throughout the interviews, it was clear that special schools in the Gaza Strip were better equipped to provide this support to deaf and hard of hearing children, with Palestinian sign language being routinely used to communicate, aiding both education and social inclusion. With the promotion of inclusive education across the world, mainstream schools in the Gaza Strip need the resources and training to better support deaf and hard of hearing children in their classrooms, including sign language training and interpretation services [47]. The relative merits of segregated and inclusive school systems are widely debated [48]. The findings of this study highlight the importance of strengthening inclusion in mainstream schools, so that deaf and hard of hearing children can attend if they wish. Although special schools in the Gaza Strip may continue to be needed, and may be preferred by some children and families, deaf and hard of hearing children should have the option to attend a mainstream school should they wish, with facilities available to support their full integration, in line with the principles of inclusive education. There is need for structures and resources by which mainstream school administrations and teachers can receive support and training to deliver inclusive education for deaf and hard of hearing children, whether this be from special school teachers or other certified personnel. Training for teachers should also include information on mental health and psychosocial support for deaf and hard of hearing

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0001635 June 6, 2023

children in their classrooms, including information on promoting wellbeing, recognising symptoms of distress, and actions to take when they identify an issue. As presented in this research, central to promoting the wellbeing and inclusion of deaf and hard of hearing children at school is partnership with caregivers, teachers and the child themselves. This promotes targeted solutions to support a child, both at school and home, and it gives the platform for caregivers to help teachers improve deaf awareness and inclusion in their classrooms. Until mainstream schools in the Gaza Strip are equipped to support all deaf and hard of hearing children, there is still need for a twin-track approach, with special schools still available and well-resourced.

In efforts to support deaf and hard of hearing children, there is need for government, funders and service providers to partner with deaf and hard of hearing communities and organisations of deaf and hard of hearing people in the Gaza Strip. Evidence has shown that organisations of persons with disabilities can produce positive outcomes for people with disabilities and these organisations are key in achieving equal rights for deaf and hard of hearing people and people with disabilities [49]. In the Gaza Strip, this may include training government, communities and families on deaf and hard of hearing rights and inclusion, building confidence for participation among deaf and hard of hearing children and families, and developing networks of deaf and hard of hearing communities. These organisations are often restricted by limited financial and human resources and they may therefore benefit from capacity building and increased funding to scale-up their work and effectively partner with government and other stakeholders looking to support deaf and hard of hearing children [49].

There is also a need to develop mental health support that is targeted to the needs of deaf and hard of hearing children. Although important to address the risk factors of poor mental health, as identified in this study, it is equally important to ensure appropriate support for those that do experience distress. In a systematic review of mental health support for deaf and hard of hearing children, appropriate interventions included peer-support, caregiver-child interaction therapy and resilience training [50]. Many interventions can be provided in schools and adaptations necessary include appropriate communication, including sign language. Mental health service provision for deaf and hard of hearing people would benefit from recruiting signing mental health professionals, deaf and hard of hearing professionals, and sign language interpreters specialized in mental health. Throughout, it is imperative to consult and coordinate with deaf and hard of hearing children and families, to ensure feasible, acceptable and sustainable interventions.

Children and adolescents in the Gaza Strip are at increased risk of mental health conditions, regardless of disability [25]. deaf and hard of hearing children face additional stressors, as demonstrated in this study. They can be said to experience double-jeopardy. They face stressors to their mental health and wellbeing as a result of living in a challenging region of the world and they face stressors as a result of living in an environment that is not fully inclusive and supportive of deaf and hard of hearing people. These stressors will intersect and raise unique challenges for deaf and hard of hearing children. This population is thus a priority for support.

Interesting to note in this discussion is the limited information reported by participants on the direct impact of conflict and blockade on the psychological wellbeing of deaf and hard of hearing children. Few participants reported on these issues, instead focusing on the topics discussed in the results of this study. This may be because questions asked in the topic guides did not include conflict-specific questions. Questions asked more broadly on factors that influence the mental health of children in the Gaza Strip. There was opportunity for participants to mention conflict and blockade under these questions. It is important to note also that data collection occurred before the recent escalation of rocket fire in 2021 and there is need to assess the

wellbeing of deaf and hard of hearing children (and indeed all children) since this time, so that appropriate support can be provided.

Limitations

This research encountered challenges resulting from the COVID-19 pandemic. These affected the process of data collection and the involvement of the lead researcher in the Gaza Strip itself. Several checks were in place to strengthen the integrity of data and interpretations, including remote training and supervision, and discussion of the findings with a steering committee, including people with lived experience.

It is also worth noting the greater number of female caregivers interviewed, compared to male. Fathers and other male caregivers were more difficult to recruit, as most were busy during working hours. Moreover, female caregivers in the Gaza Strip are most typically the primary caregiver for deaf and hard of hearing children and when we attempted to recruit to male caregivers, many told us that the mother or female caregiver would be able to give better information than they.

Conclusion

Deaf and hard of hearing children in the Gaza Strip experience a number of stressors that are detrimental to their mental health and wellbeing, including language deprivation, inaccessible communication and discrimination. Families, civil society actors and education providers must be empowered with the knowledge, skills and resources to provide an inclusive and supportive environment to deaf and hard of hearing children.

Supporting information

S1 Appendix. Participant characteristics. (DOCX)

S2 Appendix. In-depth interview and focus group discussion topic guides. (DOCX)

S3 Appendix. Questionnaire on inclusivity in global research. (DOCX)

Acknowledgments

Thank you to all of the participants who generously gave their time to take part in the study.

Author Contributions

Conceptualization: Nathaniel Scherer, Tracey Smythe, Julian Eaton, Naim Kabaja, Ritsuko Kakuma, Sarah Polack.

Data curation: Nathaniel Scherer.

Formal analysis: Nathaniel Scherer, Tracey Smythe, Ritsuko Kakuma, Sarah Polack.

Funding acquisition: Nathaniel Scherer, Julian Eaton, Naim Kabaja, Sarah Polack.

Investigation: Nathaniel Scherer, Tracey Smythe, Ramadan Hussein, Ritsuko Kakuma, Sarah Polack.

Methodology: Nathaniel Scherer, Tracey Smythe, Ramadan Hussein, Lorraine Wapling, Shaffa Hameed, Ritsuko Kakuma, Sarah Polack.

Project administration: Nathaniel Scherer, Ramadan Hussein.

Resources: Nathaniel Scherer, Ramadan Hussein.

Software: Nathaniel Scherer.

Supervision: Tracey Smythe, Julian Eaton, Naim Kabaja, Ritsuko Kakuma, Sarah Polack.

Visualization: Nathaniel Scherer.

Writing - original draft: Nathaniel Scherer.

Writing – review & editing: Tracey Smythe, Ramadan Hussein, Lorraine Wapling, Shaffa Hameed, Julian Eaton, Naim Kabaja, Ritsuko Kakuma, Sarah Polack.

References

- Haile LM, Kamenov K, Briant PS, Orji AU, Steinmetz JD, Abdoli A, et al. Hearing loss prevalence and years lived with disability, 1990–2019: findings from the Global Burden of Disease Study 2019. The Lancet. 2021; 397(10278):996–1009. https://doi.org/10.1016/S0140-6736(21)00516-X PMID: 33714390
- Bouldin E, Patel SR, Tey CS, White M, Alfonso KP, Govil N. Bullying and Children who are Deaf or Hard-of-hearing: A Scoping Review. The Laryngoscope. 2021; 131(8):1884–92. https://doi.org/10. 1002/lary.29388 PMID: 33438758
- Roland L, Fischer C, Tran K, Rachakonda T, Kallogjeri D, Lieu JE. Quality of Life in Children with Hearing Impairment: Systematic Review and Meta-analysis. Otolaryngology—head and neck surgery: official journal of American Academy of Otolaryngology-Head and Neck Surgery. 2016; 155(2):208–19. https://doi.org/10.1177/0194599816640485 PMID: 27118820
- Kral A O'Donoghue GM. Profound Deafness in Childhood. New England Journal of Medicine. 2010; 363(15):1438–50. https://doi.org/10.1056/NEJMra0911225 PMID: 20925546
- Stevenson J, Kreppner J, Pimperton H, Worsfold S, Kennedy C. Emotional and behavioural difficulties in children and adolescents with hearing impairment: a systematic review and meta-analysis. European Child & Adolescent Psychiatry. 2015; 24(5):477–96. https://doi.org/10.1007/s00787-015-0697-1 PMID: 25758233
- Lawrence BJ, Jayakody DMP, Bennett RJ, Eikelboom RH, Gasson N, Friedland PL. Hearing Loss and Depression in Older Adults: A Systematic Review and Meta-analysis. The Gerontologist. 2020; 60(3): e137–e54. https://doi.org/10.1093/geront/gnz009 PMID: 30835787
- Theunissen SC, Rieffe C, Kouwenberg M, Soede W, Briaire JJ, Frijns JH. Depression in hearingimpaired children. International journal of pediatric otorhinolaryngology. 2011; 75(10):1313–7. Epub 2011/08/25. https://doi.org/10.1016/j.ijporl.2011.07.023 PMID: 21862140
- Shoham N, Lewis G, Favarato G, Cooper C. Prevalence of anxiety disorders and symptoms in people with hearing impairment: a systematic review. Social psychiatry and psychiatric epidemiology. 2019; 54 (6):649–60. Epub 2018/12/14. https://doi.org/10.1007/s00127-018-1638-3 PMID: 30547211
- Fellinger J, Holzinger D, Pollard R. Mental health of deaf people. The Lancet. 2012; 379(9820):1037– 44. https://doi.org/10.1016/S0140-6736(11)61143-4 PMID: 22423884
- Fellinger J, Holzinger D, Sattel H, Laucht M, Goldberg D. Correlates of mental health disorders among children with hearing impairments. Developmental medicine and child neurology. 2009; 51(8):635–41. https://doi.org/10.1111/j.1469-8749.2008.03218.x PMID: 19627335
- van Gent T, Goedhart AW, Hindley PA, Treffers PD. Prevalence and correlates of psychopathology in a sample of deaf adolescents. Journal of child psychology and psychiatry, and allied disciplines. 2007; 48 (9):950–8. https://doi.org/10.1111/j.1469-7610.2007.01775.x PMID: 17714380
- 12. Ariapooran S. Relationship between mother- child interaction and symptoms of anxiety disorders in deaf and hard-of-hearing children. Empowering Exceptional Children. 2018; 9(2):51–63.
- Theunissen SC, Rieffe C, Kouwenberg M, De Raeve L, Soede W, Briaire JJ, et al. Anxiety in children with hearing aids or cochlear implants compared to normally hearing controls. Laryngoscope. 2012; 122(3):654–9. https://doi.org/10.1002/lary.22502 PMID: 22252674
- Charlson F, van Ommeren M, Flaxman A, Cornett J, Whiteford H, Saxena S. New WHO prevalence estimates of mental disorders in conflict settings: a systematic review and meta-analysis. The Lancet. 2019; 394(10194):240–8. https://doi.org/10.1016/S0140-6736(19)30934-1 PMID: 31200992

- 15. Pearce E. Disability Inclusion: Translating Policy into Practice in Humanitarian Action. New York, NY: Women's Refugee Commission, 2014.
- 16. Rohwerder B. Women and girls with disabilities in conflict and crises. Brighton: Institute of Development Studies, 2017.
- 17. UNICEF. The Gaza Strip—The humanitarian impact of 15 years of blockade—June 2022; 2022 [cited 2022 20 August. Available from: https://www.unicef.org/mena/documents/gaza-strip-humanitarian-impact-15-years-blockade-june-2022
- Abu Rayyan A, Kamal L, Casadei S, Brownstein Z, Zahdeh F, Shahin H, et al. Genomic analysis of inherited hearing loss in the Palestinian population. Proceedings of the National Academy of Sciences. 2020; 117(33):20070–6. https://doi.org/10.1073/pnas.2009628117 PMID: 32747562
- 19. Zaqqout R. Risk Factors of Hearing Impairment Among Infants and Toddlers in the Governorates: A Case Control Study. Jerusalem: Al-Quds University, 2018.
- Abu Hamad B, Jones N, Gercama I. Adolescent access to health services in fragile and conflict-affected contexts: The case of the Gaza Strip. Conflict and Health. 2021; 15(1):40. https://doi.org/10.1186/ s13031-021-00379-0 PMID: 34020683
- Elbedour S, Onwuegbuzie AJ, Ghannam J, Whitcome JA, Abu Hein F. Post-traumatic stress disorder, depression, and anxiety among Gaza Strip adolescents in the wake of the second Uprising (Intifada). Child abuse & neglect. 2007; 31(7):719–29. https://doi.org/10.1016/j.chiabu.2005.09.006 PMID: 17631959
- El-khodary B, Samara M. The effect of exposure to war-traumatic events, stressful life events, and other variables on mental health of Palestinian children and adolescents in the 2012 Gaza War. The Lancet. 2018; 391:S6.
- Khamis V. Post-traumatic stress and psychiatric disorders in Palestinian adolescents following intifadarelated injuries. Social Science & Medicine. 2008; 67(8):1199–207. https://doi.org/10.1016/j.socscimed. 2008.06.013 PMID: 18657343
- Sarhan AL, Aqtam I. Depressive Symptoms Among Palestinian Deaf Adults. European Psychiatry. 2015; 30:1.
- Abu Hamad B, Gercama I, Jones N, Al Bayoumi N. "I prefer to stay silent": Exploring opportunities for and challenges to adolescents' psychosocial and mental health in Gaza. London, UK: Overseas Development Institute, 2018.
- 26. Patton MQ. Qualitative research & evaluation methods. London: Sage; 2002.
- Braun V, Clarke V. Using thematic analysis in psychology. Qualitative Research in Psychology. 2006; 3 (2):77–101.
- Cypress BS. Rigor or Reliability and Validity in Qualitative Research: Perspectives, Strategies, Reconceptualization, and Recommendations. Dimensions of Critical Care Nursing. 2017; 36(4). https://doi. org/10.1097/DCC.00000000000253 PMID: 28570380
- Noble H, Smith J. Issues of validity and reliability in qualitative research. Evidence Based Nursing. 2015; 18(2):34. https://doi.org/10.1136/eb-2015-102054 PMID: 25653237
- UN General Assembly. Convention on the Rights of Persons with Disabilities: resolution adopted by the General Assembly, 24 January 2007, A/RES/61/106. New York, NY: UN General Assembly, 2007.
- Chapman M, Dammeyer J. The Significance of Deaf Identity for Psychological Well-Being. The Journal of Deaf Studies and Deaf Education. 2017; 22(2):187–94. https://doi.org/10.1093/deafed/enw073 PMID: 27881482
- Hintermair M. Self-esteem and Satisfaction With Life of Deaf and Hard-of-Hearing People—A Resource-Oriented Approach to Identity Work. The Journal of Deaf Studies and Deaf Education. 2008; 13(2):278–300. https://doi.org/10.1093/deafed/enm054 PMID: 17971343
- Mousley VL, Chaudoir SR. Deaf Stigma: Links Between Stigma and Well-Being Among Deaf Emerging Adults1. The Journal of Deaf Studies and Deaf Education. 2018; 23(4):341–50.
- Bat-Chava Y. Diversity of deaf identities. American Annals of the Deaf. 2000; 145(5):420–8. https://doi. org/10.1353/aad.2012.0176 PMID: 11191821
- 35. Maxwell-McCaw DL. Acculturation and psychological well-being in deaf and hard-of-hearing people: The George Washington University; 2001.
- Smythe T, Adelson JD, Polack S. Systematic review of interventions for reducing stigma experienced by children with disabilities and their families in low- and middle-income countries: state of the evidence. Tropical Medicine & International Health. 2020; 25(5):508–24. https://doi.org/10.1111/tmi.13388 PMID: 32145136
- 37. Mactaggart I, Felix L. Evidence Brief: How do we support women and girls with disabilities to overcome stigma against them? London: International Centre for Evidence in Disability, 2021.

- McKinney E. Evidence Brief: What works to eliminate stigma as a barrier to educational outcomes for children with disabilities? London: International Centre for Evidence in Disability, 2022.
- Hall ML, Hall WC, Caselli NK. Deaf children need language, not (just) speech. First Language. 2019; 39 (4):367–95.
- 40. World Federation of the Deaf, WFD Charter on Sign Language Rights for All. Helsinki: World Federation of the Deaf, 2019.
- 41. World Federation of the Deaf. WFD Position Paper on the Language Rights of Deaf Children. Helsinki: World Federation of the Deaf, 2016.
- Smythe T, Zuurmond M, Tann C, Gladstone M, Kuper H. Early intervention for children with developmental disabilities in low and middle-income countries—the case for action. International health. 2020;13.
- Duttine A, Smythe T, Calheiros de Sa MR, Ferrite S, Moreira ME, Kuper H. Juntos: A Support Program for Families Impacted by Congenital Zika Syndrome in Brazil. Global health, science and practice. 2020; 8(4):846–57.
- 44. Smythe T, Matos M, Reis J, Duttine A, Ferrite S, Kuper H. Mothers as facilitators for a parent group intervention for children with Congenital Zika Syndrome: Qualitative findings from a feasibility study in Brazil. PLoS ONE. 2020;15. https://doi.org/10.1371/journal.pone.0238850 PMID: 32913354
- 45. Smythe T, Reichenberger V, Pinzón E, Hurtado I, Rubiano L, Kuper H. The feasibility of establishing parent support groups for children with congenital Zika syndrome and their families: a mixed-methods study. Wellcome Open Research. 2021; 6:158.
- Salomone E, Pacione L, Shire S, Brown FL, Reichow B, Servili C. Development of the WHO Caregiver Skills Training Program for Developmental Disorders or Delays. Frontiers in psychiatry. 2019; 10:769. https://doi.org/10.3389/fpsyt.2019.00769 PMID: 31780960
- Mendoza M, Heymann J. Implementation of Inclusive Education: A Systematic Review of Studies of Inclusive Education Interventions in Low- and Lower-Middle-Income Countries. International Journal of Disability, Development and Education. 2022:1–18.
- Haug P. Understanding inclusive education: ideals and reality. Scandinavian Journal of Disability Research. 2017; 19(3):206–17. https://doi.org/10.1080/15017419.2016.1224778
- Young R, Reeve M, Grills N. The Functions of Disabled People's Organisations (DPOs) in Low and Middle-income Countries: A Literature Review. Disability, CBR & Inclusive Development. 2016; 27:45.
- Scherer N, Bright T, Musendo DJ, O'Fallon T, Kubwimana C, Eaton J, et al. Mental health support for children and adolescents with hearing loss: scoping review. BJPsych Open. 2021; 8(1):e9.

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0001635 June 6, 2023

PLOS GLOBAL PUBLIC HEALTH

BESEARCH ARTICLE

Development of mental health and psychosocial support (MHPSS) guidelines for deaf and hard of hearing children in the Gaza Strip

Nathaniel Scherer^{1*}, Ramadan Hussein², Julian Eaton^{3,4}, Naim Kabaja², Ritsuko Kakuma³, Tracey Smythe^{1,5}, Sarah Polack¹

1 International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Atfaluna Society for Deaf Children, Gaza City, Palestinian Territories, 3 Centre for Global Mental Health, London School of Hygiene & Tropical Medicine, London, United Kingdom, 4 CBM Global Disability Inclusion, London, United Kingdom, 5 Division of Physiotherapy, Department of Health and Rehabilitation Sciences, Stellenbosch University, Stellenbosch, South Africa

* nathaniel.scherer@lshtm.ac.uk

G OPEN ACCESS

Citation: Scherer N, Hussein R, Eaton J, Kabaja N, Kakuma R, Smythe T, et al. (2023) Development of mental health and psychosocial support (MHPSS) guidelines for deaf and hard of hearing children in the Gaza Strip. PLOS Glob Public Health 3(10): e0002427. https://doi.org/10.1371/journal. paph.0002427

Editor: Mohamed Salama, The American University in Cairo, EGYPT

Received: May 5, 2023

Accepted: September 2, 2023

Published: October 16, 2023

Copyright: © 2023 Scherer et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files. Data is not available via a public data repository. This was not approved by ethics committee and was therefore not included in the informed consent procedure.

Funding: This study was financially supported by the German Federal Ministry for Economic Cooperation and Development (BMZ: Bundesministerium für wirtschaftliche

Abstract

Deaf and hard of hearing children in the Gaza Strip may be at risk of mental health conditions and psychological distress, as a result of social exclusion and limited accessible communication. This article presents the process and research methods used to develop guidelines for schools in the Gaza Strip on mental health and psychosocial support for deaf and hard of hearing children. The process was guided by the GIN-McMaster guideline development checklist across four steps: (1) priority settings; (2) searching for evidence; (3) developing recommendations; (4) evaluation. Priority setting was spearheaded by local and international researchers, and a local steering committee comprised of deaf and hard of hearing representatives, school administration and staff, mental health specialists, family members and government officials. In searching for evidence, and in order to generate evidence-based recommendations for the guidelines, we utilised a scoping review of global mental health support for deaf and hard of hearing children and qualitative research with deaf and hard of hearing children and adults, families and teachers. Two pilot studies were conducted in mainstream and specialist educational settings as way of evaluation. The scoping review and qualitative research identified various content for the guidelines, including the importance of information on disability and deafness, promoting social inclusion and self-esteem, and accessible learning environments. The pilot studies demonstrated feasibility and acceptability among teachers and deaf and hard of hearing children, although teachers need sufficient support and resources to implement. Now finalised, the guidelines are being distributed to schools in the Gaza Strip to support the mental health and wellbeing of deaf and hard of hearing children.





Zusammenarbeit und Entwicklung) through CBM International under the project "Improving the psychosocial resilience of children with and without disabilities in the Gaza Strip" in the form of an award received by NS, SP, JE, RH and NK. NS and RH received salary through this funding for the research undertaken on this study. The specific roles of these authors are articulated in the 'author contributions' section. No additional external funding was received for this study. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Introduction

The Gaza Strip has a population of 2.1 million, with 1.4 million being Palestinian refugees. The region has been affected by conflict and a deteriorating socioeconomic situation [1, 2]. The employment rate is 49% and access to clean water and electricity has been deemed a crisis by the United Nations [2]. These hardships are associated with mental health problems amongst youth, with evidence demonstrating high rates of depression, anxiety and post-traumatic stress disorder among children and adolescents [3–6]. Research has suggested that education and family support can improve the psychosocial wellbeing of youth in the Gaza Strip [3]. However, few young people receive formal mental health support, and many in the Gaza Strip experience challenges accessing mental health services [3, 7].

Data from the Palestine Population, Housing and Establishments Census 2017 estimate a disability prevalence of 5.8% in Palestine and 6.8% in the Gaza Strip [8]. The prevalence of disability among children aged 0–17 in the Gaza strip is 0.9%, with 0.2% reporting difficulty hearing. This equates to approximately 20,000 children in the Gaza Strip who experience difficulties with hearing. Evidence suggests that deaf and hard of hearing children worldwide may face difficulties in education, socialization, and community engagement [9–11], and are more likely than their hearing peers to report lower quality of life in school and social settings [11]. These negative experiences can impact emotional wellbeing and mental health, and deaf and hard of hearing individuals, both children and adults, have an increased risk of developing mental health conditions [12–20].

Deaf and hard of hearing youth in the Gaza Strip may face double-jeopardy with regards to maintaining psychosocial wellbeing; they may face stressors associated with living in a challenging environment, as well as those associated with being deaf or hard of hearing in a predominantly hearing world. Qualitative research in the Gaza Strip, conducted with deaf and hard of hearing children, parents of deaf and hard of hearing children, teachers of deaf and hard of hearing children, deaf and hard of hearing children, teachers of deaf and hard of hearing children deaf and hard of hearing children deaf and hard of hearing children in the Gaza strip at risk of mental health problems [21]. Stressors included lack of accessible communication, exclusion from community and community events, negative attitudes towards hearing impairment, and limited family knowledge on hearing impairment and deafness.

As deaf and hard of hearing children in the Gaza Strip are at risk of developing mental health conditions, there is need for structures to promote their wellbeing and to provide support if they are distressed. Mental health and psychosocial support (MHPSS) is a term that usually refers to promotion, protection and treatment for mental health and psychosocial wellbeing in emergency settings [22]. MHPSS takes a variety of forms, depending on context, resources and needs. Activities cover the full range of mental health needs, from (subclinical) distress, to treatment and prevention of mental health conditions, such as depression and anxiety. Activities include promotion of protective factors and behaviours that contribute to wellbeing, including social inclusion and access to community services [22]. Considering the important role that school plays in the lives of children and adolescents, educational settings are increasingly considered important environments in which to promote mental health and wellbeing. recognised schools as important environments in which to promote the mental health and psychosocial wellbeing of children [23]. Research with adolescents around the world identified school as a source of self-esteem, a setting for emotional support and an escape from adverse home environments, whilst also presenting academic pressures and a setting where children can experience abuse [24]. In a recent briefing note, the World Health Organization, UNESCO and UNICEF called on national governments to establish systems in schools that foster mental health and wellbeing, and effectively manage the mental health

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0002427 October 16, 2023

needs of students [25]. With deaf and hard of hearing children demonstrating unique psychosocial support needs, it is necessary to provide teachers and school staff in the Gaza Strip with tailored guidance on how to promote the mental health and psychosocial wellbeing of these children.

This article presents the methods and process used to develop guidance on MHPSS for schools to support deaf and hard of hearing children in the Gaza Strip, including information on evidence-generating activities to inform the content (such as a scoping review and qualitative research) and pilot testing of the guidelines to ensure contextual appropriateness. Throughout the study, we adopted participatory approaches to engage deaf and hard of hearing people and their families, to ensure that their voices and perspectives were heard, valued and included, in line with the principle espoused by the disability movement; "Nothing about us, without us" [26].

Method

The MHPSS guidelines for deaf and hard of hearing children in the Gaza Strip were developed in line with the principles of the GIN-McMaster Guideline Development Checklist [27], developed by the Guidelines International Network (GIN) and McMaster University. The checklist comprises 18 key development steps and 146 individual items, including budgeting, stakeholder involvement and evaluation. We adapted the GIN-McMaster guideline development process to suit project objectives and resources. For ease of interpretation in this paper, we have summarised the development process under four key steps (numbers from the GIN-Mc-Master topic checklist are provided in brackets):

- 1. Priority setting (1–9)
- 2. Searching for evidence (10-12)
- 3. Developing recommendations (13-15)
- 4. Evaluation and use (16–18)

The project started during the early stages of the COVID-19 pandemic and this influenced the methods detailed and steps taken to develop the guidelines. Fig 1 provides an overview of the guideline development process.

Priority setting (1-9)

Priority setting activities were conducted across the following steps:

1. Global stakeholder workshop. In 2017, the International Centre for Evidence in Disability (ICED) at the London School of Hygiene & Tropical Medicine (LSHTM), CBM and DeafKidz International conducted a workshop to identify research and programmatic priorities to support deaf and hard of hearing children in low- and middle-income countries (LMICs). The group identified mental health and MHPSS as priorities for deaf and hard of hearing children in LMICs, the group proposed developing guidelines that could be used by families, teachers, healthcare staff and other community stakeholders.

2. Development programme in the Gaza Strip. In 2020, ICED established a partnership with CBM and Atfaluna Society for Deaf Children to develop guidelines on MHPSS for deaf and hard of hearing children in the Gaza Strip. This partnership brought together a research institute in global disability and mental health (ICED), an international disability NGO (CBM) and a local implementing partner focused on supporting deaf and hard of hearing children (Atfaluna Society for Deaf Children).

Priority setting

- December 2017: Global stakeholder workshop to identify priorities for deaf and hard of hearing children in low- and middle-income countries
- January 2020: Priority setting in the Gaza Strip with local steering committee

Searching for evidence

- April 2020: Scoping review of mental health support for deaf and hard of hearing children worldwide
- August 2020: Qualitative research on the wellbeing of deaf and hard of hearing children in the Gaza Strip

Developing recommendations

- November 2020: Workshop with local steering committee to translate evidence into guideline structure and content
- February 2021: Guidelines reviewed by local steering committee

Evaluation and use

- April 2021: Pilot with teachers at Atfaluna Society for Deaf Children education programme in Gaza City
- October 2021: Pilot with teachers in mainstream and special schools across the Gaza Strip
- April to November 2021: Guidelines reviewed by global stakeholders with expertise in support for deaf and hard of hearing children
- November 2021: Guidelines reviewed by independent group of Deaf leaders in the Gaza Strip
- December 2021: Final review by local steering committee and global stakeholders
- January 2022: Content of the guidelines finalised, ready for roll-out to schools

Fig 1. Guideline development process and timeline.

https://doi.org/10.1371/journal.pgph.0002427.g001

Guideline development was incorporated into a wider project being delivered by CBM and Atfaluna Society for Deaf Children, which aimed to provide training and interventions to promote the psychological wellbeing of children with disabilities, and in particular, deaf and hard of hearing children. The guideline development process was managed by ICED at LSHTM, in partnership with Atfaluna Society for Deaf Children. The team at ICED and LSHTM was comprised of researchers in disability and Global Mental Health, with experience in development guidance and manuals on a variety of health-related topics. The work at Atfaluna Society for

Deaf Children was led by a Project Manager and Research Assistant. The Research Assistant was an audiologist at Atfaluna Society for Deaf Children, seconded to the research project for the duration. This collaborative research team comprise the authors on this paper.

3. Local steering committee. To guide the development process and to set initial priority guideline topics, the team established a steering committee in the Gaza Strip. This group comprised: five deaf and hard of hearing people and representatives from organisations of deaf and hard of hearing people; two caregivers of deaf and hard of hearing children; four teachers of deaf and hard of hearing children; three mental health and psychosocial support specialists; and five representatives from government level institutions, including the Palestinian Ministry of Education and Higher Education. Group members were identified through the networks of Atfaluna Society for Deaf Children and the Palestinian Ministry of Education and Higher Education. Each member was provided a term of reference outlining their role as a steering committee member. The steering committee gathered with the research team three times throughout the development process, as described here and in subsequent steps. The meetings were held in-person, although the lead author from LSTHM attended virtually, with international travel restricted as a result of COVID-19. Typically, steering committee meetings were structured as half-day workshops, with opportunity for breakout sessions and group discussion. In the first meeting, the steering committee gave input to the research methodology, guideline development plan and priorities for the guidelines. The steering committee gathered two further times throughout the project, as described in the following sections.

4. Consultation with global experts. In the early stages of the guideline development, the lead author consulted with international experts in mental health and support for deaf and hard of hearing children, including expertise in intervention development and delivery within low- and middle-income and humanitarian settings. Six experts from the UK, Syria, and Spain spoke to the lead author for approximately one hour via online communication technology, discussing priority topics, useful resources and other pertinent information. This information was used to guide the planning of the research components described.

Search for evidence (10–12)

To inform the guideline content, we included two evidence-generating components: a systematic scoping review on mental health support for deaf and hard of hearing children globally; and qualitative research to explore experiences of deaf and hard of hearing children in the Gaza Strip and in schools, and their recommendations on MHPSS guideline content. Systematic review to search for evidence is explicitly recommended by the GIN-McMaster Checklist, although recommendations regarding qualitative research are absent. We conducted the additional qualitative research to provide an understanding of local contexts, cultural norms and attitudes to disability and mental health, and the experiences of deaf and hard of hearing children that may impact their wellbeing. There is a dearth of evidence on these areas, both in the Gaza Strip and globally, and this qualitative research was considered key to the development of contextually appropriate guidelines, as well as contributing to the wider evidence base.

1. Scoping review. A scoping review was undertaken to map the global evidence on initiatives and interventions designed to protect and promote mental health and wellbeing support for deaf and hard of hearing children. Further details of the scoping review methods are available in the peer-reviewed article published in 2021 [28]. The review was guided by the framework developed by Arksey and O'Malley (2005) [29].

2. Qualitative research in the Gaza Strip. To further understand the context and the experiences of deaf and hard of hearing children in the Gaza Strip, we conducted

phenomenological qualitative research. This included in-depth interviews with 35 participants; 17 deaf and hard of hearing children aged 6–12 in mainstream and special schools, 10 parents or caregivers of deaf and hard of hearing children, and eight teachers of deaf and hard of hearing children from mainstream and special schools. Participants were purposively sampled from across the Gaza Strip (North, Gaza City, Central, South), and as following Patton's maximum variation sampling [30], to reach a variety of ages, genders, severities of hearing impairment, location and types of school.

We also conducted three focus group discussions. The first comprised of representatives from Organisations of Persons with Disabilities (OPDs) and Deaf focused NGOs. Participants of this group included three deaf adults, one individual with a physical impairment and one with a visual impairment, who also works as a mental health counsellor. The second group included five school counsellors and two mental health professionals working with children in the community. The third group included seven special school teachers and six mainstream school teachers from across the Gaza Strip. These teachers were not the same as those included in the in-depth interviews.

Participant characteristics are presented in S2 Appendix. Data were collected in July 2020. The Research Assistant at Atfaluna Society for Deaf Children conducted the interviews and focus group discussions after receiving online training and remote supervision from ICED due to international COVID-19 travel restrictions. The interviews and focus groups themselves were conducted in-person, with safety measures adopted, as advised by the local government.

The interview guides were developed by the research team, piloted with six participants (two children, two caregivers, two teachers) and adapted as needed thereafter. Ouestions explored the experiences of deaf and hard of hearing children, families and teachers in education, health and community settings. For example, caregivers were asked, "How do people in the community view deaf and hard of hearing people?", and children were asked, "How is the experience of learning in school?". Specific questions were also asked on the planned MHPSS guidelines to elicit recommendations for the content and roll-out. For example, caregivers were explicitly asked, "What information should be included in the guidelines?", and teachers were asked, "What support would you need to use the guidelines?". Interviews were conducted orally in Arabic, with Palestinian sign language interpretation available. Children were interviewed individually, unless the child or caregiver desired otherwise. There was occasionally need for a family member to sit in as interpreter, when a child communicated through a signed language developed with family at home. Emotion cards were available to help children express their feelings. Each interview lasted 30-60 minutes. Five pilot interviews were conducted, in which to practice interviewing techniques and refine the topic guides. The data from these interviews has been included in the analysis. Audio-recordings of the interviews and focus group discussions were transcribed into English and thematically analysed [31]. Analysis was guided by the six steps outlined by Braun and Clarke (2006). First, the lead author familiarised themselves with the data, developing a codebook which was reviewed by coauthors. This codebook was piloted and iteratively adapted throughout the coding process. Coding was completed by the lead author in NVivo 12, with emerging themes identified, refined, and relationships compared between codes. The key themes were subsequently reviewed by co-authors and mapped against the data and emerging narrative. Finally, participant codes and narrative case studies were extracted. The themes were discussed with the steering committee to aid interpretation. Detailed analysis on factors contributing to the wellbeing of deaf and hard of hearing children in the Gaza Strip have been published elsewhere [21]. In this paper, we present findings relevant specifically to guideline development.

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0002427 October 16, 2023

Developing recommendations (13-15)

Next, we held a series of participatory workshops with the steering committee, in order to develop consensus on the structure and content of the guidelines. During the workshops, key themes and recommendations emerging from the evidence-generating activities were presented. These were used by the group to collectively agree on the planned structure and content of the guidelines.

Two workshops were held with steering committee members at this stage. The first was held with parents, deaf adults, teachers, head teachers and mental health specialists. The second was held with the representatives from the Palestinian Ministry of Education and Higher Education and UNRWA. This separation aimed to reduce power dynamics and to promote a bottom-up approach to guideline development. The first session developed the core outline of the guidelines, with the second session focused more on review, comment and consideration of the recommendations in the context of education systems and infrastructure in the Gaza Strip. From these recommendations, the lead author developed the initial draft of the guidelines for pilot.

The steering committee gathered for a third time once the first draft of the guidelines was complete, in order to review the draft and discuss amendments to the content. The first draft was also reviewed in an in-person workshop with a separate group of Deaf leaders in the Gaza Strip, who provided their opinions on the content and needed changes. Further, global stake-holders with expertise in supporting deaf and hard of hearing children reviewed the first draft via email, including Al Damair Society, Dair Al Balah Society for Rehabilitation, and Al Amal Society for Rehabilitation (all located in the Palestinian Territories), and the World Federation of the Deaf.

Evaluation and use (16-18)

Once redrafted, the guidelines were pilot tested twice in the Gaza Strip. Initially, the plan was to test the guidelines with teachers in mainstream and special schools, but with widespread school closures as a result of COVID-19, the team had to find other ways to test the guidelines. Atfaluna Society for Deaf Children had dispensation to continue running a face-to-face education programme throughout the pandemic and we decided this would serve as the first pilot site. Once mainstream and special schools re-opened, a second pilot was held.

1. Pilot one: Atfaluna Society for Deaf Children education programme (April 2021). The education programme at Atfaluna Society for Deaf Children provides educational opportunities, as well as broader support and counselling for children with disabilities and education challenges. All 10 teachers involved in the programme agreed to take part in the first pilot of the guidelines. Each received a half-day orientation on the guidelines and then used the guidelines within their class for three-weeks.

A pre- and post-questionnaire, based on Bowen's feasibility framework, was administered to teachers before and after guideline use [32]. The questionnaire explored four of the eight components of the framework: Acceptability (To what extent the guideline is judged as suit-able?), Demand (To what extent the guideline is likely to be used?), Implementation (To what extent the guideline can be successfully delivered by teachers?) and Practicality (To what extent can the guideline be implemented with existing means and resources?). Additionally, we asked questions on knowledge about disability, deafness and mental health support among teachers before and after using the guidelines. We further included open-ended questions, in which participants could make suggestions on amendments to the content and the format.

After completing the pilot and questionnaire, the 10 teachers participated in a half-day workshop with the research team, to discuss their experience of using the guidelines and suggestions for improvement.

2. Pilot two: Mainstream and special schools in the Gaza Strip (August 2021). Once schools re-opened, we proceeded with the second pilot, within mainstream and special schools. In total, 30 teachers were included in the second pilot across nine schools; 19 from mainstream schools, 11 from special schools. Teachers were purposively selected to represent a diverse group across gender, age group taught, region and school type (mainstream/ special).

As with the first pilot, teachers received orientation, used the guidelines with their class (for four-weeks) and completed pre-and post- questionnaires They also participated in a half-day workshop to discuss their experience and suggestions for improvement. Additionally, we spoke to the 10 head teachers of the included schools, with regards to their views of the teacher's experience and feedback.

It is important to note that pilot two was slightly disrupted by COVID-19, with some schools closing for periods of the pilot and school days shorter than usual. No school was closed for the duration of the pilot, however.

Based on the feedback in pilots one and two, the guidelines and orientation were updated, as described in the results. Members of the steering committee and the World Federation of the Deaf were invited to provide final review and comment to this updated draft via email.

Ethical approval

Ethical approval was provided for the qualitative research and pilot studies by the ethics board at the London School of Hygiene & Tropical Medicine (Ref: 19144 and 25113, respectively) and the Palestinian Health Research Council (Ref: PHRC/HC/697/20 and PHRC/HC/855/21). Informed written consent was obtained from all participants. Caregivers and guardians provided informed consent for children. Assent was sought from children using a simplified information sheet. For clarification, caregiver consent and child assent was obtained because of the children's age, not because the children were deaf or hard of hearing.

Results

Priority setting (1-9)

From the outset, the project partners identified schools as a setting to provide the MHPSS guidelines, given the benefits of early intervention, the important and regular contact between teachers and deaf and hard of hearing children, and the importance of the setting to aspects of importance to psychological wellbeing, such as social inclusion, peer-support and education [33–35].

In the early stages of the project, both the local steering committee and international experts recommended that the guidelines should focus on improving teachers' understanding of hearing impairment and deafness in the Gaza Strip. They believed that before supporting deaf and hard of hearing children's mental health and wellbeing, teachers must first have insight into what deaf and hard of hearing children experience. This included learning about child development, communication and learning preferences of deaf and hard of hearing children. When identifying priority topics for MHPSS specifically, we were told that teachers in the Gaza Strip needed information on identifying mental distress and information on support to provide. This should also include promotion and prevention techniques to promote wellbeing in the classroom. The committee said that it was vital to include contextually appropriate MHPSS activities that teachers would feasibly be able to deliver with the resources available. They recommended searching for evidence on low-intensity interventions, such as safe spaces for deaf and hard of hearing children, emotional literacy training and peer-support.

Search for evidence (10-12)

1. Scoping review. Results of the scoping review of MHPSS interventions for deaf and hard of hearing children are published elsewhere [28]. In summary, 27 articles were included, the majority (81%) of which were sourced from high-income settings. Interventions were typically therapy-based (30%) or skills training (30%) and included ice-skating, parent-child interaction therapy and resilience training.

From the review, key lessons taken to inform content of the MHPSS guidelines included the importance of: peer-support building resilience, emotional literacy and behavioural management, physical activity, knowledge and awareness of disability and hearing impairment, appropriate communication, accessible physical environments and materials, collaboration between teachers, parents and children, and conducting pilot studies (Table 1).

2. Qualitative research. The qualitative research uncovered key themes that contribute to psychological distress and psychological wellbeing of deaf and hard of hearing children in the Gaza Strip, including language deprivation, community attitudes and family support. Findings relevant to these themes have been published elsewhere [21]. Recommendations for the guide-lines extracted from these data included the need for appropriate communication, sign language training, promoting social inclusion of deaf and hard of hearing children in school, greater provision of assistive technology and training, community awareness and reduction of stigma, and parent support and sensitisation.

Key themes derived from data specific to the MHPSS guidelines included the desire from teachers for resources and guidance, the need for information on disability and deafness, actions to promote social inclusion, developing children's self-esteem and emotional literacy, how to build inclusive learning environments, and how teachers and parents can work together.

With regard to perceptions of the proposed MHPSS guidelines and their recommendations for content, the majority of participants reacted positively. Many teachers, especially those in mainstream schools, stated that they needed and wanted training and guidance on supporting deaf and hard of hearing children.

"[When asked if the guide will be useful] Of course it will be. It is going to include new information and knowledge that a teacher can realize and apply with the children. As someone still

Table 1. Summar	y of content derived	from evidence-generation	ating activities.
-----------------	----------------------	--------------------------	-------------------

Evidence generating activity	Content included in the guidelines		
Scoping review	Importance of peer-support Promoting emotional literacy Behavioural management Supporting physical activity Improving knowledge and awareness of deafness and hearing impairment Communication is key Developing accessible environments Importance of collaboration		
Qualitative research	 Information on deafness and hearing impairment, improve knowledge of teachers, parents and other children Promote social inclusion, include a buddy system with a hearing partner Build self-esteem, give deaf and hard of hearing children responsibility in the classroom Improve emotional literacy in deaf and hard of hearing children Collaborate with school counsellors to help children needing support Create an accessible learning environment, consider appropriate communication and classroom setup Develop partnership between parents and teachers to provide continuity of support 		

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0002427 October 16, 2023

learning, this will enhance my knowledge, because you still are unaware of many things. The guide will help in filling the gaps, especially in solving problems." (Teacher at a special school)

Key content areas that participants felt should be in the guidelines included information on disability, hearing impairment, deafness, Deaf culture and mental health (Table 1). We were told that many teachers, school staff and hearing children didn't know much about these topics and this sensitisation was vital if deaf and hard of hearing children were to be included and respected in the classroom.

Exclusion and isolation were said to be common for deaf and hard of hearing children, especially in mainstream schools, and this exclusion was an important determinant of mental distress. Suggestions to improve inclusion included encouraging group work and extra-curricular activities, to build relationships and friendships. Some participants also recommended a buddy system, in which a hearing child is partnered with a deaf or hard of hearing child in a mainstream school to help them learn and feel included in the classroom. Many deaf and hard of hearing children expressed an interest in participating in social groups with other deaf and hard of hearing children, citing a common experience and understanding.

In addition to understanding hearing impairment and deafness, participants also requested information on assistive technology, particularly hearing aids and cochlear implants. This included practical information on basic maintenance, such as changing hearing aid batteries. Many teachers expressed that they and hearing children in mainstream schools did not understand what the assistive technology was nor its purpose. deaf and hard of hearing children told us of examples in which other children removed their hearing aid, thinking they were listening to music.

"Once the teacher was afraid of her hearing aid, because it was making a buzzing sound, which is something annoying. When the teacher saw the hearing aid she was frightened. [...] They should be educated about hearing aids and such problems." (Father of a deaf child aged 10–12)

To protect against mental distress, participants said that teachers must help develop a child's sense of identity and self-esteem. Teachers were encouraged to give deaf and hard of hearing children responsibility in the classroom, such as being class secretary for a period. To further support mental health and wellbeing, participants wanted information on emotional literacy and how to help children understand their own feelings and emotions. It was also vital that teachers be aware of how to recognise mental distress and the process to follow if they identify mental distress, including referral to school counsellors. School counsellors urged teachers to work with them when they have a concern, although teachers in the study told us that this collaboration was often not strong.

"Interviewer: What do you think would contribute to improving the psychosocial status of deaf and hard of hearing children?

Respondent: By improving their skills. For example, look at what the child loves and work on improving this and improving the child's ability in this field. When a child with a disability is left behind because of their disability, this impacts their mental health and causes them psy-chosocial problems. We need to improve their skills, look into their talents and work on developing them. This would improve their self-trust and help them overcome mental health problems." (Counsellor at a mainstream school)

As well as an explicit focus on mental health and wellbeing, participants also suggested that the guidelines include information on how to provide an accessible and inclusive learning environment, especially in mainstream education. This was echoed by the steering committee in early-stage meetings. Recommended areas to focus on included communication, accessible teaching materials and the physical environment, including consideration of seating arrangements and minimising distracting noises.

Finally, throughout the qualitative research, participants emphasised the need for teachers and parents to work together. Many parents wanted to be involved and to help teachers support their child's learning and wellbeing. They also wanted to be given the tools to continue support at home. Regular communication was said to be needed and some participants suggested a joint notebook, in which teachers and parents could keep a track of a child's experience at school and home, and actions being taken.

"We always prefer parents to be aware of persons with disabilities needs, sign language and support services available to them. If I'm the only one who communicates with the child and there's no communication at home, our efforts will be in vain. This is why we keep encouraging parents to communicate in sign language with their deaf and hard of hearing children. It's really dangerous to depend on the teacher only." (Teacher at a special school)

One father told us of an instance in which his hard of hearing son, attending a mainstream school, was exhibiting challenging behaviours and difficulties with concentration and learning. The father worked with the teacher and discussed how much of this arose from inequalities and exclusion in the classroom. The teacher was responsive and made efforts in the classroom to promote the inclusion of the hard of hearing child. The child's behaviours improved thereafter.

In addition to guideline content, participants also discussed the eventual roll-out of the guidelines across mainstream and special schools in the Gaza Strip. Teachers expressed a need for institutional awareness and buy-in to support them in implementing the guidance in their classroom. It was said to be particularly important that school administration, head teachers and staff understand the value of the guidelines and give teachers the support they need. Teachers also said they would need training on using the guidelines, including ongoing mentorship from school counsellors.

Developing recommendations (13-15)

Table 2 presents the guideline structure and content summary developed with review by steering committee, Deaf leaders and experts in mental health and deaf and hard of hearing children. Throughout the guidelines, teachers are provided with practical activities to support deaf and hard of hearing children. Examples include role-play to help demonstrate positive behaviours and setting up a 'buddy' system, pairing a deaf or hard of hearing child with a hearing peer, who can help them navigate school life.

Evaluation and use (16-18)

The results of pilots one and two have been summarised against the four areas of feasibility from Bowen's framework [32]. Data from pilot one and two have been combined. Responses from open-ended questions are provided.

Acceptability: To what extent were the guidelines judged as suitable, satisfying or attractive to teachers?. Across the two pilots (n = 40), 90% of teachers reported that the guidelines met/exceeded their expectations, and 95% were satisfied/very satisfied with the guidelines. Teachers reported the perceived value of the guidelines, that were seen to be rich in

PLOS GLOBAL PUBLIC HEALTH

Section	Summary of content
Understanding mental health and psychosocial support	What is mental health and wellbeing?Risks to mental health and wellbeingMental health in the Gaza Strip
Understanding disability	What is disability? Barriers experienced by people with disabilities Impact of disability Disability rights
Understanding hearing impairment and deafness	 What is hearing impairment? Severity of hearing impairment What is deafness? Deaf culture Assistive technology (e.g. hearing aids) Myths and facts
Impact of hearing impairment and deafness	Communication and language Influence on social skills Impact on education Impact on mental health
Supporting deaf and hard of hearing children	Promoting understanding and information on hearing impairment and deafness Addressing stigma Promoting social inclusion Accessible communication Accessible classroom environment Accessible learning materials
Supporting the mental health and wellbeing of deaf and hard of hearing children	Building positive deaf and hard of hearing identity Developing inclusion and friendship Promoting self-esteem Emotional literacy Resilience training Promoting positive behaviour Physical health
Identifying a mental health issue	Signs of mental distress Steps to take on identification, including referral to school counsellor
Collaboration	How to coordinate with other teachers How to collaborate with school counsellors Working in partnership with parents and caregivers Continuing support at home
Appendices	List of hearing services in the Gaza Strip List of MHPSS services in the Gaza Strip Simple Palestinian sign language Hearing aid care Creating an inclusive environment: self-assessment for teachers Questions to ask deaf and hard of hearing children about their school experience Fmotions cards

https://doi.org/10.1371/journal.pgph.0002427.t002

detail and easy to use. The simple language was easy to understand and commonly cited a positive of the guidelines

"I am satisfied with the content. It contains information that has added to my knowledge and made me aware of some of the problems facing deaf and hard of hearing children in my classroom." (Teacher at mainstream school)

With regards to perceived effect of the guidelines, most (81%) of the teachers reported the guidelines to moderately/significantly improve their knowledge of MHPSS, with 76%

reporting moderately/significantly improved knowledge on MHPSS for deaf and hard of hearing children specifically. There was reported increased understanding of the challenges facing deaf and hard of hearing children and the need for MHPSS, increased understanding of MHPSS and how to support wellbeing, and increased understanding of inclusion within the classroom. Among those that reported a slight improvement in knowledge only, many stated that they needed more time to become familiar with the topics of the guidelines.

Nearly all (90%) teachers told us that they had moderately/significantly changed their actions and behaviours in their classroom. Teachers told us that they have started to speak to deaf and hard of hearing children and families about their needs, including communication. Teachers promoted inclusion in the classroom and held sessions in mainstream schools to explain hearing impairment, deafness and hearing aids to other students. Many encouraged deaf and hard of hearing children to talk to their class on their hearing impairment and on their use of hearing aids. Teachers also reported actively encouraging deaf and hard of hearing children to participate in classroom activities.

Further, 79% of teachers reported that deaf and hard of hearing children in their classrooms moderately/significantly accepted use of the guidelines and many teachers reported positive outcomes among deaf and hard of hearing children. Deaf and hard of hearing children and their families had reacted well to teachers showing an interest in their needs, and the children noticed a difference in the teacher's and classmates' understanding and approach with them. According to teachers, deaf and hard of hearing children were included more by their peers and they were more active in class. Many reported that deaf and hard of hearing children's confidence had grown and they were more open with the class and teachers. Teachers reported improved relationships between them and the children.

"We are in a better relationship now. They love the activities we are doing together and they ask for more." (Teacher at education programme)

Teachers at special schools recognised that much of the information in the guidelines was not relevant to them, as they already have a strong understanding of disability and hearing impairment, but the elements of mental health were important and they perceived a benefit to distribution across special schools.

Demand: To what extent were the guidelines likely to be used by teachers?. All teachers said that they would be likely/very likely to use the guidelines in the future. Overall, 83% of teachers believed that the guidelines would be moderately/significantly useful for teachers to support the mental health and wellbeing of deaf and hard of hearing children. The remaining reported it slightly useful, reporting reservations over capacity.

When asked how often they apply the recommendations of the guidelines, 36% of teachers reported weekly, 48% two to three times per week, and 17% daily. Teachers told us that they often referred to the guidelines to understand a child's behaviour and to identify concerns.

"I read and referred to the guidance frequently. I wanted to be more aware of all aspects of MHPSS and to notice any concerns with the children I am teaching." (Teacher at mainstream school)

The majority (93%) of the teachers said that they were moderately/very confident to use the guidelines in the future, with all teachers reporting that the induction on the guidelines was needed and useful.

Implementation and practicality: To what extent could the guidelines be successfully delivered to deaf and hard of hearing children in mainstream and special schools using

existing resources?. Although teachers were interested to use the guidelines, results from the questionnaire and workshop highlighted concerns among teachers about their capacity to use them; 40% said the guidelines would be difficult/very difficult to use alongside their current teaching schedule and activities. This increased to 53% for mainstream and special school teachers in pilot two only. For some, this reflected the situation related to COVID-19 and shorter school days. Some believed they would have capacity once normal schedules resumed. Others said it was difficult to find the time to read and implement the guidelines, although there was acknowledgement that the pilot had run for a relatively short-time only and with more time they would be able to better take in the information and make changes.

The majority of teachers said that they needed practical support from the school administration to implement the guidelines, including provision of accessible learning materials and resources, time in their schedule, and ongoing capacity-development.

"Teachers need training and support from the school administration to be able to apply the guideline... If this would be possible, the guideline will be very helpful to ensure needed positive change." (Teacher at education programme)

Some of the teachers reported that they had received ongoing advice and support from school counsellors as they learned to use the guidelines during the pilot, citing this a significant facilitator to guideline implementation. Nearly all (95%) said that the orientation was needed to prepare teachers to use the guidelines, an important consideration when aiming to scale-up guideline use across all schools.

Concerns of teacher capacity were further reported by head teachers. In 2021, there had been a shift in the school system. The new structure consists of four quarters (instead of two semesters); the children go to school for two months, then have exams, followed by 10 days off. This is repeated four times throughout the year. We were told this results in shorter and busier days and teachers have more work to do. The teachers felt this change fostered a greater focus on academic achievement and less time for wider activities, such as mental health and wellbeing support. This was reported important to consider when rolling-out the guidelines across the Gaza Strip.

Adaptations to the guidelines. After completing pilots one and two, small adaptations were made to the guidelines, based on feedback from the teachers. Amendments included additional information on the impact of hearing impairment of child development, more examples on how to promote social inclusion of deaf and hard of hearing children in class-rooms, and clarity on the referral process when a teacher identifies a child in distress. Additionally, teachers asked for more information on the impact on parents of deaf and hard of hearing children and more information on how to involve parents in their child's education. After making these amendments, the content was reviewed online by steering committee members and the World Federation of the Deaf. The guidelines were subsequently finalised, ready for use in schools in the Gaza Strip.

Guideline use. With the guidelines complete, Atfaluna Society for Deaf Children, in collaboration with the Palestinian Ministry of Education and Higher Education, are systematically rolling-out the guidelines across mainstream and special schools in the Gaza Strip. To aid this roll-out, they have developed a dissemination plan, including presentation events at schools. They have also developed a training programme for parents and school staff, including teachers, administration and counsellors, on mental health and psychosocial support, including use of these guidelines.

The guidelines are now available online in English and Arabic, and we have encouraged others to adapt and use the guidelines as suitable.

Discussion

This article has described the participatory process of developing guidelines on MHPSS for deaf and hard of hearing children in the Gaza Strip, including detail on activities to generate evidence-based content and evaluation to assess feasibility and acceptability. Developing the guidelines throughout the COVID-19 pandemic resulted in challenges, but using the GIN-Mc-Master checklist provided best-practice guidance to support the development process.

Practicality of the guidelines

From the outset, the research team and local steering committee identified schools as the priority setting for these MHPSS guidelines, congruent with long-standing research and practice on the suitability of schools as a natural setting to deliver mental health support [36, 37]. Children spend a substantial amount of their time in these settings and schools play a significant role in a child's social and emotional development, providing opportunity for peer-interaction, behavioural management, and academic progression [37]. Schools often have existing resources and support systems on which to scaffold mental health support, whether that be prevention, promotion or low-intensity intervention, and they are well-placed to connect children and families to further services. In the Gaza Strip, schools are meant to have a counsellor on staff, who can offer direct support or refer to external mental health services. However, our findings aligned with other research highlighting that teachers often have limited mental health literacy, skills and confidence with which to identify mental health needs and provide support [38–40]. Lack of training is a key barrier to teacher-delivered mental health support in different settings [39]. Teachers need training and guidance if they are to support the psychosocial wellbeing of children in schools, supporting the rationale for the guidelines developed.

The pilot phase was a key step in assessing the feasibility and acceptability of the guidelines. Positive feedback suggested that the guidelines included useful information, which was readily actioned by many of the teachers. However, it is important to recognise findings regarding teacher capacity and availability of time. Over half of the teachers in pilot two and 40% overall reported that it was difficult to use the guidelines alongside their current teaching duties. This is an issue seen with many school-based mental health initiatives; teachers' experience a conflict in their role, stretching their capacity [39, 41]. This is a crucial consideration when looking at the long-term viability and scalability of these guidelines in the Gaza Strip and other such mental health support structures in schools. Support of school leadership, both national and local, is needed to facilitate teachers in the delivery of mental health support [41]. Leadership must provide appropriate time and resources to teachers, to ensure that they are not overburdened. This includes hiring enough school counsellors to provide support to teachers aiming to promote wellbeing in their classrooms, as well providing more direct support to children who experience mental distress. In this development process, we included the Palestinian Ministry of Education and Higher Education from the outset, to encourage buy-in and input to the content and scale-up of the guidelines. Their support to schools and teachers is vital to the viability of guideline use. Further, the results of the pilot highlight the value of the half-day orientation and ongoing mentorship from school counsellors to support teachers to implement the guidelines. Building in these systems is important when scaling-up to all schools. As mentioned, Atfaluna Society for Deaf Children are training school staff, including school counsellors, on psychosocial support and use of the guidelines, in order to promote sustainable and effective utilisation of the guidelines.

With the guidelines being rolled-out to schools, it is important that national actors recognise that many deaf and hard of hearing children in the Gaza Strip may not be attending school and will not benefit from the support provided in these guidelines. Estimates from the

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0002427 October 16, 2023

Population, Housing and Establishments Census 2017 suggest that 35% of adolescents with disabilities aged 16+ have never enrolled in school [8]. It is imperative, therefore, that additional strategies are developed to support the mental health and wellbeing of deaf and hard of hearing children without access to educational settings.

Challenges and mitigations in the guideline development process

The data collection and development process were impacted by the COVID-19 pandemic as well as periods of heightened conflict in the Gaza Strip, namely May 2021. In the context of lockdown measures, online data collection was considered for the qualitative research. However, we believed this would reduce rapport and therefore data quality. Further, this may have been challenging for children and some deaf and hard of hearing participants using a sign language interpreter. When lockdown measures were lifted, we interviewed participants in-person, following local safety guidance. With international travel limited, the UK researchers could not travel to the Gaza Strip and therefore they supported remotely, training the local researcher and attending interviews over the internet (using Zoom), utilising ICED qualitative remote data collection tools to structure the process [42].

The initial planned pilot was also disrupted, following increases in COVID-19 cases and lockdown measures in 2021. However, as a result we conducted two pilots to understand the feasibility and acceptability of the guidelines. School closures also highlighted the need for us to include information on how teachers can continue to provide MHPSS remotely, in case of future lockdowns. This is particularly relevant in a context of regular crises impacting on the delivery of education.

Strengths of this guideline development process

As recommended in the GIN-McMaster checklist, we formed a multi-disciplinary steering committee, with expertise and knowledge on supporting the inclusion and wellbeing of deaf and hard of hearing children in the Gaza Strip. This steering committee was instrumental in guiding the development and content of the guidelines, providing both input and critical review. Utilising their knowledge throughout the process encouraged discussion and consensus on the tools that would best equip teachers to support deaf and hard of hearing children in their classrooms. Utilising participatory approaches in the workshops was vital to their success [26].

The appropriateness of the content to the needs of deaf and hard of hearing children was achieved by generating evidence on best practice. Conducting qualitative research enhanced our understanding of the context for deaf and hard of hearing children in the Gaza Strip, enabling tailored initiatives in the final guidelines. This qualitative research further added to the literature on mental health and wellbeing for deaf and hard of hearing populations in low-and middle-income countries, an area lacking in evidence.

The evaluation of the guidelines further strengthened the appropriateness of the guidelines for use by teachers in the Gaza Strip and resulted in a real-world understanding of its use in practice and what teachers could reasonably achieve in their classrooms. Having the guidelines peer-reviewed by leading deaf and hard of hearing global stakeholders added further validity to the guidelines and has created an opportunity for these guidelines to be adapted to other contexts and settings, which we encourage.

Limitations of the guideline development process

The sample size of the pilot was small and it would have been valuable to conduct a larger scale pilot to understand feasibility and acceptability. Further, each pilot could only run for three to

four weeks (as a result of COVID-19 restrictions and heightened conflict in the region). Teachers reported needing more time to utilise the guidelines and we do not have an understanding about sustained use of the guidelines.

We also did not have the capacity or time to conduct in-depth qualitative assessment during the pilot and were restricted to feedback through questionnaires and the workshops. Although these included open-ended questions, the understanding of the guidelines in practice would have been strengthened by further in-depth qualitative assessment with teachers. Additional qualitative research with deaf and hard of hearing children about their experience of guideline implementation would have further strengthened our understanding.

Future of the guidelines

The MHPSS guidelines are being rolled-out across mainstream and special schools in the Gaza Strip, in partnership with the Palestinian Ministry of Education and Higher Education. These guidelines are freely available online in English and Arabic, and we encourage governments and service providers in similar settings to adapt the guidelines to their context.

This study and development process identified a need for information and guidance on inclusive learning environments, communication, assistive technology, social inclusion, emotional literacy, and partnership with parents. These findings are consistent with previous literature from different settings, highlighting common needs in the context of MHPSS for deaf and hard of hearing children [35, 43–47]. However, it is important to respond to these needs in a context-specific manner, and adapting the guidelines developed in this study should include appropriate research methodologies to ensure contextual relevance. When adapting to other settings, it is important to do so in partnership with deaf and hard of hearing groups, parents, teachers, and other key stakeholders.

Future research

We recommend conducting a longer-term assessment of guideline feasibility and acceptability. Assessing deaf and hard of hearing child outcomes, including social inclusion and mental health status, would also be beneficial. In addition, it would be useful to conduct a process evaluation on the guideline roll-out procedures and long-term implementation.

Conclusion

This article has described the process of developing MHPSS guidelines for deaf and hard of hearing children in the Gaza Strip. This process was supported by a structured guideline development framework, and adoption of participatory approaches, evidence-generating activities and evaluation. The guidelines were found to be acceptable and feasible to implement, with support from school administration. These guidelines will aim to support to psychosocial wellbeing of deaf and hard of hearing children in the region and we encourage adaptation of these to other contexts.

Supporting information

S1 Appendix. Questionnaire on inclusivity in global health. (DOCX)

S2 Appendix. Qualitative participant characteristics. (DOCX)

Acknowledgments

We extend our deepest thanks to the steering committee for their guidance throughout this project. We further offer our thanks to the participants who participated in the research for generously giving their time and insight. Thank you also to those that provided input to and technical review of the guidelines. Further details on acknowledgements can be found within the guidelines themselves.

Author Contributions

Conceptualization: Nathaniel Scherer, Ramadan Hussein, Julian Eaton, Naim Kabaja, Ritsuko Kakuma, Sarah Polack.

Data curation: Nathaniel Scherer, Ramadan Hussein.

Formal analysis: Nathaniel Scherer, Tracey Smythe, Sarah Polack.

Funding acquisition: Nathaniel Scherer, Ramadan Hussein, Julian Eaton, Naim Kabaja, Sarah Polack.

Investigation: Nathaniel Scherer, Ramadan Hussein, Tracey Smythe, Sarah Polack.

Methodology: Nathaniel Scherer, Ramadan Hussein, Julian Eaton, Ritsuko Kakuma, Tracey Smythe, Sarah Polack.

Project administration: Nathaniel Scherer, Ramadan Hussein.

Resources: Nathaniel Scherer, Ramadan Hussein.

Software: Nathaniel Scherer.

Supervision: Julian Eaton, Naim Kabaja, Ritsuko Kakuma, Tracey Smythe, Sarah Polack.

Validation: Julian Eaton, Naim Kabaja.

Visualization: Nathaniel Scherer.

Writing - original draft: Nathaniel Scherer.

Writing – review & editing: Ramadan Hussein, Julian Eaton, Naim Kabaja, Ritsuko Kakuma, Tracey Smythe, Sarah Polack.

References

- United Nations Office for the Coordination of Humanitarian Affairs. Gaza Strip: The Humanitarian Impact of 15 Years of the Blockade. 2022 [cited January 16 2023]. Available from: https://www.unicef. org/mena/media/18041/file/Factsheet_Gaza_Blockade_2022.pdf
- United Nations Relief and Works Agency for Palestine Refugees in the Near East. Where We Work: Gaza Strip. 2023 [cited January 16 2023]. Available from: https://www.unrwa.org/where-we-work/gazastrip.
- Abu Hamad B, Gercama I, Bayoumi N, Jones N, Caprarelli P. "I prefer to stay silent." Exploring opportunities for and challenges to adolescents' psychosocial and mental health in Gaza. London: Overseas Development Institute; 2018.
- El-Khodary B, Samara M, Askew C. Traumatic Events and PTSD Among Palestinian Children and Adolescents: The Effect of Demographic and Socioeconomic Factors. Front Psychiatry. 2020; 11:4. https:// doi.org/10.3389/fpsyt.2020.00004 PMID: 32296346
- Elbedour S, Onwuegbuzie AJ, Ghannam J, Whitcome JA, Abu Hein F. Post-traumatic stress disorder, depression, and anxiety among Gaza Strip adolescents in the wake of the second Uprising (Intifada). Child Abuse Negl. 2007; 31(7):719–29. https://doi.org/10.1016/j.chiabu.2005.09.006 PMID: 17631959
- Khamis V. Post-traumatic stress and psychiatric disorders in Palestinian adolescents following intifadarelated injuries. Soc Sci Med. 2008; 67(8):1199–207. https://doi.org/10.1016/j.socscimed.2008.06.013 PMID: 18657343

- Abu Hamad B, Jones N, Gercama I. Adolescent access to health services in fragile and conflict-affected contexts: The case of the Gaza Strip. Conflict and Health. 2021; 15(1):40. https://doi.org/10.1186/ s13031-021-00379-0 PMID: 34020683
- Palestinian Central Bureau of Statistics. Characteristics of Individuals with Disabilities in Palestine: An Analytical Study Based on the Population, Housing and Establishments Census 2007, 2017. Ramallah, Palestine: Palestinian Central Bureau of Statistics; 2020.
- Bouldin E, Patel SR, Tey CS, White M, Alfonso KP, Govil N. Bullying and Children who are Deaf or Hard-of-hearing: A Scoping Review. Laryngoscope. 2021; 131(8):1884–92. https://doi.org/10.1002/ lary.29388 PMID: 33438758
- 10. Kral A O'Donoghue GM. Profound deafness in childhood. N Engl J Med. 2010; 363(15):1438–50.
- Roland L, Fischer C, Tran K, Rachakonda T, Kallogjeri D, Lieu JE. Quality of Life in Children with Hearing Impairment: Systematic Review and Meta-analysis. Otolaryngol Head Neck Surg. 2016; 155 (2):208–19. https://doi.org/10.1177/0194599816640485 PMID: 27118820
- 12. Ariapooran S. Relationship between mother- child interaction and symptoms of anxiety disorders in deaf and hard-of-hearing children. Empowering Exceptional Children. 2018; 9(2):51–63.
- 13. Fellinger J, Holzinger D, Pollard R. Mental health of deaf people. Lancet. 2012; 379(9820):1037–44. https://doi.org/10.1016/S0140-6736(11)61143-4 PMID: 22423884
- Fellinger J, Holzinger D, Sattel H, Laucht M, Goldberg D. Correlates of mental health disorders among children with hearing impairments. Dev Med Child Neurol. 2009; 51(8):635–41. https://doi.org/10.1111/ j.1469-8749.2008.03218.x PMID: 19627335
- Lawrence BJ, Jayakody DMP, Bennett RJ, Eikelboom RH, Gasson N, Friedland PL. Hearing Loss and Depression in Older Adults: A Systematic Review and Meta-analysis. Gerontologist. 2020; 60(3):e137– e54. https://doi.org/10.1093/geront/gnz009 PMID: 30835787
- Shoham N, Lewis G, Favarato G, Cooper C. Prevalence of anxiety disorders and symptoms in people with hearing impairment: a systematic review. Soc Psychiatry Psychiatr Epidemiol. 2019; 54(6):649– 60. https://doi.org/10.1007/s00127-018-1638-3 PMID: 30547211
- Stevenson J, Kreppner J, Pimperton H, Worsfold S, Kennedy C. Emotional and behavioural difficulties in children and adolescents with hearing impairment: a systematic review and meta-analysis. Eur Child Adolesc Psychiatry. 2015; 24(5):477–96. https://doi.org/10.1007/s00787-015-0697-1 PMID: 25758233
- Theunissen SC, Rieffe C, Kouwenberg M, De Raeve L, Soede W, Briaire JJ, et al. Anxiety in children with hearing aids or cochlear implants compared to normally hearing controls. Laryngoscope. 2012; 122(3):654–9. https://doi.org/10.1002/lary.22502 PMID: 22252674
- Theunissen SC, Rieffe C, Kouwenberg M, Soede W, Briaire JJ, Frijns JH. Depression in hearingimpaired children. Int J Pediatr Otorhinolaryngol. 2011; 75(10):1313–7. https://doi.org/10.1016/j.ijporl. 2011.07.023 PMID: 21862140
- van Gent T, Goedhart AW, Hindley PA, Treffers PD. Prevalence and correlates of psychopathology in a sample of deaf adolescents. J Child Psychol Psychiatry. 2007; 48(9):950–8. https://doi.org/10.1111/j. 1469-7610.2007.01775.x PMID: 17714380
- Scherer N, Smythe T, Hussein R, Wapling L, Hameed S, Eaton J, et al. Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip. PLOS Global Public Health. 2023:Forthcoming. https://doi.org/10.1371/journal.pgph.0001635 PMID: 37279194
- 22. Inter-Agency Steering Committee. IASC Guidelines on Mental Health and Psychosocial Support in Emergency Settings. Geneva: IASC; 2007.
- Margaretha M, Azzopardi PS, Fisher J, Sawyer SM. School-based mental health promotion: A global policy review. Frontiers in Psychiatry. 2023;14. https://doi.org/10.3389/fpsyt.2023.1126767 PMID: 37(39309)
- 24. Johns Hopkins Bloomberg School of Public Health and United Nations Children's Fund. On My Mind: How adolescents experience and perceive mental health around the world. Baltimore and New York: JHU and UNICEF; 2022.
- 25. United Nations Children's Fund. Five essential pillars for promoting and protecting mental health and psychosocial well-being in schools and learning environments: A briefing note for national governments. New York: UNICEF; 2022.
- Kuper H, Hameed S, Reichenberger V, Scherer N, Wilbur J, Zuurmond M, et al. Participatory Research in Disability in Low- and Middle-Income Countries: What have we Learnt and what Should we Do? Scandinavian Journal of Disability Research. 2021; 23:328–37.
- Schünemann HJ, Wiercioch W, Etxeandia I, Falavigna M, Santesso N, Mustafa R, et al. Guidelines 2.0: systematic development of a comprehensive checklist for a successful guideline enterprise. Cmaj. 2014; 186(3):E123–42. https://doi.org/10.1503/cmaj.131237 PMID: 24344144

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0002427 October 16, 2023

- Scherer N, Bright T, Musendo DJ, O'Fallon T, Kubwimana C, Eaton J, et al. Mental health support for children and adolescents with hearing loss: scoping review. BJPsych Open. 2021; 8(1):e9.
- Arksey H O'Malley L. Scoping studies: towards a methodological framework. International Journal of Social Research Methodology. 2005; 8(1):19–32.
- 30. Patton MQ. Qualitative research & evaluation methods. Saint Paul, MN: Sage; 2002.
- Braun V, Clarke V. Thematic analysis. APA handbook of research methods in psychology, Vol 2: Research designs: Quantitative, qualitative, neuropsychological, and biological. APA handbooks in psychology. Washington, DC, US: American Psychological Association; 2012. p. 57–71.
- Bowen DJ, Kreuter M, Spring B, Cofta-Woerpel L, Linnan L, Weiner D, et al. How we design feasibility studies. Am J Prev Med. 2009; 36(5):452–7. https://doi.org/10.1016/j.amepre.2009.02.002 PMID: 19362699
- Anderson LM, Shinn C, Fullilove MT, Scrimshaw SC, Fielding JE, Normand J, et al. The effectiveness of early childhood development programs: A systematic review. American Journal of Preventive Medicine. 2003; 24(3, Supplement):32–46.
- McGorry PD, Mei C. Early intervention in youth mental health: progress and future directions. Evidence Based Mental Health. 2018; 21(4):182. https://doi.org/10.1136/ebmental-2018-300060 PMID: 30352884
- Xie Y-H, Potměšil M, Peters B. Children Who Are Deaf or Hard of Hearing in Inclusive Educational Settings: A Literature Review on Interactions With Peers. The Journal of Deaf Studies and Deaf Education. 2014; 19(4):423–37. https://doi.org/10.1093/deafed/enu017 PMID: 25052819
- Wells J, Barlow J, Stewart-Brown S. A systematic review of universal approaches to mental health promotion in schools. Health Education. 2003; 103(4):197–220.
- Fazel M, Hoagwood K, Stephan S, Ford T. Mental health interventions in schools 1: Mental health interventions in schools in high-income countries. Lancet Psychiatry. 2014; 1(5):377–87. https://doi.org/10. 1016/S2215-0366(14)70312-8 PMID: 26114092
- Franklin CGS, Kim JS, Ryan TN, Kelly MS, Montgomery KL. Teacher involvement in school mental health interventions: A systematic review. Children and Youth Services Review. 2012; 34(5):973–82.
- O'Farrell P, Wilson C, Shiel G. Teachers' perceptions of the barriers to assessment of mental health in schools with implications for educational policy: A systematic review. British Journal of Educational Psychology. 2022. https://doi.org/10.1111/bjep.12553 PMID: 36300781
- Padmi Dhyah Y, Endang Retno S, Dwi Yuwono Puji S, Nurul H. Mental Healthy Literacy of Teachers: A Systematic Literature Review. Journal of Educational, Health and Community Psychology. 2021; 10 (2):339–50.
- Gee B, Wilson J, Clarke T, Farthing S, Carroll B, Jackson C, et al. Review: Delivering mental health support within schools and colleges–a thematic synthesis of barriers and facilitators to implementation of indicated psychological interventions for adolescents. Child and Adolescent Mental Health. 2021; 26 (1):34–46. https://doi.org/10.1111/camh.12381 PMID: 32543016
- 42. Banks L, Hameed S. Qualitative Researchers' Handbook: COVID-19 & disability–a multi-country study. London: International Centre for Evidence in Disability; 2021.
- Bouldin E, Patel SR, Tey CS, White M, Alfonso KP, Govil N. Bullying and Children who are Deaf or Hard-of-hearing: A Scoping Review. The Laryngoscope. 2021; 131(8):1884–92. https://doi.org/10. 1002/lary.29388 PMID: 33438758
- Constantinou V, Ioannou A, Klironomos I, Antona M, Stephanidis C. Technology support for the inclusion of deaf students in mainstream schools: a summary of research from 2007 to 2017. Universal Access in the Information Society. 2020; 19(1):195–200.
- Goldman SE, Burke MM. The Effectiveness of Interventions to Increase Parent Involvement in Special Education: A Systematic Literature Review and Meta-Analysis. Exceptionality. 2017; 25(2):97–115.
- 46. Rowan L, Bourke T, L'Estrange L, Lunn Brownlee J, Ryan M, Walker S, et al. How Does Initial Teacher Education Research Frame the Challenge of Preparing Future Teachers for Student Diversity in Schools? A Systematic Review of Literature. Review of Educational Research. 2020; 91(1):112–58.
- 47. Tristani L, Bassett-Gunter R. Making the grade: teacher training for inclusive education: A systematic review. Journal of Research in Special Educational Needs. 2020; 20(3):246–64.

PLOS Global Public Health | https://doi.org/10.1371/journal.pgph.0002427 October 16, 2023

20/20

View publication stats



Article

'Do they ever think about people like us?': The experiences of people with learning disabilities in England and Scotland during the COVID-19 pandemic

NATHANIEL SCHERER London School of Hygiene & Tropical Medicine, England

PHILLIPPA WISEMAN, NICHOLAS WATSON AND RICHARD BRUNNER University of Glasgow, Scotland

JANE CULLING WORTH University of Strathclyde, Scotland

SHAFFA HAMEED London School of Hygiene & Tropical Medicine, England

CHARLOTTE PEARSON

University of Glasgow, Scotland

TOM SHAKESPEARE

London School of Hygiene & Tropical Medicine, England

Abstract

People with learning disabilities in England and Scotland have experienced an increased risk of illness and death during the COVID-19

Corresponding author: Nathaniel Scherer, London School of Hygiene & Tropical Medicine, England. Email: nathaniel.scherer@lshtm.ac.uk

Critical Social Policy 1–25 © The Author(s) 2022 Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/02610183221109147 journals.sagepub.com/home/csp SAGE pandemic. Drawing on data of a longitudinal qualitative study with 71 disabled people and 31 disability organisations, this article examines the experiences of 24 people with learning disabilities in England and Scotland during the pandemic, reflecting on what rendered them vulnerable and placed them at risk. Qualitative interviews were conducted with participants and key informants at two timepoints; June-August 2020 and February-April 2021. Findings emerged across four key themes: failure to plan for the needs of people with learning disabilities; the suspension and removal of social care; the impact of the pandemic on people's everyday routines; and lack of vaccine prioritisation. The inequalities experienced by people with learning disabilities in this study are not particular to the pandemic. We explore the findings in the context of theoretical frameworks of vulnerability, including Fineman's conceptualisation of a 'vulnerability paradigm'. We conclude that the structured marginalisation of people with disabilities, entrenched by government action and inaction, have created and exacerbated their vulnerability. Structures, policies and action must change.

Key words

COVID-19, disability, health, learning disabilities, vulnerability

Introduction

The COVID-19 pandemic has had a catastrophic impact on the 1.5 million people with learning disabilities in the UK. People with learning disabilities have been more likely than the general population to contract COVID-19, and more likely to experience poorer health outcomes and mortality. In 2020, people with learning disabilities in Scotland were twice as likely to have severe symptoms and three times more likely to die of COVID-19 (Henderson et al., 2021). In England, people with learning disabilities had 8.2-fold higher rates of COVID-19 related death during the first wave (March-May 2020) and 7.2-fold higher rates during the second (September 2020-February 2021) (Williamson et al., 2021). The Learning Disabilities Mortality Review (LeDeR) indicated that 35% of people with learning disabilities who died from COVID-19 lived in residential care settings, and many who died experienced difficulties accessing COVID-19 tests, learning disability nurses and other healthcare support (LeDeR Programme, 2020). Baksh et al. (2021) further exemplified the disparities in healthcare experienced by people with learning disabilities, which have been 'magnified' during the pandemic (Baksh et al., 2021). The authors found people with learning disabilities more likely than the general population to present with severe

Scherer et al.

illness at the point of admission, corresponding to late presentation at hospital and postulated a result of poor symptom recognition, communication difficulties and exclusion from digital information and public health campaigns. Despite having more severe symptoms on admission, people with learning disabilities were less likely to access appropriate treatment, including intensive care units. Research led by Hastings and Hatton, and authored by Flynn et al. (2021), found severe reductions in social care and support for people with learning disabilities, associated with negative impacts on wellbeing and mental health, with two-thirds of their sample having experienced anger, sadness or anxiety (Flynn et al., 2021). Similar experiences for people with learning disabilities have been demonstrated internationality, with evidence emerging from the Netherlands, Ireland and Italy (Buonaguro and Bertelli, 2021; Embregts et al., 2020; McCausland et al., 2021). These experiences have renewed debate on what makes people with learning disabilities vulnerable to morbidity and mortality. In this paper, we explore this vulnerability in the context of the pandemic, with consideration of Fineman's conceptualisation and theory.

Health and social inequalities experienced by people with learning disabilities in the UK are not new or unique to COVID-19. LeDeR has demonstrated increased rates of mortality among people with learning disabilities in England every year since 2016, with care reported to fall short of expected good practice (LeDeR Programme, 2019). Research over recent years has shown that people with learning disabilities face daily exclusion and inequality, with evidence demonstrating higher risk of abuse, discrimination, isolation, loneliness, unemployment and long-term mental health conditions (Cooper et al., 2015; Emerson and Baines, 2011; Malli et al., 2018; Wiseman and Watson, 2021). In 2008, the Joint Committee on Human Rights presented the UK government with a striking report on the human rights violations experienced by people with learning disabilities, resulting from the neglect of government policy towards their health and social care, in such a way to dehumanise them and create a system that allows abuse, discrimination and indifference, perpetuating a life of isolation, poverty and social exclusion (Joint Committee on Human Rights, 2008). The report highlighted the failure of government and public authorities to improve the lives of people with learning disabilities, stemming from limited understanding, funding and commitment to human rights. Further, in 2016, an inquiry from the United Nations Committee on the Rights of Persons with Disabilities concluded that since 2010 UK Government policies had resulted in 'grave and systematic' violations on the rights of disabled people, including concern on evident negative attitudes and discrimination towards people with learning disabilities, high suicide rates among this population, limited employment opportunities for people with learning disabilities, and cases in which no attempt was made to resuscitate people with learning disabilities

(Committee on the Rights of Persons with Disabilities, 2016). Despite these damning reports, people with learning disabilities remain one of the most excluded groups in the UK.

The COVID-19 pandemic has presented a unique situation, in which the disadvantages and inequalities experienced by people with learning disabilities have been amplified and brought to the public's eye. In the early stages of the pandemic, attempts to curtail the spread of the virus meant that many of their support systems had to stop, as the country tried to minimise transmission. Large numbers of people with learning disabilities require multiple and intersecting forms of support across employment, education and wellbeing, and many have complex and additional needs, requiring social care support at home or day services. Most of these services closed in March 2020, with many still not fully open towards the end of 2021. With these closures and the disproportionate impact of the virus, people with learning disabilities found themselves at risk and without formalised support (Flynn and Hatton, 2021; Shakespeare et al., 2021; Pearson et al., 2022). Evidence from the first wave of the pandemic found people with learning disabilities socially isolated and worried about lost support, lost routine, and decreased health and wellbeing (Flynn et al., 2021).

Through longitudinal qualitative research with disabled people in England and Scotland, we sought to further understand the experiences of people with learning disabilities and their families during the pandemic, including responses from national government, local authorities and social care providers. Before presenting our findings, we explore the ideas of Martha Fineman and others to discuss the concept of vulnerability among people with learning disabilities, both during and before this pandemic.

Learning disability, vulnerability and COVID-19

Much of the debate on high mortality and morbidity from COVID-19 among people with learning disabilities has focused on vulnerability, and in particular, clinical vulnerability to the virus. Courtney and Cooper (2021), in an editorial in the British Medical Journal, drew together evidence from a range of studies to point to what they described as the 'extreme vulnerability' of this group to COVID-19 (Courtenay and Cooper, 2021). They argued that people with learning disabilities often have high levels of co-morbidity, many of which make them more clinically susceptible to COVID-19 and its effects. They also highlighted high levels of prejudice and discrimination within healthcare; for example, the way in which people with learning disabilities were disadvantaged in vaccine programmes.

In a rejoinder to this editorial, Hatton (2021) warned against and challenged the systemic use of the word 'vulnerable' in relation to people with learning disabilities and COVID-19. He argued that describing people with

Scherer et al.

learning disabilities as vulnerable to the virus 'locates the disproportionate risk of death from Covid 19 (and before Covid 19 too) as a property of individual people with learning disabilities. This not only removes agency from over a million UK citizens, but also can serve to remove any sense of urgency or even responsibility to see these inequalities for the injustices they are and to do something about them that matches the scale of the injustice.' (Hatton, 2021).

According to Hatton, the work cited in Courtney and Cooper's editorial focused not on clinical vulnerability, but on the way that society treats people with learning disabilities. It is this treatment, he argues, that makes people with learning disabilities 'vulnerable'. Hatton cites, as examples, the limited support received from health and social care during the pandemic, the discrimination of healthcare practitioners, and the living arrangements of people with learning disabilities, with large numbers placed in congregate settings, a hotspot for virus transmission. It is these social influences that have created the health inequalities experienced by people with learning disabilities during the COVID-19 pandemic. Hatton expresses the concerns of many disability theorists around the use of the term vulnerability, who argue that the concept may construct those deemed vulnerable as less able, less competent and more prone to harm (Scully, 2014).

Grear (2006) and Turner (2021) argue that all human beings are corporeally vulnerable, resulting not just from human embodied frailties, but through structural processes and institutions that produce social vulnerabilities (Grear, 2006; Turner, 2021). Further, Andorno (2016) defines 'vulnerability' as the ways in which some groups (in particular disabled people) are likely to experience greater harms due to their identities (Andorno, 2016). From Andorno's perspective, the purpose of understanding particular groups as vulnerable is not to construct those groups as inherently vulnerable, but rather to ensure greater protections and rights from state violence, discrimination and oppression. As such, vulnerabilisation can be understood as a social process and outcome of marginalisation, rather than being relegated to biomedical spheres and individual impairments.

We can further point to these social processes of vulnerabilisation through the work of Martha Fineman (2008) and the way she has deconstructed vulnerability to create what she termed the 'Vulnerability Paradigm' (Fineman, 2008). Fineman argues that focusing on acts of discrimination alone is too narrow; it produces an analysis of a particular action or moment of harm that privileges the individual and there is a danger that the 'historical, systemic and institutional structures that surround that moment' remain unexplored (Fineman, 2015). Equality has traditionally focused on fighting forms of discrimination with regards to race, gender, religion, etc. by providing the same treatment to all, in what Fineman has called the 'equal protection doctrine'. Fineman argues that this understanding of equality is inadequate
(Fineman, 2008). By highlighting individuals and individual actions it fails to address the impact of economic and social disadvantage on wellbeing. The equal protection doctrine does not challenge the underlying structures and practices. This is particularly harmful when applied to disability. Instead, Fineman calls for an approach that takes account of the context and structures that surround and create harm. Fineman argues that this is best achieved by focusing on differences rather than equality, by understanding the way in which different subjects are constructed in political and legal discourse. In recent years, Fineman's concepts have been increasingly applied to disability and learning disabilities (Clough, 2017; Heikkilä et al., 2020; Scully, 2014; Snipstad, 2021).

Fineman's approach starts from the basis that all people share common characteristics but that there are differences between them (Fineman, 2004). According to Fineman, we are all vulnerable; it is, she argues, part of the universal human condition. We are all susceptible to change and physical or social harm. Vulnerabilities are then embodied and embedded (Fineman, 2015; Fineman, 2017). Embodied differences are those that arise from biology, development, social relations or conventions. These include identity categories, such as ethnicity, gender and disability, and the way in which these have been constructed to create hierarchies and bias. People with learning disabilities are often excluded because they have been marked as incapable, inferior, weak or dangerous, for example. This exclusion is not universal; it is socially imposed, contingent and can take many forms (Scully, 2014). Vulnerabilities are also embedded in 'social relationships and within societal institutions' (Fineman, 2015). These economic, social, cultural and institutional relationships create embedded differences and it is these, Fineman argues, that shape our lives and create vulnerability. Emphasis is placed not on the individual, but on the structural, societal or institutional failings that cause vulnerability. People with learning disabilities are thus made vulnerable through social systems that exclude, actively harm and invisibilise, to render them devalued in everyday communities. These combine with and reinforce embodied vulnerabilities.

We have used Fineman's vulnerability paradigm as an overarching concept to explore the experiences of people with learning disabilities. In our analysis we have aimed to locate the experiences of people with learning disabilities during COVID-19 within the structures and practices that have rendered them vulnerable, placing them at greater risk.

Methods

The findings presented here are drawn from a subset of data collected as part of a longitudinal qualitative study into the experiences of disabled people in

England and Scotland. In this larger study, we conducted semi-structured, in-depth interviews with 71 disabled people and 31 key informants from disability support organisations across England and Scotland (Shakespeare et al., 2021). Two rounds of interview were held with this group of participants and key informants in June–August 2020, during the first wave of the pandemic, and February–April 2021, towards the end of the second. Of the 71 participants in the larger study, 24 were people with learning disabilities, caregivers or proxy respondents, from which the findings of this article are drawn.

Ethical approval for this study was obtained from the Research Ethics Committee at the London School of Hygiene & Tropical Medicine (Ref: 21878).

Participants

Participants were recruited through Disabled People's Organisations (DPOs) and other third sector organisations, via online advertisements and mail-outs. Participants were purposively selected, in order to maximise variation in line with Patton's sampling strategy, based on impairment (e.g. physical, sensory, intellectual), gender, age and geographic location (Patton, 1990).

Of the 71 disabled participants interviewed, 13 were adults with learning disabilities, six caregivers and proxy respondents of adults with learning disabilities, and five caregivers of children with learning disabilities. In total, 24 participants with learning disabilities were included (Table 1), along with 31 key informants based in organisations of and for disabled people.

Data collection

The interview guide was developed by the research team and adapted iteratively in response to participants' engagement in the first interviews. Questions explored the experiences of disabled people and their caregivers during the pandemic across key life areas including work, education, leisure, social care, health and the government pandemic response. The key informant interview guide included similar topics of relevance to the experiences of disabled people, but also focused on organisational and sector response, including changes in the organisation's work and member support, collaboration with government and other third sector organisations, and challenges in service provision.

Inclusive research

Social distancing measures were in place in the UK at the time of each interview and we adjusted research practices to maintain inclusivity (Walmsley and Johnson, 2004). Participants were provided the information sheet and consent

Variable	Number (%)
Participant	
Direct	13 (54%)
Proxy (caregiver)	11 (46%)
Location	
England	8 (33%)
Scotland	16 (77%)
Age	
<18	3 (13%)
18–65	20 (83%)
>65	1 (4%)
Gender	
Female	13 (54%)
Male	11 (46%)
Living arrangement before COVID-19	
Own home	6 (25%)
Family home	13 (54%)
Residential care	5 (21%)
Total	24 (100%)

Table 1. Sample characteristics.

form in easier to read formats during recruitment and were invited to interview via telephone, Zoom or email, depending on their preference. People with learning disabilities are a digitally excluded group and we worked with third sector organisations to recruit those without immediate access to online technology, ensuring their inclusion. For those that found telephone, Zoom or similar online communication technology inaccessible, we offered response via email or proxy. We encouraged participants to have a supporter present for the interview if helpful to them. Some interviews were conducted with a caregiver and self-advocate together. We also invited participants to speak with us a day or two before the scheduled interview, providing an opportunity to familiarise themselves with the online technology, to ask questions about the process and to get comfortable with the research team. Researchers used Easy Read materials to support the informed consent procedures. The methods used and described were trialled across six interviews and adapted in response to participant feedback. Each interview lasted between 30-60 min. All participants received a £20 voucher for taking part and accessible research summaries have since been provided to all.

Scherer et al.

Data analysis

Interviews were transcribed verbatim, anonymised and stored on a secure server. Each transcript was coded in NVivo 12. The research team developed an initial coding frame and adapted this as needed throughout the analysis using mechanisms of reflexive thematic analysis (Braun and Clarke, 2006). The teams at the London School of Hygiene & Tropical Medicine and the University of Glasgow cross-reviewed eight transcripts in the early stages of analysis to ensure consistent coding. During the analysis, themes were identified, reviewed and refined by the research team. Participant narratives and quotes are presented in this paper and have been anonymised through use of pseudonyms. Quotes are from people with learning disabilities, unless otherwise stated. The round of interview from which a quote is derived is noted by T1 (first round) and T2 (second round).

Findings

Findings emerged across four key themes. First, we explore the limited inclusion of people with learning disabilities in pandemic response strategies. Second, we examine how the suspension and removal of social care impacted on people with learning disabilities and their families. Third, we discuss the impact of the pandemic on people's everyday routines, which left many isolated. Finally, we explore the initial lack of vaccine support for people with learning disabilities and the fight for prioritisation.

Afterthought, lack of thought or no thought

From the outset of the pandemic, respondents described how people with learning disabilities were rarely considered in the national response. They felt invisible and ignored.

When I watched them [news briefings] with mum I never once heard the word learning disability. We were completely forgotten about.

((voice breaking and close to crying)) Do they ever think about people like us? When they mention the word 'shielding', you think about the elderly, the vulnerable. Aren't we vulnerable? They're not the only ones shielding. We're missing out on our social care and our healthcare because of the coronavirus, because of shielding. But could they care less? No. Do they care? No. (Kelly, England, T1) Government neglect was experienced not just in the early stages when the response could perhaps be expected to be uncertain, but throughout.

I think we've just been pushed underneath the carpet, because I don't think the government give a damn. As long as they're okay and they've got their wage packet, they don't give a – excuse my language here – they don't give a toss. (Sally, Scotland, T2)

Policy focus was given to hospitals and it is clear from our data that people with learning disabilities were not considered a priority. At the root of this problem was neglect and limited understanding on both their needs and the way that they live their lives. This was seen among both national and local actors.

I inveigled myself into a meeting of [Local Authority senior management] quite recently and I was just fascinated about how little they knew about people with learning disabilities and how little they knew about people in their communities who had learning disabilities.

(CEO, national learning disability organisation, Scotland, T2)

This lack of engagement and understanding is illustrated by the failure to provide information and guidance to people with learning disabilities in multiple and accessible formats.

Poppy: The government should give more information, that's what I think. They should give more information. They should get more information and I think they should explain it more to people with learning disabilities, 'cause some people don't understand what they're saying.

Basil: No, that is the problem, isn't it?

Poppy: That's what they should've... that's what they should have done. 'Cause I've found out that their words, that they're using big words, not little words. They don't break it down. (Poppy and Basil, England, T1)

As the pandemic progressed, there were attempts by governments to provide information in Easy Read format, but these were often only available online, making it difficult for digitally excluded people with learning disabilities to access. Inaccessible information resulted in misunderstanding and increased fear and anxiety, creating and deepening risk. Third sector organisations stepped in to provide accessible materials on the virus, with examples produced by Beyond Words and the Scottish Consortium for Learning Disabilities. Third sector organisations also hosted COVID-19 information sessions on Zoom and helped people with learning disabilities access the regular government briefings and other news stories.

Insufficient government consideration on the needs of people with learning disabilities impacted on the quality of care and support they received and it is to this we now turn.

Dissolution of care and support

At the start of the pandemic, many people with learning disabilities saw their social care services and packages cut. Often social care was removed overnight and with very little warning.

[name of provider], who provide most of Maurice's care, phoned me up and said:

'Obviously this virus is getting more serious, so we've had to prioritise the support we provide, and in two days, Maurice's care will be ceasing altogether, and it's over to you. You need to now cover all of it.'

So as you can imagine, that was a surprise. (Pearson et al., 2022) (Abby, sister of Maurice, Scotland, T1)

For the majority of our participants, support packages were cut, with no further word from providers as the pandemic progressed. Where social workers and others did get in touch, it was often cursory and had little effect. Some told us, for example, that the only contact they received was related to their annual review. This was seen by many as very poorly timed and insensitive, after no contact for months.

The day centre called me maybe about a month ago to say it was time for his review and could they perhaps come round to do it in the garden? And I said: 'Are you having a laugh here? You haven't seen him for five months. What is there to review?'. And they said that: 'We have to do it, we've got paperwork to do'. And I said: 'No, I'm sorry, it's just a tick box exercise'. (Kris, mother of Jim, Scotland, T1)

Although the governments talked often about increased payments for unpaid family carers, our respondents reported little additional funding being made available in practice, even at the second-round of interviews in 2021. In fact, many families were asked to continue paying for support, despite support being withdrawn.

The first wave of activity and the first wave of calls we received from families were about the withdrawal of social care support. In one or two instances we had a situation where some very well-known social care providers, who also are charities, not only withdrew support but asked the families to continue paying for support. Along the lines of saying to them: 'Well we need to keep your space open for your son or daughter so it would be better if you kept paying'. And these weren't isolated incidents, these were coming from a number of families across Scotland. (CEO, national learning disability organisation, Scotland, T2)

People also received little or no guidance from the government or local authority on how they could get support in their home, or how they could make that support COVID secure. We were told of occasions where staff who normally worked with people with learning disabilities were transferred to work in the care of older people. As the Care Manager of a service in Scotland told us, her service provided support to 50 people with learning disabilities before the pandemic. At the time of interview in June 2020, they supported only 14 people. The rest were left unsupported.

This shift of care staff meant a lack of consistency in the delivery of care. Where support was provided, the workers were often changed at short notice, putting people with learning disabilities at increased risk. Participants had to develop their own guidelines and procedures on COVID-19 measures in their home, and in many cases, the provision of personal protective equipment (PPE) for staff in the domiciliary sector was hard to source. It was also, for some, hard to persuade staff to wear PPE, and unlike in care homes and hospitals, it does not appear that guidelines were provided for domiciliary care staff. If they were, they were not well disseminated and there was limited accountability.

Participants often had to fight with local authorities to restart their support packages and some of our participants had still not seen these reinstated by early 2021. Throughout the pandemic, there was an expectation from governments, local authorities and service providers that families would be able to revert to becoming chief providers of care. For many family members this produced stress, burnout and mental health concerns, as they tried to juggle 24/7 care, with no support or respite. One mother told us of how she was accidentally copied into the minutes of a social care provider, who had discussed her request for short break care. They were going to refuse, as they believed she, as a mother, needed to be doing more. All of this despite her caring for her son 24/7 for five months, without a break.

It made me understand that there was a complete lack of awareness of caring for someone in your own home in a national lockdown, 24 hours 7 days a week and not even being able to go out to work. That's unforgiveable. (Donna, mother of Ryan, England, T2)

Most participants expressed grave concern that cuts to social care packages would continue once the pandemic was over, as families had effectively 'proven' that they could undertake the role of the state, despite substantial socio-economic impact and impact on wellbeing.

I feel as well that both social and healthcare services have a very limited - I'm going to use limited - understanding of what challenges people with learning disabilities have. I think they feel that it's like people with disabilities are making special demands... wanting special treatment. But actually that's not really it, they just want the same outcomes as other people. They just want to live their lives.

(Abby, sister of Maurice, Scotland, T2)

Ultimately, participants felt abandoned. They were asked to fend for themselves and often without the proper means. Providers were rarely in touch, despite payment for support frequently remaining in place. In their joint interview Kris, the mother of Jim, a 30-year-old man with learning disabilities who lived with her, told us how, even by March 2021, nobody had been in touch with them:

I've never been asked the question if it's okay, if we're doing okay... Nobody called me to see if I was managing without respite for a year, so from that point of view, I wouldn't say that the local authority had looked after us. I think it's been a case of, if we were in crisis, they would expect that we would get in touch, and because we haven't, they're assuming everything's alright. (Kris, mother of Jim, Scotland, T2)

Participants told us one of the major issues throughout the pandemic was the lack of consultation with them and their families from government, local authorities and social care services about their social care needs, with most having to react as best they could to a top-down approach, which resulted in withdrawn and inadequate provision.

Disrupted routines

The suspension of services and support disrupted daily routines. Of course, everyone's normal routines were disrupted during the pandemic, but the impact of closures on the lives of people with learning disabilities was profound. For many of our participants, activities such as drama groups and book clubs provided structure, routine, and security, in otherwise precarious and isolated lives. Day services also acted as platforms to socialise, and many reported missing their friends and connections regularly seen at these settings. With these closed during the pandemic, many participants felt isolated and lonely, with a sense of loss and worthlessness. Some felt they had lost their independence, and now relied too much on family.

I also feel my independence has gone, due to now I am always with my parents or if we go out they are always supporting me but I'm not having my independence. Before I was out on my own and using public transport.

Also my freedom of choice is reduced as less opportunities to choose that are of interest to me.

Now I am transported by cars all of the time by my parents. It's very hard as they changed daily depending on how I am coping.

I often feel sad no clubs and hobbies are available yet. (Frances, Scotland, T2)

These feelings of isolation and lost independence led some to show increased instances of distress; for example, through challenging behaviours, including violence towards themselves and family members.

That was really hard, you know, people were saying that they felt they had lost their children, that their child now was so low in mood and, you know, lethargic, not carrying out activities. Self-injurious behaviours, communicating in adverse ways, had put on weight, lost skills, not able to do the things that they used to be able to do. And a couple of families saying: 'I don't know whether I'll ever have my son back again'. Oh, I mean, honestly it was enough to bring you to tears really. (CEO, national learning disability organisation, Scotland, T2)

Many expressed concern that without the stimulation provided by these services, including the social interaction central to activities, they would lose hard fought life skills. Caregivers were concerned that they could not provide as stimulating an environment or the variety of experiences as these services and as a consequence were seeing their loved ones regress.

It's a real concern to, well, most of us, that it might just never reopen, and that has been suggested as a possible case scenario. That is a real worry, because as I was saying earlier, Lucy's health and wellbeing have been impacted on, because she hasn't been able to go to her day service. She needs to be out and about, she needs to be doing, she needs more challenge than she gets from just staying in the house all the time. (Priscilla, mother of Lucy, Scotland, T2)

With the cycle of routine broken, many were less stimulated and expressed anxiety about being able to return to their pre-pandemic routine. As the pandemic progressed, the strains and demands of lockdown started to deepen. Mental health support was, however, severely disrupted. It was removed for many, or was offered only online, inaccessible to many people with learning

14

Scherer et al.

disabilities. One mother in England asked for mental health support from Child and Adolescent Mental Health Services (CAMHS) in March 2020, at the start of lockdown, but she did not receive a response until October 2020, with services offered only by Zoom, which was inaccessible to her son. They asked for reasonable accommodation but were told this was not possible. She paid £2000 for private mental health care instead and her son has shown substantial improvements in his wellbeing. Private mental health support is only available to a small minority of the general and learning disability population, however.

Again, third sector organisations stepped in with online activities when services closed, including online book clubs, cooking courses and quizzes, as well as mental health support lines, which were all received very positively by our participants. To overcome digital exclusion, third sector organisations provided training on how to use Zoom and some were able to provide digital tablets and laptops to those in need. That said, we were told that many people with learning disabilities remained digitally excluded throughout the pandemic.

At the second round of interviews, some had begun to see their day services and other activities resume at reduced hours, and after a few 'teething' issues on return, some had seen a boost to their mental wellbeing.

They were really supportive within the parameters they could work in, which for a long time was no access to the day centre whatsoever, which was hellish for my brother. And they've gone back to two days a week. So, he has some access then, which has given him a bit of a routine and a bit of a sense of purpose, and his mental health has levelled out a bit. So that's helpful. (Alice, sister of Michael, Scotland, T2)

As well as social care, health management routines were also severely impacted during the pandemic. Routine health checks were suspended or delayed with consequences to the everyday health of participants. Day centre closure contributed to reduced mobility and physical health, with many of the participants missing out on organised activities, such as swimming. There were further concerns from participants with regards to foot care, dental care and diet management. Participants described how the pandemic had adversely affected the management of their epilepsy and control of their diabetes, for example.

Delayed vaccine priority

There was also concern expressed with regards to the vaccine and protection from COVID-19. For months, people with learning disabilities were ignored for vaccine priority, despite early evidence that they had a high-risk of morbidity and mortality from the disease (Mahase, 2021; Public Health England, 2020). Self-advocates, families and third sector organisations had to fight for priority status in the vaccination programme, with a number of coordinated campaigns directed towards the government.

I have a simple example of that when I was – I'm not sure I was lobbying – but I was in a meeting with the Scottish Government's lead for vaccination... I said: 'So what I'm actually trying to help with here is the efficacy of your administration of your vaccination programme'. And the response I got was: 'that would be taking somebody else's vaccine'. And I thought, you're not even listening to the argument, you're not even engaging with me on this. It's similar to that [previous] response: 'but there's only 8000 of them' [people with Down Syndrome]. The human rights of people with learning disabilities have been absolutely trampled on. (CEO, national learning disability organisation, Scotland, T2)

Only after a media storm, brought into the public's view by well-known relatives of people with learning disabilities, such as Ian Rankin and Jo Wiley, were people with learning disabilities eventually given priority. For many of our participants, their vaccination was a good experience. However, for others, there were issues. The vaccination programme prioritised people with learning disabilities registered with a GP, but people with learning disabilities frequently fall through the net and they are less visible to health services (Buszewicz et al., 2014). Many of our participants were not invited for the vaccine because they were not registered as having learning disabilities. Many had to contact their GP themselves to arrange their vaccine and many expressed confusion with the system and the eligibility criteria.

For others who received the vaccine, the experience itself was a challenge. Although people with learning disabilities had been placed in the priority group, there were often no accommodations made at vaccination centres. Information on the vaccine process was not provided in an accessible format, and often those administering had no experience working with a person with additional needs. Donna told us how her son Ryan refused the vaccine because there were no accommodations made. She asked for support from a learning disability nurse, Easy Read materials, a private space and a longer appointment. These were denied.

It's the same – there's no facility. It's a backhanded compliment, saying all these people with autism or learning disability can go and have their vaccine, but they're not going to take it up if they can't access it. It's so cruel really. There would be uproar, wouldn't there, with no wheelchair access to a building for vaccination. But it's exactly the same - accessibility. (Donna, mother of Ryan, England, T2)

Inaccessible vaccine procedures and delayed prioritisation increased vulnerability to the virus. Uptake of vaccinations by people with learning disabilities has been historically low (flu vaccine uptake was just 58% in 2019 to 2020) and the findings of our study reinforce the need for inclusive, targeted programmes (Public Health England, 2021).

Discussion

Our findings leave no doubt that the pandemic and the measures introduced to stop the spread of the virus had a significant impact on the lives and wellbeing of people with learning disabilities in England and Scotland. Government action (and inaction) exposed people with learning disabilities to risk and to harm. The pandemic revealed how poorly their needs are understood and how rarely people with learning disabilities are considered in policy planning.

Based on these findings, this paper has explored how people with learning disabilities have been made vulnerable during the COVID-19 pandemic. These vulnerabilities are both embodied, in that they are underpinned by greater risk of morbidity and mortality, and also embedded in social processes and practices. Vulnerabilities are produced not just because of people with learning disabilities' increased risk of acquiring and dying from the virus, but also because of inappropriate, inadequate and discriminatory social structures, and the processes and practices that they are embedded within. As Scully (2014) argues, vulnerabilities associated with an impairment can be 'amplified through structural and institutional processes that distribute unequally the resources that people might use to shield themselves and foster resilience against the impact of disability' (Scully, 2014). Our findings demonstrate the social production of vulnerability experienced by people with learning disabilities.

People with learning disabilities in England and Scotland were, initially at least, abandoned in the Westminster and Scottish governments' response to the COVID-19 pandemic. The pandemic exposed the limited understanding and neglect of governments and other statutory agencies of the needs and lives of people with learning disabilities and their families. While the marginalisation of people with learning disabilities is not new, they have, in this pandemic, faced the same crisis in a new form. To date, almost all of the official dialogue surrounding people with learning disabilities and COVID-19 has focused on their clinical vulnerability to the virus, largely ignoring the social and cultural practices that have placed them at increased risk. As we look back over the course of the pandemic and our longitudinal data, we see a focus from government on the biomedical, with little consideration for social dimensions and the impact these have on health and wellbeing.

Our findings suggest that the marginalisation and vulnerabilisation of people with learning disabilities has been the result of government action, or rather, inaction, entrenching pre-existing inequitable social structures (Fineman, 2008; Hatton, 2021). The removal of social care during COVID-19 presents the clearest of example of this. Throughout the pandemic, social care was seen as secondary to health care; disabled people's social care needs were neglected and restricted by the UK and Scottish governments (Pearson et al., 2022; Shakespeare et al., 2021). In a 2020 analysis of national government policy response during the first wave of the pandemic, the Health Foundation concluded that government support for social care came too late and faced widespread implementation problems (Dunn et al., 2020). This led to limited protection and support for people using adult social care, increasing unmet need for social care (Dunn et al., 2020). Removing care increased demands on people with learning disabilities and their families at a time when support was most needed, causing stress, isolation, anxiety and fear. This neglect came after years of austerity-imposed erosion in social care, which was, as noted by the Health Foundation, 'underfunded, understaffed, undervalued and at risk of collapse', prior to the pandemic (Dunn et al., 2020). This neglected social care, both before and during the pandemic, has deepened legacies of exclusion and disregard for the health, wellbeing and rights of people with learning disabilities (Glasby and Needham, 2020; Martinelli, 2017; Pearson et al., 2022). Further, inaccessible COVID-19 information, delayed vaccination priority, and limited care guidance for families and providers all increased vulnerabilities experienced by people with learning disabilities, impacting their wellbeing and quality of life. Our findings are consistent with other evidence generated during the pandemic; government inaction and indifference exacerbated the pre-existing vulnerabilities of people with learning disabilities during the pandemic, whilst producing new precarities, putting their lives and wellbeing at risk (Flynn et al., 2021; Armstrong and Pickering, 2020).

In September 2021, the UK government announced plans to review strategies, policies and funding for health and social care in England. In Scotland, the Feeley Review into Adult Social Care is currently under consultation (Scottish Government, 2021). If they are to have the necessary impact, these reviews must transform the existing chronically underfunded system, rather than prop up that which already exists (Pearson et al., 2022). Governments must also prioritise areas that contribute to wider health inequalities experienced by people with learning disabilities. Emerson and Hatton have identified causes of health inequalities for people with learning disabilities, from which the government must learn and act (Emerson and Hatton, 2014). Key amongst these are the social determinants of health, which include social factors such as socioeconomic disadvantage, inadequate housing, discrimination, isolation, exclusion and violence. These are central to Fineman's model of vulnerability, and have played a key role in creating the disadvantages experienced by people with learning disabilities during the pandemic.

Scherer et al.

Entwined in this process of transformation is the need for governments to work in co-production and partnership with people with learning disabilities and their families. In part, the failure to account for people with learning disabilities during the pandemic arose because of limited consultation when planning the response, resulting in abrupt and inappropriate actions. Not only did the policies not meet the needs of this group, they served to amplify their disadvantage. People with learning disabilities were placed at increased risk because governments failed to work in partnership with them, their families and DPOs. Such engagement would have been relatively easy; there are large numbers of DPOs and other third sector organisations that could have acted as proxies at the start of the pandemic. As we discussed in an earlier article, these non-statutory agencies played a central role in covering the inaction of central government (Shakespeare et al., 2021; Cullingworth et al., 2021). DPOs across England and Scotland have called for increased funding, support and responsibility going forwards, and there must be recognition of the value they play in helping disabled people access their rights (Inclusion London, 2020; Glasgow Disability Alliance, 2020; Inclusion Scotland, 2020; Greater Manchester Disabled People's Panel, 2020).

Over the last two decades, governments across the UK have received evidence, reports and advocacy on the systematic inequalities experienced by people with learning disabilities, stemming from limited understanding, neglect and indifference (Health and Social Care Committee, 2021; Joint Committee on Human Rights, 2008; Mencap, 2007; Scior and Werner, 2015; Simmonds et al., 2018). They have had a long time to act, but have shown little willingness to do so. In December 2021, the UK Government published a follow-up report to the 2016 inquiry by the United Nations Committee on the Rights of Persons with Disabilities, covering actions in 2020 and 2021 taken in line with United Nations recommendations for welfare reform, accessible information and consultation (Department for Work and Pensions, 2021). The UK Government's reported actions to these recommendations do not, however, reflect the information told to us by people with learning disabilities and their families, and as seen in numerous pieces of research conducted during COVID-19. Independent inquiry, review and accountability continue to be needed to ensure that governments in the UK transform the health and social landscape into one that empowers people with learning disabilities to reduce health inequalities, vulnerabilities and help them to reach their potential.

In our analysis and by employing Fineman's vulnerability paradigm we have shown how the response of the governments has increased risk and vulnerabilised people with learning disabilities, exacerbating existing vulnerabilities and imposing new. Given this, it is not surprising that people with learning disabilities have suffered poor health outcomes and wellbeing during the COVID-19 pandemic. These inequalities are not new and urgent reform is needed; for people with learning disabilities, returning to the status quo is not an option. Governments must work in partnership with people with learning disabilities, families and disability organisations to build back a better and fairer society, addressing the challenges that contribute to poorer health, wellbeing and quality of life. In order to understand and respond to the harms that people with learning disabilities have experienced throughout the COVID-19 pandemic, we need to pay attention to the ways in which they are made vulnerable, the processes involved in making them vulnerable and the consequences on their lives.

Acknowledgements

Firstly, we thank the study participants who generously gave their time to take part in the study. Thank you also to Nicola Burns and Veronika Reichenberger for supporting the research and dissemination.

Declaration of conflicting interests

The authors declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The authors disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This work was supported by UK Research and Innovation (grant number ES/V004069/1).

References

- Andorno R (2016) Is vulnerability the foundation of human rights? In: Masferrer A and García-Sánchez E (eds) Human Dignity of the Vulnerable in the Age of Rights: Interdisciplinary Perspectives. Cham: Springer International Publishing, pp. 257– 272.
- Armstrong S and Pickering L (2020) Left Out and Locked Down. Glasgow: University of Glasgow.
- Baksh RA, Pape SE, Smith J, et al. (2021) Understanding inequalities in COVID-19 outcomes following hospital admission for people with intellectual disability compared to the general population: A matched cohort study in the UK. BMJ Open 11(10): e052482.
- Braun V and Clarke V (2006) Using thematic analysis in psychology. Qualitative Research in Psychology 3(2): 77-101.
- Buonaguro EF and Bertelli MO (2021) COVID-19 and intellectual disability/autism spectrum disorder with high and very high support needs: Issues of physical and mental vulnerability. *Advances in Mental Health and Intellectual Disabilities* 15(1): 8–19.

- Buszewicz M, Welch C, Horsfall L, et al. (2014) Assessment of an incentivised scheme to provide annual health checks in primary care for adults with intellectual disability: A longitudinal cohort study. *The Lancet Psychiatry* 1(7): 522–530.
- Clough B (2017) Disability and vulnerability: Challenging the capacity/incapacity binary. *Social Policy and Society* 16(3): 469–481.
- Committee on the Rights of Persons with Disabilities (2016) Inquiry concerning the United Kingdom of Great Britain and Northern Ireland carried out by the Committee under article 6 of the Optional Protocol to the Convention. Geneva: United Nations Committee on the Rights of Persons with Disabilities.
- Cooper SA, McLean G, Guthrie B, et al. (2015) Multiple physical and mental health comorbidity in adults with intellectual disabilities: Population-based cross-sectional analysis. *BMC Family Practice* 16: 110.
- Courtenay K and Cooper V (2021) COVID 19: People with learning disabilities are highly vulnerable. *BMJ* 374: n1701.
- Cullingworth J, Waston N and Shakespeare T. (2021) 'They Have Been a Saving Grace' - The Role of the Third Sector in Disabled People's Experiences of COVID-19 and Implications for Third Sector-State Relations. In: 2021 Voluntary Sector and Volunteering Research Conference, Birmingham, UK, 6–7 Sept 2021. Birmingham: Voluntary Sector Studies Network.
- Department for Work and Pensions (2021) UN Committee on the Rights of Persons with Disabilities 2016 Inquiry – UK 2021 Follow-up Report. London: UK Government.
- Dunn P, Allen L, Humphries R, et al. (2020) Adult Social Care and COVID-19: Assessing the Policy Response in England so far. London: Health Foundation.
- Embregts PJCM, van den Bogaard KJHM, Frielink N, et al. (2020) A thematic analysis into the experiences of people with a mild intellectual disability during the COVID-19 lockdown period. *International Journal of Developmental Disabilities*. Epub ahead of print 05 October 2020. DOI:10.1080/ 20473869.2020.1827214.
- Emerson E and Baines S (2011) Health inequalities & people with learning disabilities in the UK. *Tizard Learning Disability Review* 16: 42–48.
- Emerson E and Hatton C (2014) Health Inequalities and People with Intellectual Disabilities. New York, NY: Cambridge University Press.
- Fineman M (2004) The Autonomy Myth: A Theory of Dependency. New York, NY: The New Press.
- Fineman M (2008) The vulnerable subject: Anchoring equality in the human condition. Yale Journal of Law & Feminism 20(1): 8-40.
- Fineman M (2015) Equality and difference the restrained state. *Alabama Law Review* 66(3): 609–626.
- Fineman M (2017) Vulnerability and inevitable inequality. Oslo Law Review 4(3): 133-149.
- Flynn S, Bailey T, Hastings R, et al. (2021) Coronavirus and People with Learning Disabilities Study Wave 1 Results: March 2021 (Full Report). Coventry: University of Warwick.

- Flynn S and Hatton C (2021) Health and social care access for adults with learning disabilities across the UK during the COVID-19 pandemic in 2020. *Tizard Learning Disability Review* 26(3): 174–179.
- Glasby J and Needham C (2020) The neglect of adult social care during COVID-19. *BMJ* 370: m3103.
- Glasgow Disability Alliance (2020) Supercharged: A Human Catastrophe. Inequalities, Participation and Human Rights Before, During and Beyond COVID19. Glasgow: GDA.
- Grear A (2006) Human rights human bodies? Some reflections on corporate human rights distortion, the legal subject, embodiment and human rights theory. *Law and Critique* 17(2): 171–199.
- Greater Manchester Disabled People's Panel (2020) GM Big Disability Survey Covid-19 Report. Manchester: GMDPP.
- Hatton C (2021) COVID 19: People with learning disabilities are highly vulnerable; rapid response: People with learning disabilities and COVID 19 - vulnerability. *BMJ* 374: n1701.
- Health and Social Care Committee (2021) The Treatment of Autistic People and People with Learning Disabilities. London: United Kingdom Parliament.
- Heikkilä M, Katsui H and Mustaniemi-Laakso M (2020) Disability and vulnerability: A human rights reading of the responsive state. *The International Journal of Human Rights* 24(8): 1180–1200.
- Henderson A, Fleming M, Cooper S-A, et al. (2021) COVID-19 infection and outcomes in a population-based cohort of 17,173 adults with intellectual disabilities compared with the general population. *Journal of Epidemiology and Community Health* 76(6): 550–555.

Inclusion London (2020) Abandoned, Forgotten and Ignored. London: Inclusion London.

- Inclusion Scotland (2020) Rights at Risk: Covid-19, Disabled People and Emergency Planning in Scotland. Edinburgh: Inclusion Scotland.
- Joint Committee on Human Rights (2008) A Life Like Any Other? Human Rights of Adults with Learning Disabilities. London: United Kingdom Parliament.
- LeDeR Programme (2019) Annual Report 2019. Bristol: University of Bristol.
- LeDeR Programme (2020) Deaths of People with Learning Disabilities from COVID-19. Bristol: University of Bristol.
- Mahase E (2021) COVID-19: All adults on learning disability register should be prioritised for vaccination, says advisory committee. *BMJ* 372: n547.
- Malli MA, Sams L, Forrester-Jones R, et al. (2018) Austerity and the lives of people with learning disabilities. A thematic synthesis of current literature. *Disability & Society* 33(9): 1412–1435.
- Martinelli F, Anttonen A and Mätzke M (2017) Social Services Disrupted Changes, Challenges and Policy Implications for Europe in Times of Austerity. Cheltenham: Edward Elgar Publishing.
- McCausland D, Luus R, McCallion P, et al. (2021) The impact of COVID-19 on the social inclusion of older adults with an intellectual disability during the

first wave of the pandemic in Ireland. *Journal of Intellectual Disability Research* 65(10): 879-889.

Mencap (2007) Death by Indifference. London: Mencap.

- Patton MQ (1990) *Qualitative Evaluation and Research Methods*. Thousand Oaks, CA: SAGE.
- Pearson C, Watson N, Brunner R, et al. (2022) COVID-19 and the crisis in social care: Exploring the experiences of disabled people in the pandemic. *Social Policy and Society*. Epub ahead of print 08 April 2022. DOI:10.1017/ S1474746422000112.
- Public Health England (2020) COVID-19: Deaths of People with Learning Disabilities. Reportno. London: PHE.
- Public Health England (2021) Seasonal Influenza Vaccine Uptake in GP Patients: Winter Season 2020 to 2021. London: PHE.
- Scior K and Werner S (2015) Changing Attitudes to Learning Disability. London: Mencap.
- Scottish Government (2021) Independent Review of Adult Social Care in Scotland. Edinburgh: Scottish Government.
- Scully JL (2014) Disability and vulnerability: On bodies, dependence, and power. In: Mackenzie C, Rogers W and Dodds S (eds) Vulnerability: New Essays in Ethics and Feminist Philosophy. Oxford: Oxford University Press, pp. 204–221.
- Shakespeare T, Watson N, Brunner R, et al. (2021) Disabled people in Britain and the impact of the COVID-19 pandemic. Social Policy & Administration 56(1): 103–117.
- Simmonds R, Burke C, Ahearn E, et al. (2018) A Life Without Fear? A Call for Collective Action Against Learning Disability Hate Crime. London: Mental Health Foundation.
- Snipstad ØIM (2021) Concerns regarding the use of the vulnerability concept in research on people with intellectual disability. *British Journal of Learning Disabilities* 50(1): 107–114.
- Turner BS (2021) Vulnerability and Human Rights. Pennsylvannia, PA: Penn State University Press.
- Walmsley J and Johnson K (2004) Inclusive Research with People with Learning Disabilities: Past, Present and Futures. London: Jessica Kingsley Publishers.
- Williamson EJ, McDonald HI, Bhaskaran K, et al. (2021) Risks of COVID-19 hospital admission and death for people with learning disability: Population based cohort study using the OpenSAFELY platform. *BMJ* 374: n1592.
- Wiseman P and Watson N (2021) 'Because I've got a learning disability, they don't take me seriously:' Violence, wellbeing, and devaluing people with learning disabilities. *Journal of Interpersonal Violence*. Epud ahead of print 02 February 2021. DOI: 10.1177/0886260521990828. 0886260521990828.

Author biographies

Nathaniel Scherer is a Research Fellow at the International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine. His research is centred on mental health, wellbeing and the quality of life of disabled people and caregivers in the UK and low- and middle-income countries. Address: International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine, Keppel St, London WC1E 7HT, England, United Kingdom.

Email: Nathaniel.Scherer@lshtm.ac.uk.

Recent publication: Scherer N, Banks R, Murko M and Chisholm D (2022) Better Health, Better Lives? 10-Years on From the World Health Organization's Declaration on the Health of Children With Intellectual Disabilities. Journal of Intellectual Disabilities Epud ahead of print 09 March 2022. DOI:10.1177/17446295221076687.

Phillippa Wiseman is a Lecturer in Sociology at the Centre for Disability Research and the Institute of Health and Wellbeing, University of Glasgow. Her research is focused on disability inclusion, human rights and citizenship, including work on people with learning disabilities' experiences of hate crime and violence in Scotland and its impacts on belonging and wellbeing.

Recent publication: Wiseman P and Watson N (2021) 'Because I've Got a Learning Disability, They Don't Take Me Seriously:' Violence, Wellbeing, and Devaluing People With Learning Disabilities. Journal of Interpersonal Violence Epud ahead of print 02 February 2021. DOI: 10.1177/0886260521990828. 0886260521990828.

Nicholas Watson is Professor of Disability Research and Director of the Centre for Disability Research, University of Glasgow. His research has focused on a wide range of disability related topics including disability and social care, disability and aids for daily living, hate crime and disability, and access.

Recent publication: Gormley C and Watson N (2021) Inaccessible justice: exploring the barriers to justice and fairness for disabled people accused of a crime. Howard Journal of Crime and Justice 60(4): 493–510.

Richard Brunner is a Research Associate at the Centre for Disability Research, University of Glasgow. His ongoing research seeks to understand the experiences and inequalities of people with mental distress, disabled people, people with health issues, and other easy-to-ignore groups, using sociological concepts.

Recent publication: Bennett H and Brunner R (2022) Nurturing the buffer zone: conducting collaborative action research in contemporary contexts. Qualitative Research 22(1): 74–92.

Jane Cullingworth is a Research Associate at the University of Strathclyde. Her research is focused on the relationship between the third sector and the state, with a particular interest in the role of intermediary bodies in representing third sector interests in governance networks, as well as the experiences and rights of disabled people in accessing public services.

Recent publication: Cullingworth J, Waston N and Shakespeare T. 'They Have Been a Saving Grace': The Role of the Third Sector in Disabled People's Experiences of COVID-19 and Implications for Third Sector-State Relations. 2021 Voluntary Sector and Volunteering Research Conference, Birmingham, UK, 6–7 Sept 2021.

24

Scherer et al.

Shaffa Hameed is an Assistant Professor at the International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine. Her research includes focus on the health, wellbeing and social inclusion of disabled people, and the sexual and reproductive health and rights of vulnerable groups in the UK and low- and middle-income countries.

Recent publication: Hameed S, Walsham M, Banks LM, Kuper H (2022) Process evaluation of the Disability Allowance programme in the Maldives. International Social Security Review 75(1):79–105.

Charlotte Pearson is a Senior Lecturer in Social and Public Policy in Urban Studies/Institute of Health & Wellbeing at the University of Glasgow. Her research interests focus on disability and social policy, particularly in relation to social care, personalisation and independent living.

Recent publication: Pearson C, Watson N, Brunner R, Cullingworth J, Hameed S, Scherer N and Shakespeare T (2022) Covid-19 and the crisis in social care: exploring the experiences of disabled people in the pandemic. Social Policy and Society. Epub ahead of print 08 April 2022. DOI: 10.1017/S1474746422000112

Tom Shakespeare is Professor of Disability Research and co-Director of the International Centre for Evidence in Disability, London School of Hygiene & Tropical Medicine. His qualitative social research has been with disabled people, in the UK and Africa, exploring social and economic consequences of impairment and illness.

Recent publication: Shakespeare T, Watson N, Brunner R, Cullingworth J, Hameed S, Scherer N, Pearson C and Reichenberger V (2022) Disabled people in Britain and the impact of the COVID-19 pandemic. Social Policy and Administration 56(1): 103–117.

2.3. Key concepts in the analytic commentary

To guide the analytic commentary, I have utilised a social-ecological model of mental health. I did not use a social-ecological framework to guide my research in papers 1-4, but I have applied it in this thesis to structure my synthesis and discussion of the findings.

First proposed by Bronfenbrenner in the 1970s and formalised by McLeroy et al. in the 1980s, the social-ecological approach aims to understand the personal and environmental factors that contribute to development, health and wellbeing, including individual, family, community, systems and wider societal components [51-56]. Social-ecological models and approaches have often been utilised to guide international health. For example, the *Ottawa Charter for Health Promotion* is founded in social-ecological principles [57]. The Charter holds that peace, shelter, education, food, income, stable ecosystems, sustainable resources, social justice and equity are prerequisites for health.

In 2022, the United Nations Children's Fund (UNICEF) launched the *Global Multisectoral Operational Framework for Mental Health and Psychosocial Support of Children, Adolescents and Caregivers Across Settings* [58]. MHPSS is a "composite term used to describe any type of local or outside support that aims to protect or promote psychosocial wellbeing or prevent or treat mental disorders" [58]. The UNICEF framework utilises a social-ecological framework (Figure 2), life-course approach and the Inter-Agency Standing Committee (IASC) MHPSS intervention pyramid to understand mental health conditions in children across the world and to inform support, including prevention, promotion and treatment services.



Figure 2. Social-ecological framework of mental health and psychosocial support

In this social-ecological framework, child wellbeing is conceptualised across four levels. At the centre is the child, where individual characteristics and experiences influence wellbeing, including identity, self-esteem, coping skills and exposure to trauma. Support for children at this level may include psychoeducation, access to nurturing and safe spaces, early child development activities and personalised psychological interventions. Next comes caregivers, primarily parents and family members, as well as teachers, social workers and others influential in a child's development. Risk and protective factors include family functioning, parental support, home environment, household socio-economic status, abuse and neglect. Caregivers' own mental health and wellbeing may also influence the child's wellbeing and mental health. Support for caregivers may include caregiver mental health services, peer-support groups, caregiver-skills training, and fostering collaboration between parents and teachers. The community level reflects the community environment in which a child lives, including community attitudes and culture, as well as the availability of accessible community-led MHPSS activities and multisectoral care systems across primary health care, social welfare and education. Support across this sphere may include anti-stigma campaigns, strengthened social care systems, and participation of children in community improvement and service activities. Influencing all of these factors is an enabling environment that provides the policies, systems and structures to support mental health and wellbeing. Important at this level is financing, appropriate MHPSS policies and legislation, government implementation strategies, workforce development, and research and evidence.

UNICEF have used a social-ecological framework to operationalise their global action for child mental health. In this analytic commentary, I aim to start discussion on how to operationalise the literature for action on MHPSS for children with disabilities. UNICEF include a short segment on disability and inclusion within MHPSS programming, but information on operationalising the framework for children with disabilities is limited. Social-ecological frameworks have been used to understand the experiences of people with disabilities [59, 60], but, to the best of my knowledge, they have not been used to understand the mental health and wellbeing of children with disabilities. In the analytic commentary, I apply the framework to synthesise my findings and discuss the implications of these findings in the context of the wider literature with regards to gaps in evidence and future action for research, policy and practice.

The analytic commentary incorporates critical reflection on papers 1-4, as well as broader literature, including those to which I am first or co-author. I reflect on factors influencing mental health and wellbeing among children with disabilities, including risk and protective factors associated with the four levels of the social-ecological framework. The topics listed at each level of the framework are not exhaustive. They are informed by my research in paper 1-4. There will be factors that influence

94

mental health and wellbeing among children with disabilities in the wider literature that I have not had scope to explore. I begin my discussion at the macro-level of an enabling environment, progressing through each level to individual child experiences. This order is different to UNICEF's presentation of the social-ecological framework. Each level of the framework is affected by the policy environment and national systems and starting at the macro-level helped me to reflect on policy and national systems at each subsequent level.

Often, factors discussed relate to the social determinants of health [61]. These are non-biological factors that influence health outcomes, and are associated often with the conditions in which children grow, including access to care, environment, housing, discrimination, education and household income. In addition to discussion on factors that influence mental health and wellbeing, I examine MHPSS strategies that may support children with disabilities. My discussion includes focus on low-level interventions that can be delivered by non-specialised personnel that are trained and supported by specialist mental health professionals. This process, task-sharing, is often deployed where there are low numbers of mental health professionals to provide support, especially in LMICs and in the field of Global Mental Health [62].

3. Analytic commentary

3.1. Vulnerability and mental health

To start the analytic commentary, I want to reflect on some of the language used in the socialecological framework and my published work. The framework used by UNICEF frequently makes use of the term 'vulnerable' when describing MHPSS for certain children across the world. But is this language always useful for children with disabilities?

In paper 4 I discussed the concept of vulnerability, with reference to the work of Martha Fineman [36, 63] and the social determinants of health [61, 64]. I argued that UK government action (and inaction), including inaccessible health information and removal of social care, *made* people with disabilities vulnerable. However, in this article, I did not consider the appropriateness of vulnerability as a term.

People with disabilities are often classified as vulnerable in legal, political and societal discourse, although the term is not used consistently across countries and context [65-68]. Proponents of this term argue that it stimulates State actors to provide special protection and care for groups that need additional support. This may include social protection or priority medical support. In the UK, this terminology likely played a role in providing people with disabilities priority vaccination. However, there are critics of the term in disability studies [65, 66]. Many consider the term disempowering and one that promotes the medical model of disability. The language places risk on the individual and ignores the wider determinants of health associated with systems and environments. Further, the term vulnerable has become synonymous with fragility and weakness and can promote stigma and discrimination towards people with disabilities. The term can therefore support systems that deprive people with disabilities of their autonomy, agency and decision-making. In a 2011 review of adult social care in the UK, the Law Commission recommended that the UK social care sector move away from the term vulnerable, as it was "stigmatising, dated, negative and disempowering" [69]. The *Care Act 2014* subsequently referred to 'adults at risk' and did not use the term vulnerable.

Despite this criticism, the term vulnerable was used extensively to refer to people with disabilities during the COVID-19 pandemic in the UK. Often the term 'clinically vulnerable' was used. Classification as vulnerable provided priority groceries, medication and access to other needs, and some people with disabilities, including participants in paper 4, described themselves as vulnerable [36]. Yet, research has shown that many people with disabilities in the UK felt embarrassed by the term and believed that it contributed to people with disabilities being perceived differently and as 'other' [70]. Many people with disabilities were also deemed 'not vulnerable enough' to receive priority support and there was widespread confusion over use of the terminology and changing classifications [71, 72].

I argue that the term 'vulnerable' should only be used in the context of children with disabilities and mental health when clearly defined as it relates to social vulnerability, as rooted in the social determinants of health and mental health [61, 73, 74]. In the analytic commentary, I do not use the term vulnerable. I aim to use language that highlights the environmental and structural conditions that exclude and marginalise children with disabilities to negatively impact mental health.

3.2. Social-ecological framework of mental health: enabling environment

In this section, I discuss the macro-level of the social-ecological framework and consider the value of an enabling environment across policy, systems and society.

3.2.1. Government policy and action

Disability inclusion paves the foundation from which to promote the mental health of children with disabilities. My findings from the Gaza Strip and the UK underline the negative impact of exclusion in community, school and government support on mental health and wellbeing. Local and national government policy and strategy is vital to ensure disability inclusion. The UNCRPD has been ratified by 185 UN Member States, but evidence has demonstrated widespread failings in implementation

[75-79]. Many countries have developed new disability policies and taken positive action, but limited knowledge on disability inclusion at government-level, limited resource allocation and gaps in monitoring compliance have led to low levels of implementation, nearly two decades on from the UNCRPD [75-79].

These failings are highlighted in papers 2 and 4 [35, 36]. Families in the Gaza Strip and in the UK did not hold trust in government support for children with disabilities. Organisations of persons with disabilities (OPDs) in the Gaza Strip felt that disability inclusion within government was "talk with no action" [35], and families of children with intellectual disabilities felt abandoned by the UK government during COVID-19 [36]. Limited and insufficient inclusion in government strategy, even in a well-resourced country such as the UK, have resulted in systems that negatively impact on the mental health and wellbeing of children with disabilities [35, 36]. My findings demonstrate systemic failures in governmental support for children with disabilities across diverse settings. They underscore the urgent need for robust policy reforms and enhanced political commitment to address deficiencies and promote the inclusion and wellbeing of children with disabilities, especially during times of crisis, such as the COVID-19 pandemic. Failure to act perpetuates systemic injustices and exacerbates the challenges faced by an excluded and marginalised group.

Failings in implementing the UNCRPD include limited engagement with persons with disabilities, caregivers and OPDs in policy-development and implementation [76, 78-80]. Globally, OPDs have reported dissatisfaction with the extent and quality of their participation [81]. Paper 4 and associated articles are consistent with these findings [36, 46-48]. My research highlighted the limited involvement of persons with disabilities, family members and OPDs in developing COVID-19 strategies, resulting in inappropriate action that effected poorer health, mental health and social outcomes [36, 47]. Where government failed during the pandemic, OPDs stepped in, providing information and support to children (and adults) with disabilities and their families, including mental health support [36, 46]. Limited collaboration with OPDs and people with lived experience is not limited to the pandemic. In 2022, the UK's new National Disability Strategy was ruled unlawful because it had not been developed with appropriate consultation from people with disabilities. This decision has since been overturned, although this has been criticised by disability rights organisations [82, 83]. Well-resourced countries such as the UK should be paving the way for collaborative, participatory practice to promote disability inclusion, but failings remain. Continued avoidance of effective collaboration by government will entrench systems that exclude and marginalise child and adults with disabilities, negatively impacting their mental health and wellbeing, as well as education, employment, mortality and morbidity.

Collaboration and partnership between researchers, government, civil society and people with lived experience is essential for disability inclusion and promoting the mental health and wellbeing of children with disabilities. Although findings in the Gaza Strip indicated limited government action on disability inclusion, partnership between researchers, OPDs, people with lived experience and government was central to our research project [34]. Partnership and shared ownership from the outset increased government engagement and the quality of the guidelines developed. Opportunity to collaboratively discuss research findings with people with lived experience and in the context of practical considerations for government implementation promoted evidence-based, feasible and actionable MHPSS guidelines. This partnership and collaboration likely contributed to government commitment to implementation of the MHPSS guidelines across all schools in the Gaza Strip, although the longer-term sustained implementation and impact has not been assessed.

Collaboration can promote translation of research into policy and practice. Translation of research into policy and practice is often perceived to follow a linear model, in which government are expected to take action from research alone [84]. This does not always work in complex, real-world environments, where stakeholders have different priorities, timelines and conflicting interests. Translating research into practice is a well-recognised challenge [84]. Getting Research into Policy and Practice (GRIPP) is an important topic area, with funders, researchers and policymakers trying to find effective strategies [85, 86]. Dissemination alone may have little impact on policy and practice, although this is often the most common activity by researchers [86]. There are a number of strategies to promote GRIPP beyond the scope of this thesis to explore, but drawing on my experience in the Gaza Strip, and congruous with the literature [85, 86], effective collaboration between stakeholders and participatory approaches can promote the uptake of evidence-based policy and practice. Engagement and partnership early in our process ensured that research questions were relevant to real-world need and increased the likelihood of uptake by national stakeholders. However, time can be a limiting factor. Research funding is short-term, insecure and project-based. Longer-term research funding is needed to promote translation of research to practice [85, 86]. It can allow sustained partnership between stakeholders to promote evidencebased action and allow for researchers to more meaningfully impact policy [85, 86]. In the Gaza Strip, longer-term funding would have enabled in-depth evaluation of the MHPSS guidelines to promote uptake and sustainable practice, drawing on implementation science, such as the Medical Research Council and the National Institute for Health and Care Research guidance on evaluating complex interventions [87]. Longer-term funding would also have provided research capacity to support and monitor implementation, as well as additional opportunities for shared learning between the project partners. Longer-term funding is not guaranteed to result in greater impact or

98

progress on GRIPP, but funders should consider long-term funding to promote evidence for systems building.

3.2.2. MHPSS systems for children with disabilities

Globally, and in LMICs in particular, mental health service provision is low [20] and the treatment gap globally remains substantial [19]. For children with disabilities, the treatment gap is exacerbated by a lack of access to health services [11], including barriers to mental health services specifically [88, 89]. In paper 4, for example, children with intellectual disabilities were unable to access online mental health support during the pandemic because of digital poverty and limited digital literacy [36].

In settings with limited human resources, innovative low-cost solutions have been developed for mental health treatment, such as the WHO Mental Health Gap Action Programme (mhGAP) [90]. However, scale-up of these mental health interventions presents a major challenge [19]. Limited resources (including finance and expertise), limited infrastructure, and stigma towards mental health limit implementation of innovative interventions [19, 91]. Considering disability-inclusive mental health interventions now, whilst developing solutions at scale, presents opportunity. Paper 1 indicates that many interventions for children with disabilities are adapted from interventions developed for the general population [33]. Including reasonable accommodations and guidance for disability inclusion in MHPSS means that children with disabilities will not be left behind in efforts to promote mental health and wellbeing. This process is beginning, as demonstrated in the newest mhGAP guidelines, released in 2023, which include information on psychosocial support for children with disabilities [90]. Specifically, the guidelines contain the section titled, "What is the effectiveness of psychosocial interventions (apart from caregiver skills training) to improve development, wellbeing, functioning and school participation in children and adolescents with neurodevelopmental delays and disabilities?". The section provides an overview of evidence and suggested actions to promote wellbeing and address mental health conditions among children with disabilities. This is not limited to addressing mental health conditions but also promoting social skills, communication and academic performance. Gaps in research are also presented. It is encouraging to see a mainstream tool from a multilateral organisation such as WHO meaningfully include children with disabilities. The guidelines could be further strengthened with more in-depth guidance on recommendations for reasonable accommodation, as a lay worker with limited knowledge on disability may currently struggle to interpret and implement the guidance. Interestingly, much of the research presented is focused on neurodevelopmental disabilities, such as autism and attention deficit hyperactivity disorder (ADHD). Children with intellectual disabilities are only referenced with

relation to interventions that improve communication. No information is presented on deaf and hard of hearing children or children with a visual impairment. This is not a criticism of the tool; these omissions reflect gaps in the literature and evidence base that should be addressed. Findings from paper 1 reflect this limited literature, specifically for deaf and hard of hearing children [33].

Further, paper 1 highlights that evaluation research on MHPSS for deaf and hard of hearing children is often small-scale and quasi-experimental [33]. Larger-scale randomised controlled trials (RCTs) are needed to inform evidence-based policy and practice. These should be accompanied by process evaluations to understand facilitators and barriers to implementation. Understanding and addressing barriers to implementation should be underpinned by collaboration and partnership between researchers, policymakers and OPDs, to facilitate contextually appropriate solutions to implementation challenges for iterative adaptation and evaluation. Formative research, as I conducted in the Gaza Strip [34], is also needed to inform evidence-based development of MHPSS. Without formative and evaluation research, interventions may remain siloed, small-scale projects that fail to reach desired impact at scale. This is an issue consistent in health intervention research and delivery for people with disabilities [92]. This is also a limitation of my research in the Gaza Strip. The pilot studies in paper 3 were small-scale and a larger RCT is needed.

Disability-inclusive MHPSS interventions and systems can only exist in a disability-inclusive health system. People with disabilities experience health inequity and challenges to healthcare access [4, 12]. Health systems worldwide are failing to provide disability-inclusive services. Global efforts and multisectoral action are required to improve disability inclusion. In the *Global report on health equity* for persons with disabilities, WHO outline needed action in political commitment, policy, health financing, physical infrastructure and human resources [4]. The Missing Billion Initiative also provide a framework for disability-inclusive health systems [12]. Action to develop disability-inclusive MHPSS will need to be situated and contextualised within action for disability-inclusive health systems. Disability-inclusive MHPSS is intrinsically tied to a broader disability-inclusive health system. Increased efforts are needed on both, but disability-inclusive MHPSS may not be possible until core health systems are inclusive. To promote disability-inclusive MHPSS systems, it may be beneficial to adapt frameworks such as that of the Missing Billion Initiative to the specifics of MHPSS for adults and children with disabilities. The framework outlines action needed at system level and service delivery and has previously been adapted to the needs of people with intellectual and developmental disabilities. A similar process could adapt the framework for disability-inclusive MHPSS systems.

In this section, I have primarily discussed intervention services, but MHPSS systems for children with disabilities should not focus solely on treatment. Too narrow a focus on treatment has been a criticism of scale-up in mental health support and the field of Global Mental Health specifically [93]. Intervention services and manualised tools, such as mhGap, are needed to identify and treat mental health conditions, but MHPSS systems that support the mental health and wellbeing of children with disabilities will need to extend to environments that promote inclusion, quality of life and wellbeing. This principle forms a central pillar of my discussion throughout this thesis. Disability inclusion and action to promote mental health and wellbeing are needed to support children with disabilities.

3.2.3. Crisis and complex settings

Given the contexts in which my research was conducted, it is important to consider the impact of crisis on the mental health and wellbeing of children with disabilities. Children and adolescents in times of emergency and crisis are at increased risk of mental health conditions [94-96]. In paper 4, children with disabilities experienced isolation and poor mental health during COVID-19, created in large by the removal of support and exclusion from government policies [36]. These findings are consistent with the broader literature [36, 97-99]. These failings reflect widespread failings across disability-inclusive disaster management, disability-inclusive emergency preparedness and disaster risk reduction, where limited awareness on disability inclusion, lack of prioritisation of disability inclusion and lack of partnership with persons with disabilities has resulted in exclusion of people with disabilities in emergency response [100]. These are important findings in the context of future crises. Based on learnings from COVID-19, the PREparedness, REsponse and SySTemic transformation (PRE-RE-SyST) model for disability-inclusive pandemic responses may provide strategic direction for future emergencies [101]. The model provides recommended action to address health and social disparities experienced by people with disabilities in a pandemic, including actions that address key challenges highlighted in paper 4, including the need to maintain essential services during lockdowns, provide accessible health information and engage with disability advocates. Specific note is given in the model to psychological support for children with disabilities. As well as actions in response to a pandemic emergency, the model supports the argument that governments need to address the root, structural and systemic causes of inequities experienced by children with disabilities. Evidence from paper 4 and the wider literature emphasise the importance of responding to a crisis from a basis of strength; i.e. a system that is inclusive and supportive of children with disabilities [36]. Crises expose existing frailties and these are exacerbated by an inadequate government response. In the UK, government response largely forgot children with disabilities and their families [36]. Taking lessons from the pandemic, the UK government and governments globally, in collaboration with people with lived experience, researchers and civil society, need to address first the systems that exclude children with disabilities, in conjunction with crisis response strategies that are disability-inclusive. Without action, children with disabilities will again experience poor mental health during future crises.

Although my research in the Gaza Strip did not focus on the impact of conflict and humanitarian settings on mental health, the current situation in the Gaza Strip necessitates attention. Children with disabilities are particularly at risk to the impact of conflict and displacement, including greater risk of psychological impact [102, 103]. MHPSS is founded in humanitarian crises and the IASC have recently developed an information note on disability and inclusion in MHPSS in humanitarian settings, including provisions for children with disabilities [104]. These guidelines are long overdue, with the original IASC MHPSS guidance released in 2007. They provide a comprehensive guide to disability inclusion in MHPSS and were developed using participatory methods. The IASC is the global leader in MHPSS in humanitarian settings and this hopefully spells optimism for disability-inclusive MHPSS in humanitarian crises moving forwards.

3.3. Social-ecological framework of mental health: community

Community-level factors can include community MHPSS strategies, but emerging strongly from my research was the role of stigma in limiting community inclusion and participation, and the negative impact of this on wellbeing. In this section I discuss findings related to stigma experienced by children with disabilities.

3.3.1. Stigma, discrimination and social exclusion

Children with disabilities are more likely to experience loneliness and isolation than peers without a disability [26]. Deaf and hard of hearing children in the Gaza Strip experienced exclusion at school and in the community [35]. Social exclusion is a major influence on poor mental health [105], and as with deaf and hard of hearing children in the Gaza Strip, children with invisible disabilities can often be at greater risk of peer rejection [106]. Social exclusion and limited opportunity for social interaction can inhibit the development of social skills, further entrenching social exclusion when opportunities do arise.

Stigma is a major driver of social exclusion [107]. Deaf and hard of hearing children and caregivers experienced stigma and discrimination in the Gaza Strip [35], consistent with previous literature [108]. In the Gaza Strip, stigma and discrimination was driven by limited knowledge on deafness and disability, where knowledge was often centred on medical models of disability, rather than social and rights-based models. This is consistent with the wider literature. Stigma and discrimination towards disability often arises from misconceptions about the cause and nature of disability, as well as limited expectations on the capability of people with disabilities [107, 108]. Stigma is typically greater towards those with more severe impairment, as they are seen to be more 'other' and less able to contribute to their community [108]. As a result of community stigma, parents and children in the Gaza Strip demonstrated internalised stigma and some families denied their child's disability [35]. As a result, many deaf and hard of hearing children were excluded from support they needed (e.g. sign language) by their families. These findings suggest a pervasive pattern wherein public stigma and internalised stigma impedes access to essential support services for children with disabilities.

For children, school offers opportunity for inclusion [106]. However, it is also an environment in which children with disabilities are at increased risk of discrimination and bullying. In the Gaza Strip, this dichotomy in school experiences was shown. Some deaf and hard of hearing children in mainstream schools thrived, whereas others experienced bullying and exclusion from peers. Addressing stigma and improving knowledge on disability and deafness was a key component of the MHPSS guidelines in the Gaza Strip, consistent with evidence highlighting the promise of school-based interventions to reduce disability-stigma in LMICs [109]. However, more high-quality research is needed on school-based interventions, including rigorous evaluation. In the Gaza Strip, teachers noted that limited time and capacity to implement the MHPSS guidelines presented a barrier. Impact evaluations of school-based interventions should always be complimented by process evaluations to understand such barriers to long-term sustainability.

In the context of MHPSS, it is important to note that mental health in children is also a stigmatised condition [110]. Children often do not seek mental health care as a result of stigma, prolonging the negative impact on physical and mental health [111]. Mental health stigma reduction interventions, largely conducted in schools, have shown long-term improvement in child and adolescent mental health literacy and reduction in stigma towards mental health [112]. In the context of MHPSS for children with disabilities, anti-stigma campaigns should thus tackle two issues; disability stigma and mental health stigma. Incorporating disability inclusion and disability stigma within anti-stigma campaigns on mental health for the general population may be beneficial. Such actions are central to mainstreaming disability inclusion and this deserves further attention.

3.4. Social-ecological framework of mental health: caregivers

Caregivers are those who are influential in a child's development. This includes family members, as well as primary caregivers at school (e.g. teachers) and child services (e.g. social workers). In this section, I focus on parents and family members. Caregivers need support to provide children with

disabilities with a safe and nurturing environment. This section explores the mental health of family caregivers and caregiver knowledge on disability.

3.4.1. Mental health and wellbeing among caregivers of children with disabilities

Parents and family members of children with disabilities are at increased risk of stress and mental health conditions [113, 114]. This is true in 'normal' times and was exacerbated during the COVID-19 pandemic. Stressors during the pandemic increased the risk of poor mental health, as presented in paper 4 [36]. Findings of paper 4 are consistent with a recent meta-analysis which found 37% of family caregivers of children with neurodevelopmental disorders experienced anxiety during COVID-19, 41% experienced depression and 59% stress [115]. Paper 4 highlighted how worry, isolation and increased caregiver demands contributed to this increased risk.

It is important to understand and address the mental health of caregivers in order to support their quality of life. It is also important to consider the impact of poor caregiver mental health on children with disabilities. Caregiver mental health is associated with child development and mental health, including cognitive development and behavioural disorders [116, 117]. It may also influence secure parent-child attachment and nurturing care, vital to a child's wellbeing [118]. Caregiver mental health also influences access to mental health services by children with disabilities [27]. Thus, understanding and addressing the mechanisms of caregiver stress and mental health may contribute to improved wellbeing, development and behavioural outcomes among children with disabilities.

Considering specifically the mental health of caregivers of children with disabilities during COVID-19, we can reflect on the findings of paper 4, as well as other studies published from the same project, to which I am co-author [36, 46, 48]. During the pandemic, children with disabilities experienced negative impacts on their mental and physical wellbeing, including increased behaviours that challenge, isolation, depression and lost developmental skills [36, 48]. This placed strain on caregivers, increasing worry and caregiver demands. This strain was exacerbated by the withdrawal of social support, reducing the resources available to caregivers and increasing demands upon them. My findings in the UK were consistent with other countries [119, 120]. These findings are not necessarily unique to COVID-19 and are congruent with findings for caregivers of children with disabilities in normal times [113, 121]. However, findings from the pandemic highlight that crisis and the distress response is not static and changes as time progresses. Coping strategies and support provided prior to the pandemic (e.g. family, friends, formal care) may not have been available during and in the period after the pandemic. Caregivers may have had to develop new coping strategies based on these new conditions in order to adapt to the stressors of the pandemic. As society continues to recover from the pandemic, it may be appropriate to investigate these coping

104

strategies outside of the immediate crisis. For example, during the pandemic, peer-support provided a source of support when formal care failed [122]. The value of peer-support is not a new concept, but evaluation research is limited for caregivers of children with disabilities [123]. Evaluating this and other coping strategies that expanded or newly developed during COVID-19 may unearth novel coping strategies and systems that can be adopted in the post-pandemic era. This requires further attention as countries look to build back better.

In addition, it would be valuable to investigate caregiver mental health and coping strategies in humanitarian settings. My research in the Gaza Strip focused on the wellbeing of children and did not explicitly explore caregiver mental health. Further research is needed to explore the mental health of caregivers in this and other humanitarian settings. For example, paper 2 and 3 highlight the limited formal support available for deaf and hard of hearing children in the Gaza Strip, which negatively impacted their mental health wellbeing [34, 35]. Limited formal support may also impact the mental health and wellbeing of caregivers in this setting. Research is needed to explore the unique factors that influence caregiver wellbeing in similar humanitarian settings.

3.4.2. Caregiver confidence, knowledge and skills

Family knowledge on disability influences the wellbeing of children with disabilities. In the Gaza Strip, many families had little information on disability and how to support their child [34, 35]. The vast majority of deaf and hard of hearing children are born into hearing families who often have limited knowledge on deafness and Deaf culture [124]. Although many caregivers in the Gaza Strip still supported their child, limited information contributed to low acceptance and internalised stigma among some. Interventions for caregivers of children with disabilities, often skills-based training programmes, can improve parental knowledge and skills, resulting in positive child outcomes, including development, educational outcomes and behavioural outcomes [125]. These interventions also improve caregiver self-efficacy [126]. Caregiver self-efficacy is associated with more positive caregiver-child relationships, more effective parenting styles and improved mental health among caregivers of children with disabilities [128]. Caregiver training programmes also provide peer-support, providing a platform for caregivers to share experiences and foster social relationships [122]. The majority of impact evaluations of caregiver programmes have small sample sizes and larger-scale evaluations are needed [125].

Although the MHPSS guidelines I developed in the Gaza Strip were designed for teachers, family caregivers in the steering committee and as study participants requested information for families. Thus, in the guidelines teachers are encouraged to work with families to jointly improve knowledge,

promoting consistent support at home and school. The guidelines were developed in collaboration with family caregivers and future research should adapt the guidelines for family caregivers in the Gaza Strip. They receive limited information on disability or mental health and guidance would be valuable. In addition to adapting the guidelines, adapting existing caregiver training programmes for families in the Gaza Strip may be beneficial. Evaluation of such programmes in this context is needed.

3.5. Social-ecological framework of mental health: children with disabilities

At the centre of the social-ecological model is the child. My research and wider literature identify a number of risk and protective factors at the level of the child that will influence the mental health and wellbeing of children with disabilities.

3.5.1. Access to education

Education is a fundamental human right. Access to education is important for child development and long-term outcomes, including health, mental health, wellbeing and future socio-economic status [19, 129, 130]. Schools are an important setting for promoting mental health, and inversely, a setting that can present stress and negatively impact mental health. The UNICEF MHPSS framework recognises the need for safe and supportive education environments to promote child mental health [58]. However, children with disabilities have poorer access to education. In a study of 15 LMICs, disability reduced the probability of being in school by a median of 30.9% [131]. Neither individual, socioeconomic or household characteristics explained this inequity. Inaccessible school environments likely played a substantial role in inequitable access. Barriers in school environments include the physical architecture (e.g. no ramps) and inaccessible learning materials (e.g. no braille) [132-134].

This is consistent with my findings in the Gaza Strip [35]. Deaf and hard of hearing children in mainstream schools experienced inaccessible learning environments and discrimination at school. This included a lack of accessible language. Inaccessible environments negatively impacted on the quality of their education and their wellbeing. In 2019, only 14.6% of children with disabilities in the Gaza Strip attended school [42]. In my research, I did not interview deaf and hard of hearing children that were not attending school. All participants attended either a mainstream or special school. Research is needed to understand the barriers that prevent children with disabilities in the Gaza Strip, and other settings, from attending school. It will also be valuable to compare the mental health of children with disabilities attending school versus children not attending. This research is not currently possible in the Gaza Strip given the current context.

106

Inclusive education has been a major global effort in recent decades, particularly after The Salamanca Statement and Framework for Action on Special Needs Education, published in 1994 [135]. Evidence has demonstrated the positive impact of inclusive education on academic and social skill outcomes [136], however, most mainstream schools are not sufficiently inclusive of children with disabilities [137, 138], as my research demonstrates in the Gaza Strip. Thus, inclusive education is debated within disability studies. An in-depth exploration of this complex topic is beyond the scope of this thesis. The research in the Gaza Strip highlights some of the key complexities. Children with disabilities are not a homogenous group, and as evident in the Gaza Strip, children and parents will have different preferences and different experiences within available education systems [35]. Some children with disabilities thrive in a mainstream school when the environment is supportive and accessible. Others prefer a specialist school, where they are surrounded by peers of a similar experience in an accessible setting, with specialised resources and trained teachers. In the Gaza Strip, children with disabilities attending special schools appeared to have better wellbeing, because inclusion and support was superior compared to the majority of mainstream environments [35]. Calls for inclusive-education only may place some children at risk of psychological distress, if mainstream school environments are not fully accessible and inclusive. Ideally, children with disabilities and their families would have a choice. Either a fully accessible mainstream school that provides no difficulties with access, or a well-equipped specialist school.

It is important to consider the role of teacher knowledge and confidence in education for children with disabilities. Evidence shows that limited teacher training is a major barrier to implementation of inclusive education [139, 140]. Appropriate quality training is important to improve inclusive education practices, knowledge on disability and teacher attitudes [139]. In the Gaza Strip, teachers supported inclusive education. The MHPSS guidelines improved their knowledge of disability, improving disability inclusion in their classrooms. However, teachers reported concerns about their capacity to implement new measures for disability inclusion. Teachers need support and resources to support children with disabilities and encourage schools as a setting to promote wellbeing. Support for teachers is underpinned by assistance from school management [141]. Assistance from school management is underpinned by support from government, disability-inclusive education policies and appropriate funding. As with previous discussion on policy, inclusive-education and alleviating barriers to education for children with disabilities requires a collaborative, participatory approach [138].

3.5.2. Identity, self-esteem and wellbeing

Investigating social identity and self-esteem may provide further insight into the risk and protective factors that influence the mental health and wellbeing of children with disabilities. Self-esteem is vital for a child's mental health. Children with higher levels of self-esteem have higher levels of wellbeing, lower instances of mental health conditions and better social relationships [142-145]. Social identity theory can help us understand self-esteem, linking our understanding and value of self to our social identity [146, 147]. Social identity is often developed through group and cultural membership, in which supportive, stable groups nurture social identity and build self-esteem, promoting mental health and wellbeing [148]. In developing social identity, an individual may selfcategorise to a group, based on their knowledge of that group, how they connect to others in the group and how they ultimately associate with the group identity [146]. Developing social identity is not a static process, and we grow and respond to different cultural contexts and variables in our lived environment. Our social identity is thus influenced by our own internal perceptions, as well as external pressures and stimuli [146]. Social identity and a sense of belonging offers protection to life's stressors and the onset of psychological distress [148]. Social identity and group membership can provide social support and a coping resource in times of difficulty. Inversely, those without a strong sense of social identity can experience isolation and loneliness, increasing the risk of poor mental health [148]. In the following paragraphs, I explore the association between self-esteem, social identity and mental health among deaf and hard of hearing children, drawing on findings from paper 2 and 3. This discussion links topics previously discussed, such as education and stigma.

Amongst deaf and hard of hearing people, a component of social identity revolves around deaf identity. In 1993, Glickman developed the *Deaf Identity Development Model* of four deaf identities and the psychological process by which these identities are acquired [149, 150]. The first, "hearing identity", applies to deaf and hard of hearing people that strongly identify with the majority hearing culture who maintain or adopt hearing norms and behaviours in their daily life, including oral communication. People with a hearing identity may consider deafness and hearing loss as an impairment only arising from medical pathology. Next, "marginal identity" refers to deaf and hard of hearing or Deaf cultures and norms. "Immersion" or "Deaf identity", describes those that align with and identify with Deaf culture and the wider Deaf community, including language, history and other cultural attributes. People with a Deaf identity may view hearing values as audist, in that the values of the hearing world are regarded by the majority of the world's population as more important. Hearing culture may be viewed negatively by people Deaf identity. People with a "bicultural identity" will freely associate with both hearing and
Deaf communities and values. Often people with a bicultural identity are comfortable communicating in both oral and sign language.

Evidence has shown that people with hearing, Deaf and bicultural identity have better mental health and wellbeing than those with a marginal identity [151-153]. The majority of evidence on this topic is from high-income settings. My research in the Gaza Strip allows exploration of this phenomenon in a LMIC [35]. I did not explicitly assess participants' deaf identity, but have assumed probable identity, based on their communication preferences and experiences in the community and school. Children in special school environments commonly used sign language, engaged with deaf and hard of hearing peers and received specialist support. Their caregivers were well-informed on the support available, on hearing loss, on deafness and on deaf and hard of hearing rights. These children would likely identify with a Deaf or bicultural identity. They were reported at lower risk of mental health issues. Children in mainstream environments were most likely to feel included if they used oral communication. These children had hearing friends and generally positive experiences in the community. They tended to have milder hearing loss and/or used a hearing aid or cochlear implant. They are likely to associate with a bicultural or hearing identity. They did not often report experiences of psychological distress, related to fewer challenges in daily life and feeling part of the wider community. Consistent with previous literature, interviews with participants in the Gaza Strip indicated that children most at risk of poor mental health were those experiencing language deprivation and social exclusion, especially those who were in a mainstream schooling environment with limited oral communication skills. These children likely experienced marginal identity. This would be consistent with the literature from high-income countries and requires further attention in the Gaza Strip and other LMICs. Marginal identity may represent a transitional stage in forming social identity, especially in younger children, and examining identity and wellbeing over the longterm may be beneficial.

Furthering discussion in section 3.3.1., from a social identity theory the association of stigma and discrimination with poorer health and wellbeing may represent the negative internalised opinions of people associated with stigmatised identities, from which they may subsequently develop lower self-esteem and poor psychological wellbeing. However, members of stigmatised populations, when connecting their identity to a group, can experience psychological support from others in the stigmatised group [154]. Thus, deaf and hard of hearing children who identify with the Deaf community, and importantly interact with and receive support from this community, may benefit from the psychological resources provided by group membership and protection from the negative impact of discrimination, if experienced. This is consistent with findings in the Gaza Strip, where deaf and hard of hearing children experienced stigma and discrimination [34, 35]. Those in special

schools, who likely had Deaf or bicultural identity, engaged with other deaf and hard of hearing children and received support within a deaf and hard of hearing space, offering protection from stigma and discrimination, contributing to improved wellbeing. Many children with a marginal identity in mainstream schools had little interaction with other deaf and hard of hearing children and would have limited support from group membership when experiencing discrimination and bullying from hearing peers. Thus, they would be at risk of poor mental health, as reported. Strengthening the presence and visibility of disability organisations and groups within local communities to provide children with disabilities opportunity to engage with disabled peers and group membership. Strengthening disability organisations also has the potential to improve their capacity to collaborate in research, policy and programmes.

Among deaf and hard of hearing people, language and communication are associated with selfesteem, social identity and mental health [155]. However, many deaf and hard of hearing children experience language deprivation, resulting from a lack of access to oral or signed languages. This may arise, for instance, when a child is encouraged to focus on spoken language, at the expense of natural sign language, even if their hearing loss prevents access to oral communication [156]. Language deprivation can often occur when a family's language environment does not correspond to the child's natural form of communication. Language deprivation is associated with developmental delay and poorer mental health outcomes among deaf and hard of hearing populations [156]. Communication further influences a child's interaction with services, community and friends, and thus, language deprivation can lead to social isolation and poorer mental health. Communication with family is of particular importance for deaf and hard of hearing children. Deaf and hard of hearing children who cannot communicate with their family are more likely to experience a mental health condition [157]. Deaf and hard of hearing adults who use sign language are more likely to hold a Deaf identity and have better wellbeing [152, 158]. Communication and sign language are central to a positive identity and psychological wellbeing [159]. In the Gaza Strip, language deprivation resulted in isolation of deaf and hard of hearing children at school and in the community, negatively impacting wellbeing [35]. Deaf and hard of hearing assumed to have a marginal identity often had moderate or severe hearing loss, yet many families did not support them to learn sign language, despite their limited capacity for oral communication, associated with caregiver knowledge previously discussed. Other families did not have sign language services available in their local community, speaking further to the need for a multi-sectoral response to promote the mental health and wellbeing of children with disabilities. Within the context of a MHPSS system for children with disabilities, language and appreciation of a child's natural language is vital. This includes accessible communication for children with differing needs, whether that is sign

language, braille or simplified language. This supports community participation (with both disabled and non-disabled peers), social identity and self-esteem. For deaf and hard of hearing children, more value needs to be placed of sign language internationally. Only 41 countries recognise sign language as an official language; 26 of these are in Europe [160]. What message does this send to children with disabilities and their families, especially in the context of identity? Without international and national recognition, policy and legislature will not promote sign language use among deaf and hard of hearing children, nor sign language provision in schools and healthcare. Again, this brings us back to disability inclusion. It is an example of continued exclusionary cultural and political practices that marginalise and alienate, contributing to poor mental health and wellbeing.

The literature on deaf identity and wellbeing is growing but is currently limited in low- and middleincome settings. Here, I present a start in understanding the association between deaf identity and wellbeing in these settings, and the broader topic of disability identity, self-esteem and wellbeing for children with disabilities. Developing identity and self-esteem is associated with all facets of the social-ecological model and should be a primary focus of MHPSS. In this, it is important to value the role that families and caregivers play in developing a child's identity. As discussed in section 3.4.2. families need training and support to help children with disabilities develop a strong sense of self in support of their mental health and wellbeing.

3.5.3. Resilience and self-management

Coping style is a major component of the social-ecological framework, including resilience and selfmanagement. Resilience is associated with protection from mental health conditions [161, 162]. Resilience training has been found suitable and effective for children in various settings, including children with disabilities [163, 164]. Paper 1 and the wider literature indicate that resilience training may improve mental health outcomes and self-esteem among children with disabilities [33, 165]. This deserves attention in the context of MHPSS intervention. Evaluation research is needed on resilience training for children with disabilities, as the available evidence is limited by small sample sizes and limited geographic scope.

It may also be appropriate to consider self-management and self-care practices of mental health, especially for settings with limited MHPSS, whether that be in a low-resource setting or when services are removed during a crisis, as seen in the COVID-19 pandemic. Self-management and self-care techniques have been used with youth to manage emotional challenges [166, 167]. While the quality of evidence is relatively poor, self-management techniques have been used by children with disabilities to manage their disability and health [168, 169]. Future research could adapt effective

self-care and self-management approaches to children with disabilities. Adapting self-management techniques for times of crisis, such as the pandemic, also merits attention.

3.6. Social-ecological framework of mental health: summary

Disability inclusion is needed to reduce discrimination, promote education, social inclusion, selfesteem and identity among children with disabilities. Factors that influence the mental health and wellbeing of children with disabilities are thus grounded in disability-inclusive systems and societies. Increased efforts are needed to implement the UNCRPD globally. As well as disability inclusion in global, national and local communities, children with disabilities need access to disability-inclusive MHPSS services and interventions. Disability-inclusive MHPSS services and interventions are not simple to implement, particularly in resource-constrained settings, and large-scale evaluation research is needed to inform policy and practice at scale.

In this thesis, I have used a social-ecological framework of mental health to examine factors that influence the mental health and wellbeing of children with disabilities, as well as suitable MHPSS strategies. The framework provides a clear, simple model with which to conceptualise MHPSS for children with disabilities. Exploration across different levels (environment, community, caregiver, child) provides salient direction and structure. The framework I used in this thesis was derived from the UNICEF Global Multisectoral Operational Framework for Mental Health and Psychosocial Support of Children, Adolescents and Caregivers Across Settings [58]. The UNICEF guidance includes a short section on disability and inclusion, and guidance on operationalising the framework includes reference to disability inclusion. However, reference is rare and recommendations are often generic and ill-defined. Taking lessons from this thesis and proposed future research, the UNICEF framework and guidance would be strengthened by clear, replicable actions for including children with disabilities in the actions currently suggested, including reasonable accommodations and other strategies to increase accessibility. Strengthening disability inclusion in the guidance would better mainstream disability as a priority in efforts to support the mental health and wellbeing of children worldwide. This would encourage mainstreaming of disability into actions for all children within the MHPSS social-ecological framework, including safe and hygienic housing, nutrition and protection from harm. In addition, a supplementary, standalone framework of MHPSS for children with disabilities may be suitable to provide finer information where necessary, including actions specific to children with disabilities only. A standalone framework will help strengthen direct, targeted action for children with disabilities, but it should not be used in lieu of their inclusion within mainstream activities.

My analytic commentary has explored factors that can be applied to the MHPSS framework, but I have been limited by the scope of my included research and this thesis. My research was conducted in two contexts only and focused on deaf and hard of hearing children and children with intellectual disabilities. Future work should build on this analytic commentary to explore components of the social-ecological framework of mental health and wellbeing among children with other disability types and in different contexts, whether that is in an LMIC or during a global pandemic. Figure 3 on the next page begins this process, mapping the factors discussed in the analytic commentary to the social-ecological framework. There are numerous other topics for future research to explore and add to this social-ecological framework, which I have not had scope to discuss within the context of my research and this thesis.



Figure 3. Adapted social-ecological framework of mental health and psychosocial support for children with disabilities

Although the social-ecological framework has been a useful model to structure my analytic commentary, there are considerations to take forward when aiming to adapt and operationalise the framework to MHPSS for children with disabilities. Throughout the analytic commentary, it is evident that components of the social-ecological framework interact. For example, access to education influences the mental health and wellbeing of children with disabilities, but this access is affected by policy, school resources, environmental barriers, community attitudes, caregiver knowledge and teacher training. Such interaction indicates need for holistic, multi-sectoral action. Social-ecological models are designed to understand the interacting and reinforcing nature of components at different levels of the framework [53], and although the four levels of a MHPSS framework for children with disabilities are useful to consider independently, understanding the interactions of key components will enable strategic policy decisions that make best use of resources. This said, interaction between components is not immediately obvious in the MHPSS framework presented by UNICEF (including the accompanying guidance) and the framework presented for children with disabilities in Figure 3. Guidance on a framework of MHPSS for children with disabilities will be strengthened by mapping the interactions across components, as well as the sectors and diverse actors that are involved. Such mapping will also help identify opportunities for collaboration and partnership. This will be a challenging exercise to complete for a global framework, given the different health and education systems, institutions and actors that would need to be represented. It may be sensible to provide guidance within a global framework on how to contextualise to local systems. Guidance on mapping key stakeholders, sectors and interactions across components would be useful for local leadership to develop a nationally contextualised framework. Contextually adapting the framework to the culture, resources and systems will require collaboration between government, researchers and disability stakeholders (such as OPDs). Contextualising the framework nationally will likely require additional research into mental health and wellbeing of children with disabilities in the country, given the paucity of evidence on this topic.

3.7. Recommendations for policy and research

To finalise discussion on the mental health and wellbeing of children with disabilities, I summarise below brief recommendations for global, national and local stakeholders.

3.7.1. Policy

 Disability inclusion is central to promoting the mental health and wellbeing of children with disabilities. Implementation of the UNCRPD needs to improve worldwide. Children and families cannot be forgotten.

- Governments must value and meaningfully collaborate with persons with disabilities, OPDs and civil society to promote disability inclusion.
- Mental health policy, including in the field of Global Mental Health, must contain clear, actionable recommendations to support the mental health and wellbeing of children with disabilities.
- Disaster and emergency response policies need to include strategies to support children with disabilities and their families at times of crisis.
- Language in policy is important and should respect human-rights.

3.7.2. Research

- Evaluation research is needed to promote evidence-based MHPSS for children with disabilities. Large-scale, high-quality RCTs are needed, with associated process evaluation.
- Specifically, the MHPSS guidelines developed in the Gaza Strip need longer-term evaluation. Future research could also explore adapting the guidelines to other contexts.
- There is a need for more long-term funding to promote translation of research into policy and practice.
- Continued research on the mental health and wellbeing of children with disabilities is needed in diverse contexts, including LMICs and humanitarian settings.

4. Discussion and personal reflections

In this section, I reflect on the research practices in papers 1-4 and my experience as an early career researcher. To end, I consider my positionality within the field of global disability and mental health research.

4.1. Discussion on my research

My research in this thesis was undoubtedly influenced by COVID-19. In the Gaza Strip, the process of developing the MHPSS guidelines was conducted remotely because of international travel restrictions. Regarding the qualitative research, all training and supervision were conducted via WhatsApp and Zoom. The local researcher had not previously conducted qualitative research, and the level of support I was able to provide was lower than I could have provided in-person and lower than I would have liked. This was compounded by a condensed timeline, as we looked to complete interviews in-person after local lockdown restrictions had lifted. We did not know when lockdown restrictions would be re-introduced and we thus worked to a tighter time than would be ideal. This condensed timeline limited our pilot phase, including discussion on the questions and interview

techniques. The condensed timeline also limited transcription during data collection. This restricted our capacity to explore emerging codes during data collection and to iteratively reflect on and adapt the interview guides, although we held daily debriefs to mitigate this limitation. Additionally, my attending the interviews remotely rather than in-person limited my understanding of interview quality and limited advice I could provide, especially without ongoing transcription. My being present remotely during interviews may also have impacted participant comfort and response. Although I introduced myself and remained visible on screen, they may have been suspicious that others were listening in on the interview. In the UK, I conducted interviews remotely but this excluded participants for whom remote interviews were inaccessible and impacted interviewerinterviewee dynamics, potentially influencing depth of data and data quality. These challenges were not unique to my experience. Projects in health research worldwide experienced similar challenges and good practice has emerged, presenting opportunities for the future [170]. Improved videoconferencing technologies and artificial intelligence may help conduct qualitative research in diverse settings at lower cost and within shorter timeframes. I agree, but I believe that we have a way to go before online forums should replace in-person research. We need to understand how to best use these technologies without compromising relationships and data quality. Online technologies are often inaccessible for people with disabilities and may not work well in contexts with lower internet capacity and different languages. Collaboration between international and local researchers is also more difficult via remote technologies. Online qualitative research methodologies will continue improving as we gain more experience, but stakeholders, and especially funders, should not now consider in-person research support obsolete.

With regards to paper 1, there are two limitations to consider. First, there were limitations in the search strategy. I used only three databases and did not search Web of Science and LMIC databases, such as Africa-Wide Information. To overcome time-constraints, I used findings of a study of database combinations to inform a strategy that would maximise results in the least time [171]. However, excluding databases may have resulted in under-identification of relevant studies. In systematic reviews that I've conducted since, I've included a broader range of databases and will continue to do so in order to capture all possible results. Second, I consider now whether it would have been beneficial to expand the search in this scoping review to all children with disabilities. The review was undertaken specifically within a research project to develop MHPSS guidelines for deaf and hard of hearing children, and this narrower focus has its own benefits in terms of presentation and interpretation, but given the efforts expended on the literature search, expanding to children with disabilities may have been a valuable use of resources. Findings from this broader criterion could have uncovered additional elements for disability-inclusive MHPSS strategies to improve

content of the MHPSS guidelines, contribute to discussion in this thesis and provide a useful addition to the wider literature. The search strategy could now be adapted to encompass other disability types and may be a useful step in developing a disability-inclusive social-ecological MHPSS framework for children with disabilities.

Considering paper 4, a major limitation in the context of this thesis is the limited participation of children with disabilities. Although caregivers were included, along with the reflections of disability organisations supporting children, the exclusion of children themselves limits our understanding. Children with disabilities are often excluded from qualitative health research [172] and we perpetuated this exclusion. Given the lockdown measures in the UK at the time of our research, we conducted remote interviews. These would have been inaccessible to some children with disabilities and there will have remained a population we could not reach. However, interviewing children with disabilities would have enabled better understanding of their personal experience. This was a strength of my work in the Gaza Strip and would have benefitted our research in the UK.

4.2. Reflexivity

In the context of this thesis, it is important to reflect on my position in the field of global disability and mental health research. I am not a person with a disability nor am I from an LMIC. Interestingly, this thesis presents findings from my home country and an LMIC in which I had no experience or prior knowledge. My experiences then are both as an 'insider' and an 'outsider'. Both positions have their strengths and limitations. In the Gaza Strip, I held no prior contextual knowledge. This may have limited my analysis and interpretation. On the other hand, it may have promoted objectivity and deep-dive into contextual factors to which I had no prior knowledge. In the UK, my familiarity with the context and culture, including information on available services and systems, may have helped me gain the trust and acceptance of participants. However, there is risk of losing objectivity or methodological rigour. For example, there is risk that I 'glossed over' important contextual factors because of my existing knowledge. In both research projects, I would be considered an outsider because I am not a person with a disability. Even in the UK, I was an outsider in the research context as I do not have lived experience of disability. I have worked with people with disabilities and their families in the UK, but I myself have not used social care systems, applied for Personal Independence Payment or experienced barriers to healthcare. I may have been aware of the theoretical workings of systems in the UK, but I did not have an individual experience. I thus cannot fully understand experiences that I myself do not have.

This is a regular argument in qualitative research. Is the insider/outsider dichotomy useful? Many consider it overly simplistic [173]. No researcher can be fully an insider or an outsider. All

researchers shift throughout the research process, depending on the situation, as we all have similarities and differences with participants, whether that is disability, race, religion, gender or sexuality. In my experience in the UK and the Gaza Strip, our diverse, collaborative teams helped us to navigate differing positionality and the relative strengths and challenges. The research teams included people from high-income/low-income settings, disabled/non-disabled researchers and diverse organisations. Analysis and interpretation were further aided by steering committees and workshops with people with lived experience and topic experts. These collaborative structures promoted reflexivity and collaborative solutions to mitigate risks. As a non-disabled researcher, collaboration with people with disabilities will always be needed. This may include collaboration with disabled researchers or collaboration with OPDs and people with lived experience to develop research aims and methods, and to interpret findings.

Working in global disability and mental health research, it is also important to continually and critically reflect on power dynamics and perpetuating inequities within research [174, 175]. This is pertinent reflection for my research in the Gaza Strip. The research was conducted within the infrastructure of an existing partnership and project between Atfaluna and CBM. Building on from a global stakeholder workshop, the need for evidence-based MHPSS guidelines were identified within the Gaza Strip and situated within a larger body of work implemented by Atfaluna and CBM. Our role at LSHTM was to conduct research and develop the guidelines in collaboration with Atfaluna. Throughout, we sought to avoid inequitable power dynamics by promoting shared learning and building research capacity. For instance, this was my first research project with deaf and hard of hearing children (or adults) and Atfaluna shared their experience working with this group, whilst I helped build the organisation's research capacity, including research methods and analysis. We aimed to develop a locally owned resource, driven by our engagement with people with lived experience, disability organisations and government. Approaches to ensure the voices of people with disabilities and families were central. Ultimately, though, LSHTM were accountable for the research funding and adhering to funder expectations. This may have created inequities. Collaboration was made more challenging by remote working during COVID-19 and time-pressures arising from the pandemic. Remote working restricted relationship building and limited opportunities for collaboration and shared learning.

It is also important to reflect on the role and influence of research funders. In the Gaza Strip, our local research partners and I asked the funder to delay the project at the start of COVID-19, in order to minimise the impact on the research and development process. The funders denied this and stated that we must continue. This highlighted issues relevant to uneven power dynamics in research. Funders are able to influence how research is conducted, potentially jeopardising quality.

This dynamic is clear within research funding calls, where topics for funding are often based on a funder's own agenda of priority themes, rather than local need. In my research, OPDs also reported that they are constrained in their activities by funder priorities. Although my research is a small example within a major topic area, it demonstrates the need for funders to be more receptive and flexible in order to address local research needs [175].

5. Conclusion

Children with disabilities are at increased risk of mental health conditions. My research and this thesis use a social-ecological framework to contribute evidence on the factors that influence and promote mental health and wellbeing. The mental health and wellbeing of children with disabilities is influenced by a variety of factors. Central to all is disability inclusion. Disability inclusion is needed across all aspects of life to improve the mental health and wellbeing of children with disabilities. Research is needed to continue investigation into the factors that influence mental health and wellbeing among children with disabilities and evaluation research is needed to inform evidence-based disability-inclusive MHPSS systems.

References

1. United Nations General Assembly. Convention on the Rights of Persons with Disabilities - A/RES/61/106. New York, USA: United Nations; 2007.

2. Haegele JA, Hodge S. Disability Discourse: Overview and Critiques of the Medical and Social Models. Quest. 2016;68(2):193-206.

3. World Health Organization. The International Classification of Functioning, Disability and Health. Geneva, Switzerland: WHO; 2001.

4. World Health Organization. Global report on health equity for persons with disabilities. Geneva, Switzerland: WHO; 2022.

 Olusanya BO, Kancherla V, Shaheen A, Ogbo FA, Davis AC. Global and regional prevalence of disabilities among children and adolescents: Analysis of findings from global health databases.
 Frontiers in Public Health. 2022;10:977453.

6. World Health Organization. World report on disability. Geneva, Switzerland: WHO; 2011.

7. Banks LM, Kuper H, Polack S. Poverty and disability in low- and middle-income countries: A systematic review. PLOS One. 2017;12(12):e0189996.

8. International Labour Organization. New ILO database highlights labour market challenges of persons with disabilities. 2022 [Accessed February 10 2024]. Available from:

https://ilostat.ilo.org/new-ilo-database-highlights-labour-market-challenges-of-persons-withdisabilities/#:~:text=The%20labour%20force%20participation%20rate,in%20ten%20persons%20with out%20disabilities.

9. Hughes K, Bellis MA, Jones L, Wood S, Bates G, Eckley L, et al. Prevalence and risk of violence against adults with disabilities: a systematic review and meta-analysis of observational studies. Lancet. 2012;379(9826):1621-9.

10. United Nations Children's Fund. Seen, Counted, Included: Using data to shed light on the well-being of children with disabilities. New York, USA: UNICEF; 2022.

Bright T, Kuper H. A Systematic Review of Access to General Healthcare Services for People with Disabilities in Low and Middle Income Countries. Int J Environ Res Public Health.
 2018;15(9):1879.

12. Missing Billion Initiative, Clinton Health Access Initiative. Reimagining health systems that expect, accept and connect 1 billion people with disabilities. London, UK: MBI and CHAI; 2022.

13. Kuper H, Monteath-van Dok A, Wing K, Danquah L, Evans J, Zuurmond M, Gallinetti J. The impact of disability on the lives of children; cross-sectional data including 8,900 children with disabilities and 898,834 children without disabilities across 30 countries. PLOS One. 2014;9(9):e107300.

United Nations General Assembly. Sustainable Development Goals. Geneva, Switzerland:
 UN; 2015.

Ferrari A, Santomauro D, Herrera A, Shadid J, Ashbaugh C, Erskine H, et al. Global, regional, and national burden of 12 mental disorders in 204 countries and territories, 1990–2019: a systematic analysis for the Global Burden of Disease Study 2019. The Lancet Psychiatry. 2022;9(2):137-50.

16. Piao J, Huang Y, Han C, Li Y, Xu Y, Liu Y, He X. Alarming changes in the global burden of mental disorders in children and adolescents from 1990 to 2019: a systematic analysis for the Global Burden of Disease study. European Child & Adolescent Psychiatry. 2022;31(11):1827-45.

17. Cummins I. The Impact of Austerity on Mental Health Service Provision: A UK Perspective. Int J Environ Res Public Health. 2018;15(6):1145.

 Staniszewska S, Mockford C, Chadburn G, Fenton S-J, Bhui K, Larkin M, et al. Experiences of in-patient mental health services: systematic review. The British Journal of Psychiatry.
 2019;214(6):329-38.

19. Patel V, Saxena S, Lund C, Thornicroft G, Baingana F, Bolton P, et al. The Lancet Commission on global mental health and sustainable development. Lancet. 2018;392(10157):1553-98.

 Rathod S, Pinninti N, Irfan M, Gorczynski P, Rathod P, Gega L, Naeem F. Mental Health Service Provision in Low- and Middle-Income Countries. Health Serv Insights.
 2017;10:1178632917694350.

21. Scherer N, Misganaw E, Njenga M, Ryan G, Eaton J. Global Mental Health and disability in sub-Saharan Africa. In: Ned L, Velarde M, Singh S, Swatz L, Soldatic K, editors. The Routledge International Handbook of Disability and Global Health. 1 ed: Routledge; 2024.

22. Funk M, Bold ND. WHO's QualityRights Initiative: Transforming Services and Promoting Rights in Mental Health. Health Hum Rights. 2020;22(1):69-75.

23. Scott KM, Von Korff M, Alonso J, Angermeyer MC, Bromet E, Fayyad J, et al. Mental–physical co-morbidity and its relationship with disability: results from the World Mental Health Surveys. Psychological Medicine. 2009;39(1):33-43.

24. Augestad LB. Mental Health among Children and Young Adults with Visual Impairments: A Systematic Review. Journal of Visual Impairment & Blindness. 2017;111(5):411-25.

25. Buckley N, Glasson EJ, Chen W, Epstein A, Leonard H, Skoss R, et al. Prevalence estimates of mental health problems in children and adolescents with intellectual disability: A systematic review and meta-analysis. Australian & New Zealand Journal of Psychiatry. 2020;54(10):970-84.

26. Denche-Zamorano A, García-Gil MÁ, Mendoza-Muñoz M, Barrios-Fernandez S. Sadness and Loneliness in Adolescents with Physical, Sensory or Health Problems in Low/Middle-Income Countries. Children. 2023;10(6):996.

27. Lal S, Tremblay S, Starcevic D, Mauger-Lavigne M, Anaby D. Mental health problems among adolescents and young adults with childhood-onset physical disabilities: A scoping review. Front Rehabil Sci. 2022;3:904586.

Jiang F, Kubwimana C, Eaton J, Kuper H, Bright T. The relationship between mental health conditions and hearing loss in low- and middle-income countries. Trop Med Int Health.
 2020;25(6):646-59.

Rotarou ES, Sakellariou D. Depressive symptoms in people with disabilities; secondary analysis of cross-sectional data from the United Kingdom and Greece. Disability and Health Journal. 2018;11(3):367-73.

30. World Health Organization. Mental health. 2024 [Accessed January 8 2024]. Available from: https://www.who.int/health-topics/mental-health#tab=tab_1.

31. Ryan G, Iemmi V, Hanna F, Loryman H, Eaton J. Mental Health for Sustainable Development: A Topic Guide for Development Professionals. London and Brighton, UK: Mental Health Innovation Network and IDS; 2020.

32. World Health Organization. Promoting well-being. 2024 [Accessed April 18 2024]. Available from: https://www.who.int/activities/promoting-well-being.

33. Scherer N, Bright T, Musendo DJ, O'Fallon T, Kubwimana C, Eaton J, et al. Mental health support for children and adolescents with hearing loss: scoping review. BJPsych Open. 2022;8(1):e9.

34. Scherer N, Hussein R, Eaton J, Kabaja N, Kakuma R, Smythe T, Polack S. Development of mental health and psychosocial support (MHPSS) guidelines for deaf and hard of hearing children in the Gaza Strip. PLOS Global Public Health. 2023;3(10):e0002427.

35. Scherer N, Smythe T, Hussein R, Wapling L, Hameed S, Eaton J, et al. Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip. PLOS Global Public Health. 2023;3(6):e0001635.

36. Scherer N, Wiseman P, Watson N, Brunner R, Cullingworth J, Hameed S, et al. 'Do they ever think about people like us?': The experiences of people with learning disabilities in England and Scotland during the COVID-19 pandemic. Critical Social Policy. 2022;43(3):423-47.

37. Haile LM, Kamenov K, Briant PS, Orji AU, Steinmetz JD, Abdoli A, et al. Hearing loss prevalence and years lived with disability, 1990–2019: findings from the Global Burden of Disease Study 2019. Lancet. 2021;397(10278):996-1009.

38. Stevenson J, Kreppner J, Pimperton H, Worsfold S, Kennedy C. Emotional and behavioural difficulties in children and adolescents with hearing impairment: a systematic review and metaanalysis. European Child & Adolescent Psychiatry. 2015;24(5):477-96.

39. Theunissen SCPM, Rieffe C, Netten AP, Briaire JJ, Soede W, Schoones JW, Frijns JHM.
 Psychopathology and Its Risk and Protective Factors in Hearing-Impaired Children and Adolescents: A
 Systematic Review. JAMA Pediatrics. 2014;168(2):170-7.

40. Abu Hamad B, Jones N, Gercama I. Adolescent access to health services in fragile and conflict-affected contexts: The case of the Gaza Strip. Conflict and health. 2021;15(1):40.

41. Abu Hamad B, Gercama I, Jones N, Al Bayoumi N. "I prefer to stay silent": Exploring opportunities for and challenges to adolescents' psychosocial and mental health in Gaza. London, UK: Overseas Development Institute; 2018.

42. Palestinian Central Bureau of Statistics. Palestinian Multiple Indicator Cluster Survey 2019-2020, Survey Findings Report. Ramallah, Palestine: PCDS; 2021.

43. United Nations Office for the Coordination of Humanitarian Affairs. Gaza Strip: The Humanitarian Impact of 15 Years of the Blockade. 2022 [Accessed January 27 2024]. Available from: https://www.unicef.org/mena/media/18041/file/Factsheet_Gaza_Blockade_2022.pdf.

44. Office for National Statistics. Updated estimates of coronavirus (COVID-19) related deaths by disability status, England: 24 January 2020 to 9 March 2022. 2022 [Accessed January 27 2024]. Available from:

https://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarriages/deaths/article s/coronaviruscovid19relateddeathsbydisabilitystatusenglandandwales/24january2020to9march2022 #:~:text=Relative%20to%20non%2Ddisabled%20men,in%20more%2Ddisabled%20men.

45. Williamson EJ, McDonald HI, Bhaskaran K, Walker AJ, Bacon S, Davy S, et al. Risks of covid-19 hospital admission and death for people with learning disability: population based cohort study using the OpenSAFELY platform. BMJ. 2021;374:n1592.

46. Cullingworth J, Watson N, Shakespeare T, Brunner R, Pearson C, Scherer N. "They have been a saving grace in all this": the role of the third sector in disabled people's experiences of COVID-19 and implications for sector–state relations. Voluntary Sector Review. 2022;15(1):1-18.

47. Pearson C, Watson N, Brunner R, Cullingworth J, Hameed S, Scherer N, Shakespeare T. Covid-19 and the Crisis in Social Care: Exploring the Experiences of Disabled People in the Pandemic. Social Policy and Society. 2023;22(3):515-30.

48. Shakespeare T, Watson N, Brunner R, Cullingworth J, Hameed S, Scherer N, et al. Disabled people in Britain and the impact of the COVID-19 pandemic. Social Policy & Administration. 2022;56(1):103-17.

49. Watson N, Shakespeare T. Structural inequalities and disability: Expert report for the UK COVID-19 Public Inquiry. London, UK: UK COVID-19 Inquiry; 2023.

50. Shakespeare T, Ndagire F, Seketi QE. Triple jeopardy: disabled people and the COVID-19 pandemic. Lancet. 2021;397(10282):1331-3.

51. Bronfenbrenner U. Ecological systems theory. In: Vasta R, editor. Annals of Child Development. London, UK: Jessica Kingsley Publishers; 1989.

52. Eriksson M, Ghazinour M, Hammarström A. Different uses of Bronfenbrenner's ecological theory in public mental health research: what is their value for guiding public mental health policy and practice? Social Theory & Health. 2018;16(4):414-33.

53. McLeroy KR, Bibeau D, Steckler A, Glanz K. An ecological perspective on health promotion programs. Health Education Quarterly. 1988;15(4):351-77.

54. Sallis JF, Owen N, Fisher EB. Ecological models of health behavior. Health behavior and health education: Theory, research, and practice, 4th ed. San Francisco, CA, US: Jossey-Bass; 2008. p. 465-85.

55. Bronfenbrenner U. The ecology of human development: Experiments by nature and design. Cambridge, USA: Harvard University Press; 1979.

56. Bronfenbrenner U. Toward an experimental ecology of human development. American psychologist. 1977;32(7):513.

57. World Health Organization. Ottawa Charter for Health Promotion. Geneva: WHO; 1986.

58. United Nations Children's Fund. Global Multisectoral Operational Framework for Mental Health and Psychosocial Support of Children, Adolescents and Caregivers Across Settings. New York, USA: UNICEF; 2022.

59. Drew S, Lavy C, Gooberman-Hill R. What factors affect patient access and engagement with clubfoot treatment in low- and middle-income countries? Meta-synthesis of existing qualitative studies using a social ecological model. Trop Med Int Health. 2016;21(5):570-89.

60. Schwartz N, Buliung R, Wilson K. Disability and food access and insecurity: A scoping review of the literature. Health & Place. 2019;57:107-21.

61. Marmot M. Social determinants of health inequalities. Lancet. 2005;365(9464):1099-104.

62. Hoeft TJ, Fortney JC, Patel V, Unützer J. Task-Sharing Approaches to Improve Mental Health Care in Rural and Other Low-Resource Settings: A Systematic Review. The Journal of Rural Health. 2018;34(1):48-62.

63. Fineman M. The Vulnerable Subject: Anchoring Equality in the Human Condition. Yale J of Law & Feminism. 2008;20.

64. Emerson E, Madden R, Graham H, Llewellyn G, Hatton C, Robertson J. The health of disabled people and the social determinants of health. Public Health. 2011;125(3):145-7.

65. Clough B. Disability and Vulnerability: Challenging the Capacity/Incapacity Binary. Social Policy and Society. 2017;16(3):469-81.

66. Mustaniemi-Laakso M, Katsui H, Heikkilä M. Vulnerability, Disability, and Agency: Exploring Structures for Inclusive Decision-Making and Participation in a Responsive State. International Journal for the Semiotics of Law - Revue internationale de Sémiotique Juridique. 2023;36(4):1581-609.

67. Mladenov T, Brennan CS. Social vulnerability and the impact of policy responses to COVID-19 on disabled people. Sociology of Health & Illness. 2021;43(9):2049-65.

68. Kuran CHA, Morsut C, Kruke BI, Krüger M, Segnestam L, Orru K, et al. Vulnerability and vulnerable groups from an intersectionality perspective. International Journal of Disaster Risk Reduction. 2020;50:101826.

69. UK Department of Health. Reforming the law for adult care and support: the Government's response to Law Commission report 326 on adult social care. London, UK: Department of Health;
2012.

70. UK Disability Unit. The lived experience of disabled people during the COVID-19 pandemic. London, UK: Cabinet Office; 2021.

71. Eskytė I, Lawson A, Orchard M, Andrews E. Out on the streets – Crisis, opportunity and disabled people in the era of Covid-19: Reflections from the UK. Alter. 2020;14(4):329-36.

72. Herrick C. 'We thank you for your sacrifice': Clinical vulnerability, shielding and biosociality in the UK's Covid-19 response. BioSocieties. 2023;18(1):218-40.

73. Alegría M, NeMoyer A, Falgàs Bagué I, Wang Y, Alvarez K. Social Determinants of Mental Health: Where We Are and Where We Need to Go. Curr Psychiatry Rep. 2018;20(11):95.

74. Compton MT, Shim RS. The Social Determinants of Mental Health. FOCUS. 2015;13(4):419-25.

75. Buchner T, Shevlin M, Donovan M-A, Gercke M, Goll H, Šiška J, et al. Same Progress for All? Inclusive Education, the United Nations Convention on the Rights of Persons With Disabilities and Students With Intellectual Disability in European Countries. Journal of Policy and Practice in Intellectual Disabilities. 2021;18(1):7-22.

76. Chibaya G, Chichaya T, Govender P, Naidoo D. Implementation of the United Nations Convention on the Rights of Persons with Disabilities in Africa: a Scoping Review. Disability CBR & Inclusive Development. 2022;32.

77. Ferri D, Leahy A, Šubic N, Urzel L. Implementing the Right of People with Disabilities to Participate in Cultural Life across Five European Countries: Narratives and Counternarratives. Journal of Human Rights Practice. 2022;14(3):859-78.

78. Felix L. Evidence Brief: What progress has been made to operationalise the United Nations' Convention on the Rights of People with Disabilities (UNCRPD) framework to promote inclusive employment? London, UK: Disability Evidence Portal; 2021.

79. Constantino Caycho R. Evidence Brief: How can we promote the effective implementation of Legal Capacity as outlined in Article 12 of the UN CRPD in LMICs? London, UK: Disability Evidence Portal; 2022.

80. Felix L, Mactaggart I, Shakespeare T, Quereshi O. Evidence Brief: What are the core principles that should be considered in the development of policies and programmes related to persons with disabilities in LMICs? London, UK: Disability Evidence Portal; 2021.

81. International Disability Alliance. Not just ticking the disability? Meaningful OPD participation and the risk of tokenism. Geneva, Switzerland: IDA; 2022.

82. Disability Rights UK. Court Overturns Unlawful Ruling of National Disability Strategy. 2023 [Accessed 15 March 2024]. Available from: <u>https://www.disabilityrightsuk.org/news/court-</u>overturns-unlawful-ruling-national-disability-strategy.

83. UK Parliament. The National Disability Strategy: Content, reaction and progress. 2023 [Accessed 15 March 2024]. Available from: <u>https://commonslibrary.parliament.uk/research-briefings/cbp-9599/</u>.

84. Gentry S, Milden L, Kelly MP. Why is translating research into policy so hard? How theory can help public health researchers achieve impact? Public Health. 2020;178:90-6.

85. Erismann S, Pesantes MA, Beran D, Leuenberger A, Farnham A, Berger Gonzalez de White M, et al. How to bring research evidence into policy? Synthesizing strategies of five research projects in low-and middle-income countries. Health Research Policy and Systems. 2021;19(1):29.

86. Oliver K, Hopkins A, Boaz A, Guillot-Wright S, Cairney P. What works to promote researchpolicy engagement? Evidence & Policy. 2022;18(4):691-713.

Kathryn S, Lynsay M, Sharon Anne S, Peter C, Janis B, Jane MB, et al. A new framework for developing and evaluating complex interventions: update of Medical Research Council guidance.
BMJ. 2021;374:n2061.

88. Whittle EL, Fisher KR, Reppermund S, Lenroot R, Trollor J. Barriers and Enablers to Accessing Mental Health Services for People With Intellectual Disability: A Scoping Review. Journal of Mental Health Research in Intellectual Disabilities. 2018;11(1):69-102. 89. Jacobs M, Downie H, Kidd G, Fitzsimmons L, Gibbs S, Melville C. Mental health services for children and adolescents with learning disabilities: a review of research on experiences of service users and providers. British Journal of Learning Disabilities. 2016;44(3):225-32.

90. World Health Organization. Mental Health Gap Action Programme (mhGAP) guideline for mental, neurological and substance use disorders. Geneva, Switzerland: WHO; 2023.

91. Wainberg ML, Scorza P, Shultz JM, Helpman L, Mootz JJ, Johnson KA, et al. Challenges and Opportunities in Global Mental Health: a Research-to-Practice Perspective. Curr Psychiatry Rep. 2017;19(5):28.

92. Gréaux M, Moro MF, Kamenov K, Russell AM, Barrett D, Cieza A. Health equity for persons with disabilities: a global scoping review on barriers and interventions in healthcare services. Int J Equity Health. 2023;22(1):236.

Bayetti C, Bakhshi P, Davar B, Khemka GC, Kothari P, Kumar M, et al. Critical reflections on the concept and impact of "scaling up" in Global Mental Health. Transcultural psychiatry.
2023;60(3):602-9.

94. Charlson F, van Ommeren M, Flaxman A, Cornett J, Whiteford H, Saxena S. New WHO prevalence estimates of mental disorders in conflict settings: a systematic review and meta-analysis. Lancet. 2019;394(10194):240-8.

95. Keya TA, Leela A, Habib N, Rashid M, Bakthavatchalam P. Mental Health Disorders Due to Disaster Exposure: A Systematic Review and Meta-Analysis. Cureus. 2023;15(4):e37031.

96. Oliveira JMD, Butini L, Pauletto P, Lehmkuhl KM, Stefani CM, Bolan M, et al. Mental health effects prevalence in children and adolescents during the COVID-19 pandemic: A systematic review. Worldviews Evid Based Nurs. 2022;19(2):130-7.

97. Leung XY, Kavanagh AM, Quang QT, Shields M, Aitken Z. A systematic review of the impact of the COVID-19 pandemic on the mental health of adolescents and young people with disabilities aged 15–29 years. BMC Public Health. 2023;23(1):1390.

98. Mann M, McMillan JE, Silver EJ, Stein REK. Children and Adolescents with Disabilities and Exposure to Disasters, Terrorism, and the COVID-19 Pandemic: a Scoping Review. Curr Psychiatry Rep. 2021;23(12):80.

99. The Disabled Children's Partnership. Failed and Forgotten: Research by the Learning Hub at the Disabled Children's Partnership. London, UK: The Disabled Children's Partnership; 2023.

100. European Disability Forum. Review of Disability-inclusive Disaster Risk Reduction Policy and Practiceacross Europe and Central Asia. Brussels, Belgium: European Disability Forum; 2021.

101. Jesus TS, Kamalakannan S, Bhattacharjya S, Bogdanova Y, Arango-Lasprilla JC, Bentley J, et al. PREparedness, REsponse and SySTemic transformation (PRE-RE-SyST): a model for disability-inclusive

pandemic responses and systemic disparities reduction derived from a scoping review and thematic analysis. International Journal for Equity in Health. 2021;20(1):204.

102. Rohwerder B, Wong S, Pokharel S, Khadka D, Poudyal N, Prasai S, et al. Describing adolescents with disabilities' experiences of COVID-19 and other humanitarian emergencies in lowand middle-income countries: a scoping review. Glob Health Action. 2022;15(1):2107350.

103. Quereshi O. Evidence Brief: How can we ensure the safety and wellbeing of children with disabilities during humanitarian crises? London, UK: Disability Evidence Portal; 2020.

104. Inter-Agency Standing Committee. Disability and Inclusion in MHPSS. Geneva, Switzerland: IASC; 2024.

105. Kwan C, Gitimoghaddam M, Collet JP. Effects of Social Isolation and Loneliness in Children with Neurodevelopmental Disabilities: A Scoping Review. Brain sciences. 2020;10(11).

106. Woodgate R, Gonzalez M, Demczuk L, Snow W, Barriage S, Kirk S. Disability and Rehabilitation How do peers promote social inclusion of children with disabilities? A mixed-methods systematic review View supplementary material How do peers promote social inclusion of children with disabilities? A mixed-methods systematic review. Disability and Rehabilitation. 2019;42:2553-79.

107. Rohwerder B. Disability stigma in developing countries. Brighton, UK: Institute of Development Studies; 2018.

108. Virendrakumar B, Stephen C, Jolley E, Schmidt E. Disability-related stigma and discrimination in sub-Saharan Africa and South Asia: a systematic literature review. Haywards Heath, UK: Sightsavers; 2021.

109. Smythe T, Adelson JD, Polack S. Systematic review of interventions for reducing stigma experienced by children with disabilities and their families in low- and middle-income countries: state of the evidence. Trop Med Int Health. 2020;25(5):508-24.

Mukolo A, Heflinger CA, Wallston KA. The stigma of childhood mental disorders: a conceptual framework. Journal of the American Academy of Child and Adolescent Psychiatry. 2010;49(2):92-103.

111. Radez J, Reardon T, Creswell C, Lawrence PJ, Evdoka-Burton G, Waite P. Why do children and adolescents (not) seek and access professional help for their mental health problems? A systematic review of quantitative and qualitative studies. European Child & Adolescent Psychiatry. 2021;30(2):183-211.

112. Freţian AM, Graf P, Kirchhoff S, Glinphratum G, Bollweg TM, Sauzet O, Bauer U. The Long-Term Effectiveness of Interventions Addressing Mental Health Literacy and Stigma of Mental Illness in Children and Adolescents: Systematic Review and Meta-Analysis. International journal of public health. 2021;66:1604072.

Scherer N, Verhey I, Kuper H. Depression and anxiety in parents of children with intellectual and developmental disabilities: A systematic review and meta-analysis. PLOS One.
2019;14(7):e0219888.

114. Masefield SC, Prady SL, Sheldon TA, Small N, Jarvis S, Pickett KE. The Caregiver Health Effects of Caring for Young Children with Developmental Disabilities: A Meta-analysis. Maternal and child health journal. 2020;24(5):561-74.

115. Gu X, Shan X, Wang L, Gao W-Y, Liu X-Q. Prevalence of psychological problems among caregivers of children and adolescents with neurodevelopmental disorders during the COVID-19 pandemic: A meta-analysis and systematic review. Research in Developmental Disabilities. 2023;143:104632.

116. Stein A, Pearson RM, Goodman SH, Rapa E, Rahman A, McCallum M, et al. Effects of perinatal mental disorders on the fetus and child. Lancet. 2014;384(9956):1800-19.

Martucci M, Aceti F, Giacchetti N, Sogos C. The mother-baby bond: a systematic review
about perinatal depression and child developmental disorders. Rivista di Psichiatria. 2021;56(5):22336.

118. Risi A, Pickard JA, Bird AL. The implications of parent mental health and wellbeing for parentchild attachment: A systematic review. PLOS One. 2021;16(12):e0260891.

119. Northrup RA, Jones E, Singh V, Holingue C, Meck M, Gurnett CA, et al. Caregiver perspectives on the continued impact of the COVID-19 pandemic on children with intellectual/developmental disabilities. Front Pediatr. 2023;11:1196275.

120. Wormald A, McGlinchey E, D'Eath M, Leroi I, Lawlor B, McCallion P, et al. Impact of COVID-19 Pandemic on Caregivers of People with an Intellectual Disability, in Comparison to Carers of Those with Other Disabilities and with Mental Health Issues: A Multicountry Study. Int J Environ Res Public Health. 2023;20(4).

121. Fang L, Qiao S, Miao H, Hengyu Z. Factors associated with caregiver burden among family caregivers of children with cerebral palsy: a systematic review. BMJ Open. 2023;13(4):e065215.

122. Linden M, Leonard R, Forbes T, Brown M, Marsh L, Todd S, et al. Experiences of UK and Irish family carers of people with profound and multiple intellectual disabilities during the COVID-19 pandemic. BMC Public Health. 2023;23(1):2475.

123. Lancaster K, Bhopti A, Kern ML, Taylor R, Janson A, Harding K. Effectiveness of peer support programmes for improving well-being and quality of life in parents/carers of children with disability or chronic illness: A systematic review. Child Care Health Dev. 2023;49(3):485-96.

124. Mitchell R, Karchmer M. Chasing the Mythical Ten Percent: Parental Hearing Status of Deaf and Hard of Hearing Students in the United States. Sign Language Studies. 2004;4:138-63.

125. Reichow B, Kogan C, Barbui C, Maggin D, Salomone E, Smith IC, et al. Caregiver skills training for caregivers of individuals with neurodevelopmental disorders: A systematic review and metaanalysis. Dev Med Child Neurol. 2023:1-12.

126. Hohlfeld A, Harty M, Engel M. Parents of children with disabilities: A systematic review of parenting interventions and self-efficacy. African Journal of Disability. 2018;7:437.

Albanese AM, Russo GR, Geller PA. The role of parental self-efficacy in parent and child well-being: A systematic review of associated outcomes. Child: Care, Health and Development.
2019;45(3):333-63.

Lu W, Xu C, Hu X, Liu J, Zhang Q, Peng L, et al. The Relationship Between Resilience and
 Posttraumatic Growth Among the Primary Caregivers of Children With Developmental Disabilities:
 The Mediating Role of Positive Coping Style and Self-Efficacy. Frontiers in Psychology. 2022;12.

129. Cohen AK, Rai M, Rehkopf DH, Abrams B. Educational attainment and obesity: a systematic review. Obesity Reviews. 2013;14(12):989-1005.

130. Raghupathi V, Raghupathi W. The influence of education on health: an empirical assessment of OECD countries for the period 1995–2015. Archives of Public Health. 2020;78(1):20.

131. Mizunoya S, Mitra S, Yamasaki I. Disability and school attendance in 15 low- and middleincome countries. World Development. 2018;104:388-403.

132. Bani Odeh K, Lach LM. Barriers to, and facilitators of, education for children with disabilities worldwide: a descriptive review. Frontiers in Public Health. 2024;11.

133. Egan R, Wilson R, Robertson M, Scandiffio J, Ross-White A, Coderre-Ball AM. Barriers and facilitators to education experienced by students with disabilities in low- and middle-income African countries: a systematic review of qualitative evidence. JBI Evidence Synthesis. 2022;20(10).

134. Huus K, Schlebusch L, Ramaahlo M, Samuels A, Berglund IG, Dada S. Barriers and facilitators to participation for children and adolescents with disabilities in low- and middle-income countries - A scoping review. African Journal of Disability. 2021;10:771.

135. United Nations Educational Scientific and Cultural Organization. The Salamanca Statement and Framework for Action on Special Needs Education. Salamanca, Spain: UNESCO.

Oh-Young C, Filler J. A meta-analysis of the effects of placement on academic and social skill outcome measures of students with disabilities. Research in Developmental Disabilities. 2015;47:80-92.

137. Jodl J, Bian M. Global State of Inclusion in Education: A Review of the Literature.Washington, USA: Special Olympics Global Center for Inclusion in Education; 2023.

138. United Nations Educational Scientific and Cultural Organization. Global Education Monitoring Report. Paris, France: UNESCO; 2020.

139. Mendoza M, Heymann J. Implementation of Inclusive Education: A Systematic Review of Studies of Inclusive Education Interventions in Low- and Lower-Middle-Income Countries. International Journal of Disability, Development and Education. 2022;71(3):299-316.

140. Van Mieghem A, Verschueren K, Petry K, Struyf E. An analysis of research on inclusive education: a systematic search and meta review. International Journal of Inclusive Education. 2020;24(6):675-89.

141. Lindner K-T, Schwab S, Emara M, Avramidis E. Do teachers favor the inclusion of all students? A systematic review of primary schoolteachers' attitudes towards inclusive education. European Journal of Special Needs Education. 2023;38(6):766-87.

142. Henriksen IO, Ranøyen I, Indredavik MS, Stenseng F. The role of self-esteem in the development of psychiatric problems: a three-year prospective study in a clinical sample of adolescents. Child Adolesc Psychiatry Ment Health. 2017;11:68.

143. Orth U, Robins RW, Widaman KF. Life-span development of self-esteem and its effects on important life outcomes. J Pers Soc Psychol. 2012;102(6):1271-88.

144. Keane L, Loades M. Review: Low self-esteem and internalizing disorders in young people - a systematic review. Child Adolesc Ment Health. 2017;22(1):4-15.

145. Mann M, Hosman CM, Schaalma HP, de Vries NK. Self-esteem in a broad-spectrum approach for mental health promotion. Health Educ Res. 2004;19(4):357-72.

146. Stets JE, Burke PJ. Identity Theory and Social Identity Theory. Social Psychology Quarterly. 2000;63(3):224-37.

147. Stets JE, Burke PJ. Self-Esteem and Identities. Sociological Perspectives. 2014;57(4):409-33.

148. Haslam S, Jetten J, Postmes T, Haslam C. Social Identity, Health and Well-Being: An Emerging Agenda for Applied Psychology. Applied Psychology. 2009;58.

149. Glickman N, Carey J. Measuring deaf cultural identities: A preliminary investigation.Rehabilitation Psychology. 1993;38(4):275-83.

150. Glickman N. Deaf identity development : construction and validation of a theoretical model.Massachusetts, USA: Doctoral Thesis; 1993.

151. Bat-Chava Y. Diversity of deaf identities. Am Ann Deaf. 2000;145(5):420-8.

152. Hintermair M. Self-esteem and Satisfaction With Life of Deaf and Hard-of-Hearing People—A
Resource-Oriented Approach to Identity Work. The Journal of Deaf Studies and Deaf Education.
2008;13(2):278-300.

153. Jambor E, Elliott M. Self-esteem and Coping Strategies among Deaf Students. The Journal of Deaf Studies and Deaf Education. 2005;10(1):63-81.

154. Jetten J, Haslam S, Cruwys T, Branscombe N. Social Identity, Stigma, and Health. In: Major B, Dovidio J, Link B, editors. The Oxford Handbook of Stigma, Discrimination, and Health. 1 ed. 2017. p. 301-16.

155. Fellinger J, Holzinger D, Pollard R. Mental health of deaf people. Lancet. 2012;379(9820):1037-44.

156. Hall ML, Hall WC, Caselli NK. Deaf children need language, not (just) speech. First Language. 2019;39(4):367-95.

157. Fellinger J, Holzinger D, Sattel H, Laucht M, Goldberg D. Correlates of mental health disorders among children with hearing impairments. Dev Med Child Neurol. 2009;51(8):635-41.

158. Chapman M, Dammeyer J. The Significance of Deaf Identity for Psychological Well-Being. The Journal of Deaf Studies and Deaf Education. 2017;22(2):187-94.

159. McIlroy G, Storbeck C. Development of Deaf Identity: An Ethnographic Study. The Journal of Deaf Studies and Deaf Education. 2011;16(4):494-511.

160. WorldAtlas. Which Countries Recognize Sign Language As An Official Language? 2024 [Accessed 20 April 2024]. Available from: <u>https://www.worldatlas.com/articles/which-countries-</u> recognize-sign-language-as-an-official-language.html.

161. Davydov DM, Stewart R, Ritchie K, Chaudieu I. Resilience and mental health. Clin Psychol Rev. 2010;30(5):479-95.

162. Ghanei Gheshlagh R, Sayehmiri K, Ebadi A, Dalvandi A, Dalvand S, Maddah S, Tabrizi K. The Relationship Between Mental Health and Resilience: A Systematic Review and Meta-Analysis. Iranian Red Crescent Medical Journal. 2017;19(6):8.

163. Dray J, Bowman J, Campbell E, Freund M, Wolfenden L, Hodder RK, et al. Systematic Review of Universal Resilience-Focused Interventions Targeting Child and Adolescent Mental Health in the School Setting. Journal of the American Academy of Child and Adolescent Psychiatry. 2017;56(10):813-24.

164. Leppin AL, Bora PR, Tilburt JC, Gionfriddo MR, Zeballos-Palacios C, Dulohery MM, et al. The efficacy of resiliency training programs: a systematic review and meta-analysis of randomized trials. PLOS One. 2014;9(10):e111420.

165. García-Parra M, Negre F, Verger S. Educational Programs to Build Resilience in Children,
Adolescent or Youth with Disease or Disability: A Systematic Review. Education Sciences.
2021;11(9):464.

166. Town R, Hayes D, March A, Fonagy P, Stapley E. Self-management, self-care, and self-help in adolescents with emotional problems: a scoping review. European Child & Adolescent Psychiatry. 2023.

167. Truscott A, Hayes D, Bardsley T, Choksi D, Edbrooke-Childs J. Defining young people's mental health self-care: a systematic review and co-development approach. European Child & Adolescent Psychiatry. 2023.

168. Lindsay S, Kingsnorth S, McDougall C, Keating H. A systematic review of self-management interventions for children and youth with physical disabilities. Disability and rehabilitation. 2014;36(4):276-88.

169. Sandjojo J, Eltringham EG, Gebhardt WA, Zedlitz AMEE, Embregts PJCM, Evers AWM. Selfmanagement interventions for people with intellectual disabilities: A systematic review. Patient Education and Counseling. 2020;103(10):1983-96.

170. Keen S, Lomeli-Rodriguez M, Joffe H. From Challenge to Opportunity: Virtual Qualitative Research During COVID-19 and Beyond. Int J Qual Methods. 2022;21:16094069221105075.

171. Bramer WM, Rethlefsen ML, Kleijnen J, Franco OH. Optimal database combinations for literature searches in systematic reviews: a prospective exploratory study. Systematic Reviews. 2017;6(1):245.

172. Njelesani J, Mlambo V, Denekew T, Hunleth J. Inclusion of children with disabilities in qualitative health research: A scoping review. PLOS One. 2022;17(9):e0273784.

173. Milligan L. Insider-outsider-inbetweener? Researcher positioning, participative methods and cross-cultural educational research. Compare: A Journal of Comparative and International Education. 2016;46(2):235-50.

174. Mishal K, Seye A, Tammam A, Emanuele C, Sarah H, Afifah R-S. Decolonising global health in 2021: a roadmap to move from rhetoric to reform. BMJ Global Health. 2021;6(3):e005604.

175. Kumar R, Khosla R, McCoy D. Decolonising global health research: Shifting power for transformative change. PLOS Global Public Health. 2024;4(4):e0003141.

Appendices

Paper 1: Research Paper Cover Sheet and Confirmation Retention of Copyright License

Scherer N, Bright T, Musendo DJ, O'Fallon T, Kubwimana C, Eaton J, Kakuma R, Smythe T, Polack S. Mental health support for children and adolescents with hearing loss: scoping review. *BJPsych Open* (2021) 8 (1), e9 10.1192/bjo.2021.1045



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A – Student Details

Student ID Number	1901335	Title	Mr
First Name(s)) Nathaniel Joseph		
Surname/Family Name	Scherer		
Thesis Title	The mental health and wellbeing of children with disabilities		
Primary Supervisor	Sarah Polack		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B - Paper already published

Where was the work published? BJPsych O			
When was the work published?	06/12/2021		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	PhD by Publicat	ion	
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I managed all aspects of this scoping review. I led protocol development in collaboration with my supervisors and co-authors. I developed the search string and conducted the search across the databases. I trained the team in screening, developed the screening procedures and led discussions on the inclusion of papers. Once data was extracted by myself and the team, I led discussions on the synthesis of results and wrote the published manuscript. I finalised the manuscript for submission based on co-author comments and submitted to the journal. I took the lead responding to peer-review comments.
---	---

SECTION E

Student Signature	Nathaniel Scherer
Date	22/04/2024

Supervisor Signature	Sarah Polack
Date	22/04/2024

	Mental health support for children and adolescents with hearing loss: scoping review		
UNIVERSITY PRESS	Author: Nathaniel Scherer, Tess Bright, David John Musendo, Timothy O'Fallon, Chris Kubwimana, Julian Eaton, Ritsuko Kakuma, Tracey Smythe, Sarah Polack®right=Copyright © The Auth by Cambridge University Press on behalf of the Royal College of Psychiatrists Publication: BJPsych Open Publisher: Cambridge University Press	nor(s), 2021. I	Publish
	Copyright © 2021, Cambridge University Press		
Creative Commons			
This is an open access article	distributed under the terms of the Creative Commons CC BY license, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is pr	roperly cited	
You are not required to obtai	n permission to reuse this article.		

Туре	Review
Information	BJPsych Open, Volume 8, Issue 1, January 2022, e9 DOI: https://doi.org/10.1192/bjo.2021.1045 Check for updates
Creative Commons	This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution, and reproduction in any medium, provided the original work is properly cited.
Copyright	Copyright © The Author(s), 2021. Published by Cambridge University Press on behalf of the Royal College of Psychiatrists

Paper 2: Research Paper Cover Sheet and Confirmation Retention of Copyright License

Scherer N, Smythe T, Hussein R, Wapling L, Hameed S, Eaton J, Kabaja N, Kakuma R, Polack S. Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip. *PLOS Global Public Health* (2023) 3 (6), e0001635 10.1371/journal.pgph.0001635



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A - Student Details

Student ID Number	1901335 Title Mr		Mr
First Name(s)	Nathaniel Joseph		
Surname/Family Name	Scherer		
Thesis Title	The mental health and wellbeing of children with disabilities		
Primary Supervisor	Sarah Polack		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B – Paper already published

Where was the work published?	PLOS Global Public Health		
When was the work published?	06/06/2023		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	PhD by Publication		
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I managed this qualitative research project, developing the study design, methods, sampling strategy and data collection tools, in collaboration with a multidisciplinary team. With COVID-19, I was unable to travel to the Gaza Strip for data collection. I instead trained and managed the local research remotely. I trained a local researcher and supervised recruitment and data collection. I led analysis of the transcripts. I drafted the manuscript and finalised for submission based on co-author comments. I took the lead responding to peer-review comments. This published work was one component of a research project, to which I was the project lead and managed all aspects.
---	--

SECTION E

Student Signature	Nathaniel Scherer
Date	22/04/2024

Supervisor Signature	Sarah Polack
Date	22/04/2024

Citation: Scherer N, Smythe T, Hussein R, Wapling L, Hameed S, Eaton J, et al. (2023) Communication, inclusion and psychological wellbeing among deaf and hard of hearing children: A qualitative study in the Gaza Strip. PLOS Glob Public Health 3(6): e0001635. https://doi.org/10.1371/journal.pgph.0001635

Editor: Julia Robinson, PLOS: Public Library of Science, UNITED STATES

Received: December 5, 2022; Accepted: April 19, 2023; Published: June 6, 2023

Copyright: © 2023 Scherer et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Paper 3: Research Paper Cover Sheet and Confirmation Retention of Copyright License

Scherer N, Hussein R, Eaton J, Kabaja N, Kakuma R, Smythe T, Polack S. Development of mental health and psychosocial support (MHPSS) guidelines for deaf and hard of hearing children in the Gaza Strip. *PLOS Global Public Health* (2023) 3 (10), e0002427 10.1371/journal.pgph.0002427



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A – Student Details

Student ID Number	1901335	Title	Mr
First Name(s)	Nathaniel Joseph		
Surname/Family Name	Scherer		
Thesis Title	The mental health and wellbeing of children with disabilities		
Primary Supervisor	Sarah Polack		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B – Paper already published

Where was the work published?	PLOS Global Public Health		
When was the work published?	16/10/2023		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	PhD by Publication		
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	This published work describes the development process of the MHPSS guidelines, of which I managed all aspects under the guidance of my supervisors, including study design, research methodologies, data collection and managing partnership with the local researcher, implementation partner, local steering committee, and funder. This included scoping review, qualitative research and pilot. I wrote the manuscript, with feedback and support from co-authors.
---	--

SECTION E

Student Signature	Nathaniel Scherer
Date	22/04/2024

Supervisor Signature	Sarah Polack
Date	22/04/2024

Citation: Scherer N, Hussein R, Eaton J, Kabaja N, Kakuma R, Smythe T, et al. (2023) Development of mental health and psychosocial support (MHPSS) guidelines for deaf and hard of hearing children in the Gaza Strip. PLOS Glob Public Health 3(10): e0002427. https://doi.org/10.1371/journal.pgph.0002427

Editor: Mohamed Salama, The American University in Cairo, EGYPT

Received: May 5, 2023; Accepted: September 2, 2023; Published: October 16, 2023

Copyright: © 2023 Scherer et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Paper 4: Research Paper Cover Sheet and Confirmation Retention of Copyright License

Scherer N, Wiseman P, Watson N, Brunner R, Cullingworth J, Hameed S, Pearson C, Shakespeare T. 'Do they ever think about people like us?': The experiences of people with learning disabilities in England and Scotland during the COVID-19 pandemic. *Critical Social Policy* (2022) 43 (3), 423-447, 10.1177/02610183221109147



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A – Student Details

Student ID Number	1901335	Title	Mr
First Name(s)	Nathaniel Joseph		
Surname/Family Name	Scherer		
Thesis Title	The mental health and wellbeing of children with disabilities		
Primary Supervisor	Sarah Polack		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B – Paper already published

Where was the work published?	Critical Social Policy		
When was the work published?	27/06/2023		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	PhD by Publication		
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	For this study, I worked with a team from LSHTM and the University of Glasgow to develop the research methods and data collection tools. I conducted a subset of the interviews myself, collaborating with other team members to collect others. I analysed the data included in this article, discussing my interpretation and theoretical reflections with two researchers from the University of Glasgow (the second and third authors). I wrote the manuscript and led various dissemination activities, including workshops, webinars and conference presentations.
---	--

SECTION E

Student Signature	Nathaniel Scherer
Date	22/04/2024

Supervisor Signature	Sarah Polack
Date	22/04/2024

Information Authors

Published In



Article first published online: June 27, 2022

Issue published: August 2023



Keywords

COVID-19, disability, health, learning disabilities, vulnerability,

Rights and permissions

© The Author(s) 2022.



This article is distributed under the terms of the Creative Commons Attribution 4.0 License (<u>https://creativecommons.org/licenses/by/4.0/</u>) which permits any use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access page (<u>https://us.sagepub.com/enus/nam/open-access-at-sage</u>).